

An Assessment of the Livelihood Vulnerability of the Riverbank Erosion Hazard and its Impact on Food Security for Rural Households in Bangladesh

A Dissertation Submitted for the Award of Doctor of Philosophy

G M Monirul Alam MS (Agricultural Economics), BScAgEco (Hons)

> School of Commerce Faculty of Business, Education, Law & Arts University of Southern Queensland Toowoomba, Queensland, Australia

> > February 2016

Dedicated to my father

the late Kayem Uddin Ahmed

ABSTRACT

As the effects of climate change and hazards are starting to be felt worldwide, there are certain frontline countries that are most at risk and Bangladesh is genuinely at risk in terms of its economic viability and food security unless its citizens develop adaptation strategies to compensate for these effects. This study analyses how the impacts of climate change and hazards (specifically riverbank erosion) are already jeopardising the livelihood and food security of rural riparian (riverbank and *char*) households in Bangladesh, compromising their access to arable land, and thereby holding back their potential for both sustenance and economic development.

The researcher has conducted extensive research in two severe riverbank erosion-prone districts in Bangladesh to assess the severity of these problems and to seek the strategies the affected people deploy to offset the effects. This study takes a holistic approach to two key vulnerability assessment methods – the Livelihood Vulnerability Index (LVI) and the Climate Vulnerability Index (CVI). Importantly, this study also develops an indicator-based Resilience Capacity Index (RCI) in order to understand the factors influencing the resilience capacity of these households.

This study reveals that the LVI and CVI values are different between *char* (sandbar) and riverbank communities: households inhabiting *char* lands display the most vulnerability to climate change and hazards. Also, riparian households are found to be vulnerable due to their relative inaccessibility and low livelihood status which, coupled with the impact of the climate on river morphology, are causing erosion and a loss of land with a consequent decrease in economic potential, thereby perpetuating a cycle of poverty. Creating employment opportunities, increasing the level of education and ensuring access to food, water and health services are potential strategies that are likely to enhance the resilience capacity of such vulnerable households in Bangladesh.

In regards to food security, more than 50% of the households are in the 'food insecure' category, with a per capita calorie consumption of 12% less than the standard minimum daily requirement. The estimated low Food Security Index (FSI) value indicates that these households can usually manage food twice per day for their family members. The results of logit modelling indicate that household size, educational attainment, adoption of livestock and access to non-farm earnings are

important determinants of household food security. This study also finds new evidence that suggests access to improved health care also needs policy support in parallel with improved access to food to achieve and to sustain long-term food security in Bangladesh. Properly targeted income transfers and credit programs along with infrastructure and human development programs in the erosion-affected areas across the country may have very high payoffs by improving food security, and thus, reducing poverty in the long-term.

To build resilience, households are autonomously adopting adaptation strategies such as diversifying crops, tree plantation (generally by large and medium farmers), and homestead gardening and migration (generally by small and landless farmers). However, some important barriers to adaptation are felt heterogeneously among the farming groups: among these are access to credit and a lack of information on appropriate adaptation strategies. The results of multi-nominal logit modelling indicate that the choice of an adaptation strategy is influenced significantly by a household head's education, household income, farm category, access to institutions and social capital. To support adaptation locally and to enhance households' resilience to cope better with riverbank hazards and other climate change issues, government intervention through planned adaptation such as access to institutions, credit facilities and a package of technologies through agro-ecologically based research are required.

This study has contributed to our knowledge base through tailoring various theories and approaches in the context of riparian households in Bangladesh. The innovative coping and adaptation strategies could provide new insights for households in other hazard-prone regions in the world. The analytical framework used for assessing vulnerability, resilience, household food security and adaptation strategies should be replicated in other countries having similar characteristics to Bangladesh that are experiencing adverse impacts from climate change.

CERTIFICATION OF DISSERTATION

I hereby affirm that the ideas, results, analyses and conclusions presented in this dissertation are entirely my own effort, except where otherwise acknowledged. It is also certified that the intellectual content of this research is original and has not been previously submitted for any other academic award.

Signature of Candidate

G M Monirul Alam

ENDORSEMENT

Signature of Principal Supervisor

Associate Professor Khorshed Alam

(Jean really)

Signature of Associate Supervisor

Associate Professor Shahbaz Mushtaq

29/02/2016

Date

29/02/2016

29/02/2016

Date

Date

OUTCOMES OF THE THESIS

Nine journal papers, one book chapter and four conference papers have been drawn from this thesis as follows:

Journal Papers

- Alam, GMM, Alam, K, Shahbaz, M & Clarke, ML (2016) 'Drivers of vulnerability to climatic change in riparian *char* and river-bank households in Bangladesh: Implications for policy, livelihoods and social development', *Ecological Indicators*, (Under review).
- 2. Alam, GMM, Alam, K, Shahbaz, M & Filho, WL (2015) 'Hazard-prone rural communities in Bangladesh: How resilient are they to the impact of climate change and variability?', *Journal of Rural Studies*, (Under review).
- Alam, GMM, Alam, K & Shahbaz, M (2015) 'Influence of household head physical health status on food security: Evidence from vulnerable rural households in Bangladesh', *World Development*, (Under review).
- 4. Alam, GMM, Alam, K & Shahbaz, M (2015) 'Influence of institutional access and social capital on adaptation choices: Empirical evidence from vulnerable rural households in Bangladesh', *Ecological Economics*, (Under review).
- Alam, GMM, Alam, K, Shahbaz, M & Hossain, M (2015) 'Riverbank erosion hazards, food insecurity and rural household responses: An empirical evidence from Bangladesh', *Food Security*, (Under review).
- Alam, GMM, Alam, K & Shahbaz, M (2015) 'Climate change perceptions and local adaptation strategies of hazard-prone rural households in Bangladesh', *Natural Hazards*, (Under review).
- Alam, GMM, Alam, K & Shahbaz, M (2015) 'Livelihood cycle and vulnerability of rural households to climate change and hazards in Bangladesh,' *Environmental Management*, (Under review).
- Alam, GMM, Alam, K & Shahbaz, M, M, Khatun, M & Mamun, SAK (2015)
 'Does household income matter for mobile phone adoption? A case study from rural Bangladesh', *Information Technology & People*, (Under review).

Book Chapter

Alam, GMM, Alam, K, Shahbaz, M & Filho, WL (2016) 'Strategies and barriers to adaptation: Evidence from hazard-prone rural households in Bangladesh'. In: Filho, WL & Nalau, J (eds), *Limits to Adaptation: Insights and Experiences*, Springer.

Conference Papers

- Alam, GMM, Alam, K & Shahbaz, M (2014) 'Determinants of food security: An empirical analysis from Bangladesh resource-poor rural households, Paper presented at the Symposium on Research Journeys and the Production of Knowledge, 24 October 2014, USQ, Springfield.
- Alam, GMM, Alam, K, Shahbaz, M, Khatun, M & Mamun, SAK (2014) 'The causal relationship between rural household income and mobile phone adoption: A case study from Bangladesh, Paper presented at the Symposium on Broadband-enabled Communities and Regional Development, 10-11 November 2014, USQ, Toowoomba.
- Alam, GMM, Alam, K & Shahbaz, M (2015) 'Is migration is an appropriate coping strategy for hazard-prone rural households in Bangladesh?', Paper presented at the Symposium on Language, Culture and Technology in a Connected World, 1 June 2015, USQ, Toowoomba.
- Alam, GMM, Alam, K & Shahbaz, M (2015) 'Analysing Livelihood Status and Policy Options to Improving Food Access for Resource-Poor Rural Households in Bangladesh', Paper presented in the HRD Student Seminar at USQ, 19-20 November, USQ, Toowoomba.

ACKNOWLEDGEMENTS

My PhD dream has come true thanks to the contribution of many people, although it is hardly possible to acknowledge my debt and gratefulness to all of them. Let me take the opportunity to acknowledge special gratefulness to a special few. First and foremost, I give thanks to Almighty Allah for unparalleled grace and strength to bring this work to realisation.

I feel proud to express my deepest sense of gratitude and hearty appreciation to my supervisors Associate Professor Khorshed Alam and Associate Professor Shahbaz Mushtaq for their time, scholastic guidance, incisive criticism, proper instruction and continuous supervision throughout the period of this PhD journey. Without their total cooperation and sacrifice it would not have been possible to complete the thesis within two and half years only.

In addition, I would like to express my gratitude to Professor MA Sattar Mandal, Rezaul Karim Talukder, Md Ferdous Alam, SM Fakhrul Islam, M Kamruzzaman, WHM Zaim, Mizanul Haque Kazal, Akhtaruzzaman Khan, Dr Monayem Miah, Dr Rayhan Hayat Sawer, Dr Abdur Rashid Sarker and many others for their insightful comments and guidance throughout my PhD research which enrich my research from various perspectives. The contribution of Professor Michèle L Clarke, University of Nottingham, UK, Professor Walter Leal Filho, Manchester Metropolitan University, UK and Associate Professor Moazzem Hossain, Griffith University, Australia is highly acknowledged. Their comments and suggestions really help to set the right direction of this research.

I sincerely acknowledged the University of Southern Queensland, Australia, for awarding me USQ Post Postgraduate Research Scholarship for International PhD and later on top-up scholarship and award for thesis editing. I express my gratitude to Bangabandhu Sheikh Mujibur Rahman Agricultural University, Bangladesh, for granting me study leave to undergo the PhD study.

I would like to express great appreciation to the most vulnerable riparian households in the study areas for their wholehearted participation in this study. The agriculture officers, NGO workers and community leaders in the study area also deserve much appreciation for extending their kind cooperation to conduct the study. The enumerators really done a wonderful job and rightly deserve much credit.

It is also my pleasure to express my gratitude that I have gained much knowledge through working with Professor Roger Lawrey, Head, School of Commerce, Professor Geoff Cockfield and Professor Julie Cotter, Director, Australian Centre for Sustainable Business and Development, as a research assistant for a range of projects. I also acknowledged the Deputy Vice-Chancellor (Research and Innovation) Professor Mark Harvey, Professor Barbara de la Harpe, Dean, Faculty of Business, Education, Law and Arts, and Professor Peter Terry, Director Research for providing me various assistance during my PhD period at USQ. My thanks also goes to the academic staff of USQ especially Professor Chandrasekhar Krishnamurti, Dr Mafiz Rahman, Dr Rasheda Khanam and Dr Shane Zhen. I would like to extend my special thanks to many friends whom valued accompany made my stay at USQ pleasant, namely, Dr SM Arefin Khan Mamun, Dr Shiduzzaman, Dr Enamul Kabir, Md Salauddin, Md Gaziuddin, Monzur Morshed, Wahid Ferdous, Md Mohsin, Michael and many others. Libby Collett also deserves special thanks for proof reading my thesis.

I have no words to express my gratitude to my family members. My mother Jamila Khatun, wife Nilufa Khatun Akhe, only son Mostakim Alam and my siblings sacrificed a lot and always remained a source of inspiration. I missed them greatly, and all credit for this PhD research therefore goes to them. My late father Kayem Uddin Ahmed who was my teacher and my idol first gave this dream to my mind. He would be the happiest man on about this accomplishment. Therefore, my thesis is dedicated to him. May Almighty Allah keep him in peace!

Finally, I am happy to express my gratitude to all of my teachers from primary to university level, friends, neighbors and all well-wishers for their love, support, encouragement and wishes which enable me to reach in this position. I owe much to my beloved beautiful country Bangladesh forever. Therefore, my commitment is to make all efforts to make them happy.

Warm-hearted thanks to all.

G M Monirul Alam 29 February 2016

ACRONYMS AND ABBREVIATIONS

AHM	Agricultural Household Model
BBS	Bangladesh Bureau of Statistics
BLR	Binomial Logit Regression
BMD	Bangladesh Meteorological Department
BRAC	Bangladesh Rural Advancement Committee
BRRI	Bangladesh Rice Research Institute
CEGIS	Centre for Environmental and Geographic Information Services
Char	Sandbar/island
CLP	Char Livelihood Program
CSI	Coping Strategy Index
CSIRO	Commonwealth Scientific and Industrial Research
CVI	Climate Vulnerability Index
DAE	Department of Agricultural Extension
DFID	Department of International Development
FAO	Food and Agriculture Organization
FGD	Focus Group Discussion
FSI	Food Security Index
GoB	Government of Bangladesh
GBM	The Ganges, the Brahmaputra, and the Meghna
ha	Hectare
HH	Household Head
HYV	High Yielding Varieties
IFAD	International Fund for Agricultural Development
IFPRI	International Food Policy Research Institute
IIA	Independence of Irrelevant Alternatives
IPCC	Intergovernmental Panel on Climate Change
IWFM	Institute of Water and Flood Management
Khas land	Governmental Land
LVI	Livelihood Vulnerability Index
MDGs	Millennium Development Goals

MNL	Multinomial Logit
MNP	Multinomial Probit
MOA	Ministry of Agriculture
MOF	Ministry of Food
NFPCSP	National Food Policy Capacity Strengthening Program
NGOs	Non-governmental Organisations
RCI	Resilience Capacity Index
RO	Research Objective
Riparian/Riverine	Riverbank and <i>char</i>
SLA	Sustainable Livelihood Approach
Taka (Tk)	Is the Bangladesh currency
UN	United Nations
UNDP	United Nations Development Program
UNEP	United Nations Environment Program
UNESCO	The United Nations Educational, Scientific and Cultural Organization
UNISDR	United Nations International Strategy for Disaster Reduction
USAID	United States Agency for International Development
USDA	United States Department of Agriculture
USQ	University of Southern Queensland
VGF	Vulnerable Group Feeding
VIF	Variance Inflation Factor
WB	World Bank
WDR	World Disaster Report
WFP	World Food Program
WHO	World Health Organization

TABLE OF CONTENTS

ABSTRA	A <i>CT</i>	i
CERTIF	FICATION OF DISSERTATION	iii
OUTCO	OMES OF THE THESIS	iv
ACKNO	DWLEDGEMENTS	vi
ACRON	YMS AND ABBREVIATIONS	viii
TABLE	OF CONTENTS	x
LIST OF	F FIGURES	xv
LIST OF	F TABLES	xvi
LIST OI	F APPENDICES	xvii
СНАРТ	FER ONE	1
Introduc	ction	1
1.1	Background and motivation for the study	1
1.2	Statement of the research problem	4
1.3	Specific objectives of the research	7
1.4	Research questions and approach	7
1.5	Conceptual framework of the study	9
1.6	Scope of the study	
1.7	Significance of the study	
1.8	Contribution of the study	
1.9	Organisation of the thesis	
СНАРТ	FER TWO	
Literatu	re Review	
2.1	Chapter outline	
2.2	Climate change and vulnerability	
2.3	Disasters in Bangladesh	
2.4	Riverbank erosion	
2.4.1	Riverbank erosion: A global perspective	
2.4.2	Riverbank erosion in Bangladesh	
2.5	Concepts of livelihood, vulnerability, resilience and adaptation	
2.5.1	Livelihood	
2.5.2	Vulnerability	
2.5.3	Resilience	

2.5.4	Adaptation	. 24
2.6	Food security issues	. 25
2.6.1	Food security concept	. 25
2.6.2	Impact of climate change on food security	. 26
2.6.3	Food security and self-sufficiency	. 27
2.6.4	Food security situation in the world	. 28
2.6.5	Food security studies in Bangladesh	. 29
2.6.6	Determinants of household food security	. 29
2.7	Measurement of food security	. 31
2.8	Gaps in the existing literature	. 32
2.9	Summary	. 33
СНАРТ	TER THREE	. 34
Methodo	blogy	. 34
3.1	Chapter outline	. 34
3.2	Theoretical and analytical framework	. 34
3.2.1	Household food security analysis	. 34
3.2.2	Analysis of adoption of adaptation strategies	. 37
3.2.3	Vulnerability analysis	. 39
3.2.4	Resilience analysis	. 40
3.3	Descriptive statistics	. 40
3.4	Description of the study areas and data collection	. 40
3.4.1	The approach	. 40
3.4.2	Selection of the study areas	. 41
3.4.3	Unit of analysis	. 41
3.4.4	Sample size	. 44
3.4.5	Sampling technique and non-response	. 44
3.4.6	Questionnaire and data collection	. 45
3.4.7	Validity and reliability	. 46
3.4.8	Data entry and cleaning	. 47
3.4.9	Secondary data	. 48
3.4.10	Research ethical issues	. 48
3.5	Data description	. 49
3.5.1	Socio-economic characteristics	. 49
3.5.2	Income inequality among the households	. 51
3.6	Research design: An overview	. 52
3.7	Summary	. 53

СНАРТ	TER FOUR	54
Vulnera	bility of Riverine Rural Households to Climate Change and Hazards	54
4.1	Chapter outline	54
4.2	Introduction	54
4.3	Data analysis method	56
4.3.1	Developing an index for vulnerability analysis	56
4.3.2	IPCC framework approach	63
4.3.3	Limitations	64
4.4	Discussion of the findings	64
4.4.1	Livelihood vulnerability index	64
4.4.2	Climate vulnerability index	68
4.4.3	Discussions	69
4.4.4	The vicious cycle of land loss and poverty	72
4.5	Summary of the chapter	74
		-
CHAPI	LEK FIVE	70
Kesinen	Chanten estline	70
5.1	United characteries	70
5.2 5.2	Introduction	/0
5.3		//
5.4	Data analysis method	80
5.4.1	Developing resilience index	80
5.5	Results and discussions	85
5.5.1	Resilience index	86
5.5.2	Discussions	90 02
5.5.5	Summary of the chapter	92
5.0	Summary of the chapter	95
CHAPT	TER SIX	95
Policy C	Options for Improving Access to Food for Vulnerable Rural Households	95
6.1	Chapter outline	95
6.2	Introduction	95
6.3	Data analysis method	97
6.3.1	Empirical model	98
6.3.2	Specification of the variables	100
6.3.3	Econometric consideration	101

6.4	Results and discussions	. 102
6.4.1	Livelihood conditions	. 102
6.4.2	Status of household food security and expenditure	. 103
6.4.3	Econometric results	. 104
6.5	The pathways	. 109
6.6	Sectorial impacts of riverbank erosion on food security	. 111
6.6.1	Impacts on various sectors	. 111
6.6.2	Impacts on farm sizes and household food security	. 112
6.7	Household food security index	. 113
6.8	Coping strategy index	. 115
6.9	Intra household food distribution	. 117
6.10	Vulnerability to food security	. 118
6.11	Summary of the chapter	. 119
СНАРТ	TER SEVEN	. 121
Climate	Change Perceptions and Local Adaptation Strategies of Hazard-Prone	101
Housen	DIds	. 121
/.1	Chapter outline	. 121
7.2	Introduction	. 121
7.3	Review of literature	. 122
7.4	Data analysis method	. 124
7.4.1	Data analysis	. 125
7.5	Results and discussions	. 125
7.5.1	Households' perception of climate change and variability	. 125
7.5.2	Perceived impacts of climate change and hazards	. 128
7.5.3	Households' adaptive responses	. 130
7.5.4	Access to information and adaptation	. 133
7.5.5	Access to and uses of ICT by the study households	. 134
7.6	Summary of the chapter	. 135
СНАРТ	TER FIGHT	137
Determi	nants of and Barriers to Adaptation by Resource-Poor Households	137
8 1	Chapter outline	137
0.1 Q 7	Introduction	127
0.2 9.2	Paviate of literature	120
0.0	Neview Of Interature	1.109
ð.4		. 140
8.4.1	Specification of variables	. 141

8.4.2	Model diagnosis	. 142			
8.5	Results and discussions	. 143			
8.5.1	Households' main adaptation strategies	. 143			
8.5.2	Barriers to adaptation	144			
8.5.3	Econometric results	145			
8.6	Summary of the chapter	151			
СНАРТ	TER NINE	. 153			
Conclus	ions and Recommendations	. 153			
9.1	Chapter outline	. 153			
9.2	Introduction	153			
9.3	Summary of findings	154			
9.1.1	Research objective 1	154			
9.1.2	Research objective 2	156			
9.1.3	Research objective 3	157			
9.4	Policy implications	159			
9.4.1	Public sector role	159			
9.4.2	Role of private sector and NGOs	160			
9.4.3	Community involvement	161			
9.4.4	Public-Private-NGOs partnership	161			
9.5	Contribution	162			
9.5.1	Contributions to the knowledge	162			
9.5.2	Methodological contribution	163			
9.5.3	Contribution to the theory	164			
9.6	Limitations of the study	164			
9.7	Future research directions	164			
9.8	Conclusions of the study	165			
REFER	ENCES	. 167			
APPEN	APPENDICES				

LIST OF FIGURES

Figure 1. 1: Bangladesh in South Asia
Figure 1. 2: Riverbank erosion-prone areas in Bangladesh
Figure 1. 3: Conceptual framework of the study10
Figure 1. 4: Structure of the thesis
Figure 3. 1: The study area: the Nagarpur and Chauhali Upazilas
Figure 3. 2: Jumuna river-bank erosion in the study areas and livelihoods
Figure 3. 3: Lorenz curve of income of the study households
Figure 3. 4: Research design 52
Figure 4. 1: Spider diagram of major components of the LVI
Figure 4. 2: Vulnerability triangle diagram of the dimensions of the CVI69
Figure 4. 3: The Livelihood Cycle driven by hazards in the study area
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment
Figure 5. 1: Conceptual framework for resilience assessment.79Figure 5. 2: Index value of the major components

LIST OF TABLES

Table 2. 1: Projected climate change scenarios for Bangladesh. 16
Table 2. 2: Intensity of climate change impacts on various sectors. 17
Table 2. 3: Development of vulnerability index. 23
Table 3. 1: Sample size for the survey. 44
Table 3. 2: Statistical techniques, purpose of use and software
Table 3. 3: Some selected socio-economic characteristic of the study households
(Household = HH)
Table 4. 1: Major components and sub-components comprising the Livelihood
Vulnerability Index (LVI) in the study areas
Table 4. 2: Indexed value of major components and sub-components comprising the
LVI (HHs = households)65
Table 4. 3: Indexed dimensions of climate vulnerability of <i>char</i> dwellers and
riverbank households
Table 5. 1: A components and indicators comprising the Resilience Capacity Index
(RCI) developed for the study areas (HHs = Households)
Table 5. 2: Indexed value of components and indicators comprising the resilience
capacity index
Table 6. 1: Summary statistics and description of model variables. 101
Table 6. 2: Household expenditure. 104
Table 6. 3: Regression results for the likelihood determinants of food security 106
Table 6. 4: Changes in farming status due to riverbank erosion. 113
Table 6. 5: Coping strategies close to food shortages. 117
Table 7. 1: Perception of climate change parameters
Table 7. 2: Perceived impacts of climatic change and hazards
Table 7. 3: Adaptation strategies of the households in the study area. 131
Table 8. 1: Summary statistics. 142
Table 8. 2: Perceived barriers to adaptation measures. 145
Table 8. 3: Estimated results from MNL model

LIST OF APPENDICES

Appendix 1: Types of disasters and their impacts in specific disaster pr	one areas. 210
Appendix 2: Questionnaire	

CHAPTER ONE

Introduction

1.1 Background and motivation for the study

Bangladesh occupies an area of only 147,570 km² in South Asia, with a population density of about 936 persons/km² where more than 35% people live under the poverty line (BBS 2012) (see the map in Figure 1.1 for Bangladesh in South Asia). Bangladesh is widely regarded as one of the world's most vulnerable countries due to its low-lying deltaic geographical position with more than 230 rivers and vast waterways that lead to disaster-proneness, as well as serious socio-environmental concerns including over population¹, poverty, and low economic and technological capacity (WB 2013; GoB 2011; IPCC 2007). The country experiences frequent natural disasters, including extreme climatic hazards, often in the forms of floods, droughts, riverbank erosion, salinity intrusion, water logging, and cyclonic storm surges (see Appendix 1 for specific disaster-prone areas). This causes large-scale loss of life and damage to infrastructure and economic assets, and thus adversely impacts on food, water and energy security, lives and livelihoods of poor people, particularly in the southern coastal areas and in the bank of the large rivers (Jordan 2015; Penning-Rowsell et al. 2013; Thoms et al. 2013; Pouliotte et al. 2009; Huq & Ayers 2008; IPCC 2007; Choudhury et al. 2005).

Food security issues have been a key concern globally over the last four decades. In spite of a reasonable level of food availability in many countries in the world, a substantial number of the globe's 'food insecure' people are the rural poor, who account for around 70% of the world's total of undernourished people (FAO 2010). It is predicted that future climate-induced hazards like floods, cyclones, drought and erosion will have catastrophic consequences on agriculture and food security in many parts of the world, particularly in developing countries (IPCC 2014; FAO 2013b; WB 2013). In this, Bangladesh is not immune to the impacts of these future climate change issues, which the nation is battling, along with the challenge of achieving food security and eradicating poverty.

¹ About 156.6 million people in 2014 (BBS 2014).

Food security is of prime concern in Bangladesh, despite marked improvements in food production and the incidence of poverty since the country's independence in 1971. In the last decade, the country achieved marked GDP growth rate of around 6% despite global economic turmoil in recent years. The rate of poverty decreased from 62% in 1988 to 35% in 2011 (BBS 2012). In contrast, population growth rate reduced from 2.4% in 1970 to 1.47% in 2011 (BBS 2012). Production of rice, the main staple food, increased from 16 million tons in 1970 to more than 50 million tons in 2010 (more than three times) (FAO 2012). This indicates that the country is close to achieving self-sufficiency in food production.





Source: Adapted from http://www.southasiaMaps.php

Despite these successes, the country is regarded as one of the seven countries² where two-thirds of the 906 million undernourished people on the globe live (FAO 2010). A report by the USDA (2010) indicates that out of 156 million people in Bangladesh, 33 million were registered as 'food insecure' in 2010 and this is projected to be 37 million by 2020. It is mainly due to the lack of affordability of adequate food for many poor households.

There is a growing recognition that food availability at national level does not essentially assure food security at the household or at the individual level due to a lack of economic access to food by the poorest households (FAO 2012; MacFarquhar 2011; Harrigan 2008; Schmidhuber & Tubiello 2007; Cleaver 1993). For example, Hong Kong and Singapore are 'food secure' although they are not self-sufficient in food production (as in both of these areas, agriculture is non-existent). On the contrary, India is self-sufficient in production, however, a large part of its population is not 'food secure' (Reji 2013; Schmidhuber & Tubiello 2007). The important consideration for food security is whether or not the monetary and nonmonetary resources at the disposal of the population are sufficient to allow them access to adequate quantities of food (Barrett 2010; Schmidhuber & Tubiello 2007). This situation was exacerbated by the soaring price of food in 2008 that pushed many poor households into the vulnerable condition of food insecurity (Anriquez et al. 2013; Swinnen & Squicciarini 2012; FAO 2011; Barrett 2010). Scholars argued that household level food insecurity ultimately threatens food security at a national level, which is ultimately a threat to national security (Alam & Khatun 2012; Akinyele 2009; Bahiigwa 1999). Therefore, households' access to food and their level of vulnerability³ are becoming more crucial for food security analysis in the changing global market economy (Keating 2013; Quisumbing 2013; WFP 2009).

Given the severe climate-related hazards in the country, the crucial policy agenda for Bangladesh is to identify and to understand the levels of vulnerability and to develop possible adaptation strategies, particularly for marginalised riverine/

² In 2010, about 925 million people were undernourished globally of which 906 million (98%) were in developing countries. Two-thirds of these live in just seven countries, namely, Bangladesh, China, Democratic Republic of Congo, Ethiopia, India, Indonesia and Pakistan (FAO 2010).

³ Food security vulnerability is the probability of an acute decline in access to food or consumption (WFP 2002). This also refers people's propensity to fall, or stay, below a pre-determined food security threshold (Løvendal et al. 2004).

riparian⁴ communities (details follow in the next section), that could mitigate the effects of adverse climate change and hazards. For governments to target development programs and initiate appropriate social, economic and environmental policies; it is important that accurate information on livelihood vulnerability is available. Indeed, it has been argued that policy interventions would do little to affect poverty dynamics unless the context of household vulnerability is properly understood (Shah et al. 2013; Hahn et al. 2009; IPCC 2007).

In Bangladesh, a number of studies (e.g., Ahmed 2015; Lein 2010; Zaman 2007, 1991, 1989; Hutton & Haque 2004; Hossain 1993; Elahi et al. 1991; Elahi 1989; Haque 1988; Rogge & Haque 1987; Greenberg 1986; Hossain 1984) have been conducted on riverbank erosion. However, there has been a lack of in-depth empirical research on the impacts of riverbank erosion and other climate change issues on the livelihood vulnerability and food security of the riverine rural households and how they respond to such hazards in order to mitigate immediate livelihood and food insecurity conditions. Therefore, assessing the livelihood vulnerability impacts of riverbank erosion on riparian household food security, along with the appropriate response strategies, is of significant research agenda in Bangladesh.

From an international perspective, this study has unique value since the main focus of the Millennium Development Goals (MDGs) for development agencies is on achieving food security and improving livelihood among the vulnerable rural population (UN 2013). The present study is aimed at generating important policy inputs for a comprehensive understanding of livelihood vulnerability and food security of the riverbank erosion hazard-prone rural households in Bangladesh.

1.2 Statement of the research problem

World Disaster Report (2001) argues that annually one million people are distressed⁵ and at least nine thousand hectares of land are lost globally due to climatic hazards including coastal and riverbank erosions. Households in the costal and riverbank areas are more exposed to the impacts of climate change and hazards. Research indicates that a significant portion of households in many countries in the world are

⁴ Riverbank and *char*.

⁵ A situation in which people do not have enough money and/or food.

affected due to riverbank erosion such as India, Cambodia, Lao PDR, Thailand, Vietnam, Italy and Australia (Das et al. 2014; Hall & Bouapao 2010; Warner & Paterson 2008; Rinaldi 2003).

In Bangladesh, the coastal and riverine households are the most susceptible to the impacts of climate-driven hazards, including riverbank erosion (GoB 2010) and recent models of hydrological impacts of climate change in different climatic zones have shown this to be true across Asia (Eregno et al. 2013). In particular, the hazard of riverbank erosion is a common problem in Bangladesh which contributes to the loss of both physical and material endowments through loss of land, natural resources and employment opportunities of the riverine rural households and thus threatening their food security and livelihoods. Twenty districts out of 64 in the country are prone to the riverbank erosion (CEGIS 2012; GoB 2010); while another study asserted that some parts of 50 districts in Bangladesh are subject to riverbank erosion (Elahi et al. 1991) (see the map in Figure 1.2).

Moreover, resource-poor households in the riverine areas are more prone to the impacts of frequent floods and waterlogging due to their proximity to the river, which also increases their vulnerability. About 8,700 hectares (ha) of homestead and farming land are lost to riverbank erosion, which displaces approximately 200,000 people annually (GoB 2010). Scholars mentioned that the rapid changes in river courses destroy valuable agricultural land, rural settlements, markets and towns, and make the people destitute and landless (Ahmed 2015; Lein 2010; Zaman 2007; Hutton & Haque 2004).

A report by CEGIS (2012) shows that during the period from 1973 to 2011 at least 153,566 ha of cultivable land (1.80% of total land) including 50,339 ha of settlement was eroded by the three major rivers in Bangladesh - the Padma, Jamuna, and Ganges. The Jamuna river erosion rate was around 5,000 ha per year in the 1980s while in recent years the rate is around 2,000 ha per year (CEGIS 2012). The displaced and dispossessed people take shelter in open sky and/or on embankments built along the river. They are pushed into the most vulnerable condition of food insecurity and poverty. Riverine areas are considered to be the most vulnerable areas of persistent poverty in the country (IFAD 2013; Huq & Rabbani 2011).

Despite the fact that damage by riverbank erosion occurs slowly and gradually, however, it has long-term impacts that are irrecoverable (GoB 2010). The

key issue is that riverbank erosion diminishes the total area of arable land every year in a place where arable land is scarce: arable land is 0.05 ha/person in Bangladesh (WB 2015). This contributes to a shortage of aggregate food production and thereby negatively impacts upon the food and livelihood insecurity of the vulnerable riverine people. Erosion also often triggers displacement and intra-country migration in the country. The hazard of riverbank erosion is considered to be one of the important bottlenecks that is preventing Bangladesh from attaining its MDGs, particularly those of eradicating hunger and poverty (GoB 2011).



Figure 1. 2: Red colour shows the riverbank erosion-prone areas in Bangladesh Source: Bangladesh Water Development Board

To make an appropriate policy intervention towards improving food security and the livelihoods of rural people in Bangladesh, policy makers and donor agencies need to know where the vulnerable people are, and to determine what drivers are exacerbating their levels of vulnerability and food insecurity. Although food security is a global concern, empirical evidence on how hazards like riverbank erosion coupled with climate change issues influence rural households' food security and their vulnerability seem to be scarce in the extant literature. This study will contribute to the policy formulation in this field and to the development of more targeted interventions through providing relevant reliable information and thus providing the opportunity to improve the food security and livelihood conditions of the marginalised rural riverine households.

1.3 Specific objectives of the research

The aim of this study is to assess the livelihood vulnerability of the riverbank erosion hazard and its impact on rural households' food security and their coping and adaptation strategies in Bangladesh. The specific objectives are:

- i) To assess the livelihood vulnerability of the riparian households;
- ii) To assess the food security status of the respondents; and
- iii) To identify the response strategies of the vulnerable households in the face of riverbank erosion and other climate change issues.

1.4 Research questions and approach

The following research questions are set to address each of the Research Objective (RO):

Objective 1:

Research questions under this objective are sought as follows:

- (i) what are the main drivers of livelihood vulnerability of riparian households to climatic changes and hazards?;
- (ii) are households isolated from the mainland more vulnerable to climate change than other riparian households?; and

- (iii) does livelihood status serve as a driver of vulnerability for them? and
- (iv) what are the factors influencing their resilience capacity?

Objective 2:

The specific research questions under this objective are as follows:

- (i) what is the livelihood status of the riverine households?;
- (ii) what are the factors influencing households food insecurity?;
- (iii) which months the households experience more food shortage (the extent of food insecurity)?;
- (iv) what are their coping strategies to address the food shortage? and
- (v) what are the policy recommendations to improve food security of these hazards-prone rural households in a sustainable way in Bangladesh?

Objective 3:

The main research questions under this objective are sought as follows:

- (i) what are the perceptions of hazard-prone rural households to climate change and variability?;
- (ii) what are the perceived impacts of riverbank erosion and other climate change issues on the livelihoods of the households;
- (iii) what local adaptation strategies can the resource-poor households adopt to enhance their resilience?;
- (iv) what are the barriers to adaptation?; and
- (v) what are the determinants influencing adaptation strategies, especially the influence of institutional access and social capital of the resource-poor households?

The study uses both primary (cross-sectional survey) and secondary data. Building on the IPCC framework, this study takes a holistic approach to assessing the two key vulnerability assessments – the Livelihood Vulnerability Index (LVI) and Climate Vulnerability Index (CVI). The study also develops an indicator-based index for assessing resilience capacity of the households, and employs a range of tools and techniques for data analysis such as the binomial logit model for household food security analysis which is based on the theories of consumer demand and production, popularly known as the Agricultural Household Models. For the analysis of households' choice of adaptation strategies, a multinomial logit model is employed which is based on random utility theory. This study has adopted and extended those approaches in the context of riverine households in Bangladesh.

1.5 Conceptual framework of the study

The livelihood of the majority of rural people in Bangladesh depends on agriculture which is most impacted by climate change. Therefore, riverine rural households in Bangladesh are impacted through climate change issues in general and by the recurrent riverbank erosion hazard in particular. Forecasted climate change impacts may also influence the frequency of flooding (Douglas 2009; Ravi 2008) which escalate the erosion hazard along the rivers (Ahmed 2006; Warrick & Ahmad 1996). The hazard impacts are often referred to as direct or indirect, or first or second order (Kates et al. 1985). Riverbank erosion is contributing to the loss of land, homestead, ponds, crops, trees and other resources of the riverine households, annually (Figure 1. 3).

Loss of land and resources is resulted in reducing their income and employment opportunities. As shown in the Figure 1.3, homeless/displaced people usually take shelter in open space, *khas* land (government land), and embankments or at the homes of relatives. These circumstances make the livelihoods of riverine households vulnerable. Moreover, due to climate change, they are expected to face a projected increase in mean annual temperature, uncertainty in rainfall, likely reduction of cereal crop production, and surges in disease, pest and weed pressure on crops and livestock (Niang et al. 2014; Molua 2009). Due to the proximity to the rivers, they are also prone to frequent flooding and water logging which, together with erosion, create a most volatile environment for them.

Decreased food production and reduced employment opportunities together with fluctuations in food prices contribute to the reduced entitlement to food by the households, particularly for small farmers and landless labourers. Therefore, in a society where most of the households depend on agriculture as in the case of this study, food insecurity will be the main first-order observed impact. At this stage, the riverine households adopt various coping and adaptation strategies to deal with the hazards (Figure 1.3). These adaptation strategies can be farm level (autonomous) and/ or planned (government policy). It is, however, noted that not all households suffers from climate change and hazards in a uniform way, and therefore, their responses vary depending on their economic position as well as the political and social linkages involved (Paul 1998; Emel & Pett 1989). This means that differential livelihood options and resources for adaptation need to be taken into account to assess the livelihood vulnerability of the riverine households and their food security status.



Figure 1. 3: Conceptual framework of the study

This study assesses how the riverine rural households (spatial scale) respond to such hazards and other climate change issues to mitigate immediate livelihood and food insecurity conditions, that often trigger displacement and intra-country migration using cross-sectional survey data (temporal scale). Understanding the vulnerability of households and their response strategies will assist the policy makers in targeting local adaptation strategies which is considered the key to protect the food insecurity and improving livelihood of poor farmers (IPCC 2014; Lobell et al. 2008; Adger et al. 2005b).

1.6 Scope of the study

There are many vulnerable areas in Bangladesh, however, riverine areas are the most vulnerable (IFAD 2013). Therefore, this study focuses only on riverbank erosion hazards-prone households. The impact of riverbank erosion can occur both at macro (loss of infrastructure and other assets and their impact on food production and national budget) and micro level (loss of land and employment opportunities at household level). This study focuses only on micro level impact. On the other hand, though all components of food security are important but it is not possible to cover all the dimensions (e.g., availability, accessibility and utilisation) due to the time and resource constraints of this study. Maxwell et al. (2013) argued that within food security debate, food availability and accessibility are the most vital components of food security. Therefore, food availability and food accessibility components of the household food security are taken into consideration for the present study. Moreover, food accessibility is considered the key challenge for Bangladesh due to the prevalence of large poor people. The country, on the other hand, did not face greatly the problem of food shortage (availability of food) during the last decade except the global incidence of food crisis in 2008 which also touch Bangladesh.

1.7 Significance of the study

Bangladesh is predominately a riverine country where every year a large portion of rural households are impacted due to riverbank erosion and become vulnerable. Vulnerability and poverty are a related issue, since poor people have the limited resources and opportunities to reduce vulnerability. However, not being poor does not necessarily mean not being vulnerable, and vice versa (Schneiderbauer & Ehrlich 2006). Riverbank issues that affect all surrounding riverine people, are significantly challenging the national goal of eradicating poverty.

There is however, a lack of information on the degree of livelihood vulnerability of the riverine households needed for policy interventions. Therefore, a comprehensive understanding of the drivers of livelihood vulnerability and resilience, and the determining factors of food insecurity today and those will influence in future is crucial to improve marginalised riverine household food security and livelihood over time.

1.8 Contribution of the study

The findings of this study will provide valuable policy inputs towards improving food security and livelihood of the marginalised riparian communities in Bangladesh. This study customised and validated various theories and approaches in the context of riverine households. The analytical approaches developed in this study for assessing vulnerability, resilience, and household food security and adaptation strategies can be replicated in other countries having similar characteristics to Bangladesh due to the flexibility of the methods. The innovation of coping and adaptation strategies by the riverbank erosion prone-vulnerable households could provide new insights and can be shared for other hazards-prone regions in the world.

1.9 Organisation of the thesis

The thesis consists of nine chapters (Figure 1.4) including this introductory chapter. It is important to mention that all the result chapters are presented in a mini thesis format that contains the information of brief introduction and review of the literature, methodology, results and discussions and conclusions.

Chapter 2 presents the reviews of the existing literature globally in general, and Bangladesh, specifically. The first part synchronized the studies on riverbank erosion hazard and other climate change issues covering vulnerability, resilience and adaptation studies. Then studies on food security and livelihood issues are presented and justify the importance of the research through sketching the research gap.

Chapter 3 provides an account of the methodology employed in this study. The theories and procedure of primary (survey) and secondary data collection are discussed here. A brief description of the study areas are also presented in this chapter. The research design is presented here where the linkages of research objectives, theories and methods are shown.

Chapter 4 assesses the LVI and CVI of the riverine households due to riverbank erosion and other climate change issues. In this chapter a vicious cycle of land access, food security and poverty are also discussed.



Figure 1. 4: Structure of the thesis

Chapter 5 develops an indicator-based Resilience Capacity Index (RCI) to assess the current resilience activities of vulnerable riverine households resulting from their long-term knowledge, experience and practices.

Chapter 6 examines the effects of riverbank erosion on household food security. Employing a qualitative response model (logistic regression model), the study determine the main influencing factors of households' food security. The stability and robustness of the model is also checked. This chapter also focuses on the coping strategies during the time of food shortages employed/practiced by the households.

Chapter 7 documents households' perceptions about long-term climate change and variability, and perceived impacts of riverbank erosion and other climate change issues on their livelihoods. It also provides information on local adaptation strategies.

Chapter 8 assesses the determinants of households' adaptation choices. A qualitative response model (multinomial logit model) is applied for examining the main factors influencing the households' adaptation strategies, particularly the influence of institutional access and social capital.

Finally, Chapter 9 provides conclusions of the results of the study and suggests policy recommendations and an agenda for future research.

CHAPTER TWO

Literature Review

2.1 Chapter outline

Reviewing the literature is crucially imperative for any research in the sense that it helps to identify prevailing knowledge gaps and assists in developing a sound research design and methodology for carrying out the study, and relates the research findings with past experiences. Keeping this in mind, a thorough review of related literature is performed and presented in this chapter under the following sub-headings: Section 2.2 presents climate change and vulnerability in general. Section 2.3 discusses the disasters in Bangladesh. Sections 2.4 presents the problem of riverbank erosion globally. Section 2.5 presents the concept and assessment techniques of vulnerability, resilience and adaptation; then food security issues are discussed in Section 2.6. Food security measurement techniques are presented in Section 2.7. The gaps in the literature are discussed in Section 2.8 and finally, Section 2.9 contains the summary of this chapter.

2.2 Climate change and vulnerability

The issues of climate change and hazards are ongoing part of human history. However, poor people in developing countries whose subsistence livelihood depend upon the utilisation of natural resources are the first and most affected by the climate change which increases their vulnerability (IPCC 2014; Bardsley & Wiseman 2012; McDowell et al. 2013; Salick et al. 2009; Thomas & Twyman 2005). Scholars argued that climate change can be manifest in four main ways: (i) slow change in mean climate conditions, (ii) increased inter annual and seasonal variability, (iii) increased frequency of extreme events, and (iv) rapid climate changes causing catastrophic shifts in ecosystems (Tompkins & Adger 2004). Nonetheless the impacts of climate change have two dimensions: spatial and temporal. The underline meaning of spatial dimension is that the effects of climate change are heterogeneous, and region and location specific. For example, the issue of climate change – the raising temperature with reduced or abnormal rainfall – has already impacted the natural and physical ecosystem of Bangladesh. The northwest part of the country is

impacted with its recurrent drought and the southwest part through raising soil salinity (Ahsan et al. 2011). It is projected that future climate change will badly impact on agriculture in Bangladesh and thus food security and livelihood of the majority of the population (GoB 2010; IPCC 2007). Projections of the climate change scenarios for Bangladesh are presented in Table 2.1. The table indicates that there will be higher seasonal variations in temperature and reduced rainfall in the coming decades, which might have serious consequences on agricultural production in Bangladesh.

Year	Mean temperature change (°C)			Mean rainfall change (%)		
	(standard deviation)			(standard deviation)		
	Annual DJF JJA		Annual	DJF	JJA	
2030	1.0(0.11)	1.0(0.18)	0.8(0.16)	+3.8(2.30)	-1.2(12.56)	+4.7(3.17)
2050	1.4(0.16)	1.6(0.26)	1.1(0.23)	+5.6(3.33)	-1.7(18.15)	+6.8(4.58)
2100	2.4(0.28)	2.7(0.46)	1.9(0.40)	+9.7(5.80)	-3.0(31.60)	+11.8(7.97)

Table 2. 1: Projected climate change scenarios for Bangladesh.

Source: Adopted from Agarwala et al. (2003)

DJF = December, January and February; JJA = June, July and August

The temporal dimension, on the other hand, refers to the timeframe over which climate change effects are considered. Therefore, vulnerability and associated response strategies need to be assessed along both spatial and temporal scales. Despite the fact that Bangladesh is a hazard-prone country, nevertheless there is limited research focusing on vulnerability. A few studies in the past have been carried out in coping and adaptation mechanisms in coastal Bangladesh mainly focused on hazard warning and evacuation systems (Paul 2010), health security due to disasters (Ray-Bennett et al. 2010), and coastal hazards and community-coping methods (Parvin & Shaw 2012; Alam & Collins 2010; Parvin et al. 2008). Some also focused on the vulnerability issues confronting the coastal and drought prone areas in Bangladesh (Ahsan & Warner 2014; Toufique & Islam 2014). There are hardly any studies that focused on index-based livelihood vulnerability measurement especially for riverbank erosion hazard-prone households in Bangladesh. This assessment is important because vulnerability is context-specific and differs from area to area (Hahn et al. 2009). Therefore, this study assesses location specific

vulnerability (spatial scale) of riverine households using cross-sectional survey data (temporal scale).

2.3 Disasters in Bangladesh

Before going into a discussion on riverbank erosion hazards, this section presents other types of disasters which the nation of Bangladesh has often experienced. In Bangladesh, about 10% of the area is hardly 1m above to the sea level (Huq et al. 1995). The most critical impacts associated with climate change in the country are: (i) drainage congestion (flooding); (ii) reduced fresh water availability; (iii) disturbance of morphological processes (erosion); and (iv) an increased intensity of disasters (Ahmed 2005; WB 2000; Huq et al. 1998). Table 2.2 presents how different sectors are heterogeneously exposed to these climatic hazards in Bangladesh. The table indicates that the most vulnerable sector is crop agriculture on which livelihood of most rural people depends on. Due to erosion, human settlement and infrastructure are also severely affected.

	Vulnerability context (climate change and climate events)							
Sectors	Extreme	Drought	Flood		Cyclone Sea level rise		se	Erosion
	temperature		River	Flash	& storm	Coastal	Salinity	
			flood	Flood	surges	inundation	intrusion	
Crop	***	***	*	**	***	**	***	-
agriculture								
Livestock	**	**	*	**	***	**	***	-
Fisheries	**	-	**	*	*	*	*	-
Infrastructure	*	-	**	*	*	**	-	***
Industries	**	-	**	*	*	***	**	-
Biodiversity	**	-	**	-	*	***	***	-
Health	***	-	**	-	**	*	***	-
Human	-	-	-	-	***	-	-	***
settlement								
Energy	**	-	*	-	-	*	-	-

Table 2. 2: Intensity of climate change impacts on various sectors.

Source: MOEF, 2005

Notes: ***= Severely vulnerable; **= Moderately vulnerable; *= Vulnerable; -= Not vulnerable

Despite the fact that Bangladesh constitutes only about 7% of the area of the combined catchments of three major eastern Himalayan rivers: the Ganges, the Brahmaputra, and the Meghna (GBM). However, the country drains over 92% of the total annual flow of this GBM system in the region, which is one of the main causes
of erosion in Bangladesh itself. Moreover, Bangladesh has a monsoonal climate that creates frequent, heavy rainfall. It is reported that monsoonal rainfall would increase about 10–15% by 2030 (Jakobsen 2005). The country is frequently faced with overbank spillages and floods, particularly along the major rivers when monsoon-driven excessive runoff in these rivers combines with local rainfalls. It is predicted that the stronger monsoon rainfall would aggravate flood conditions while catastrophic flood events may occur with higher frequency in Bangladesh (Huq et al. 1996). Increased monsoonal flows will result in an increased sediment transport capacity and morphologic dynamics of the rivers which lead to increase riverbank erosion along the GBM rivers (Warrick & Ahmad 1996; Huq et al. 1998; Ahmed 2006).

In dry season, on the other hand, river flows would further decrease, leading to increase water shortages all over the country and salinisation in the coastal areas (Ahmed 2006; CEGIS 2006; Halcrow et al. 2001; Huq et al. 1996). Reduced winter fresh water flows might aggravate the draw-down of shallow aquifer systems, reducing its potential for drinking and irrigation water, particularly in the western part of the country. Cyclones will be stronger and more frequent due to increases in sea surface temperatures due to warming. Droughts affect many parts of Bangladesh, particularly in the western districts. The projected temperature increase of 2°C and the estimated decrease of 10% of rainfall would result in a decrease in food security (Shahid & Behrawan 2008; FAO 2006). Drought affects rice production heavily in Bangladesh. Yield reduction of different crops varies from 10% to 70% depending on the intensity of drought (Karim 1990). The coastal zone is also affected frequently by cyclones, tidal surges, floods which cause dislocation of households, particularly from the smaller islands (Thoms et al. 2013; Pouliotte et al. 2009).

2.4 Riverbank erosion

In this section research on riverbank erosion is briefly summarised below.

2.4.1 Riverbank erosion: A global perspective

A significant portion of population in the world is likely to be displaced because of climate induced natural hazards including devastating floods, tropical cyclones, coastal and riverbank erosions, as well as storm and sea water surges (IPCC 2007; WDR 2001). The impacts of land loss due to riverbank erosion is permanent and has

long-term consequences on the economy, whereas the impacts of other hazards is temporary.

In terms of the magnitude of devastation of erosion, the Mississippi-Missouri River System of North America, Ganges and Brahmaputra of Bangladesh and India, Mekong Rivers of Asia, Amazon River of South America, and River Nile of Africa are the most prominent (Das et al. 2014). Riverbank erosion is a natural hazard affecting many countries in the world. As for example, Rinaldi (2003) mentioned riverbank erosion as a problem for central Italy. Erosion of Danube river, the second longest river in Europe, creating problems for many European countries (Jones et al. 2007). Warner and Paterson (2008) asserted that flooding and riverbank erosion are the major hazards for people living on or near flood plains of the coastal rivers of New South Wales, Australia. Hall and Bouapao (2010) argued that the Mekong riverbank's erosion had had great impact on the livelihood and food security of the riverine people covering four countries, namely, Cambodia, Lao PDR, Thailand and Vietnam.

2.4.2 Riverbank erosion in Bangladesh

The pattern and severity of riverbank erosion in Bangladesh are unique. Being a densely populated country, a large portion of population live along the bank of rivers (around 230 rivers in total). Moreover, the country drains a huge amount of runoff due to its geomorphological position. The riverbank erosion is a recurring issue in Bangladesh. It has impacted on physical, economic, social and political conditions causing catastrophic impacts on lives and livelihood of the riverine households (Ahmed 2015; Lein 2010; Zaman 2007, 1991, 1989; Hutton & Haque 2004; Chowdury 2000; Hossain 1993, Elahi 1989; Rogge & Haque 1987; Greenberg 1986). Riverbank problem is severe among the landless and impoverished farmers who have least capacity to resist and recover from such natural hazards (Rogge & Elahi 1989; Greenberg 1986). Hossain (1993) estimated that over a 10-year period, the village Bhola lost almost 16 ha of farm land, about 20% of its cultivatable land, and 45% of the households were affected in one way or another due to the bank erosion of the river Kalingonga.

Akhter (1984) asserted that about one quarter of slum dwellers migrated from rural areas to Dhaka, the capital city, because they were uprooted by erosion disasters. Hossain (1984) in a study of Kazipur rural sub-district recorded that about one-tenth of the riverbank erosion induced displaces moved to urban areas in their attempts to re-establish a livelihood. Haque (1988) found that 43.5% of the displaced people moved family, 9.3% livestock, and 15.5% shifted their belongings from erosion affected areas to comparatively safer places. The riverbank displaces are subject to different hazards notably lack of adequate housing and health facilities, shortage of drinking water and sanitation facilities (Hutton & Haque 2004; Chowdury 2000; Elahi 1989; Rogge & Haque 1987; Greenberg 1986; Islam & Islam 1985). Riverbank erosion affects all the surrounding bank people through eroding land and destroying employment facilities, which the land could provide with them (Romanowski 1987). It is predicted that sea level rise will increase morphological activities in the river, inducing increased river flow which ultimately accelerated river flow and thus will increase river bank erosion too (Alam 2003).

In Bangladesh, riverbank erosion is the topmost disaster concerning the losses (Penning-Rowsell et al. 2013; Makenro 2000). Akter (2009) revealed that on an average 25%, 3% and 2% populations were displaced from different natural calamities like floods and riverbank erosion, droughts and cyclones, respectively in Bangladesh. According to Zaman (2007), environmental disasters were creating acute problems of unemployment in rural areas, urban slums and thereby, worsening the socio-economic conditions of the displaced people. It is therefore clear from the above discussion that riverbank erosion in Bangladesh causing the loss of household production-based entitlement through the reduction of total farm land and, own-labour based entitlement through reducing employment opportunities. This issue deserves the focus of greater research in terms of the future household food security in Bangladesh.

2.5 Concepts of livelihood, vulnerability, resilience and adaptation

In this section the concept of livelihood, vulnerability, resilience and adaptation are discussed.

2.5.1 Livelihood

Households and individuals employ a wide range of activities and invest their limited resources as part of their strategy for improving their livelihood. A person's livelihood is the combination of a range of farm and non-farm activities that together provide a variety of procurement strategies for food and cash (Drinkater & Rusinow 1999). Livelihoods encompass the assets, activities and the access to these that collectively determine the living gained by an individual of household (Ellis 2000). The issues of improving and sustaining livelihood are considered to be synonymous with poverty alleviation (DFID 1999). Therefore, rural livelihood improvement strategies and policy interventions must be focused on the context in which households operate and create an environment that enables them to improve their conditions. In this study, the vulnerability of households is assessed under a livelihood framework. The livelihood framework presents the main factors that affect people's livelihoods and the typical inter-relationship between these factors. These include human capital, social capital, natural capital, physical capital and the financial capital (DFID 1999).

2.5.2 Vulnerability

There are a cluster of concepts that have developed around the notion of vulnerability over the years. The Intergovernmental Panel on Climate Change (IPCC) defines the term vulnerability as: the degree to which a system is susceptible to, and unable to cope with, adverse effects of climate change and variability and extremes (IPCC 2007). Various scholars define vulnerability in different ways. For example, Adger (1999) defines vulnerability as the exposure of a group or individual to stress due to socio-environmental change that disrupts livelihoods. United Nations International Strategy for Disaster Reduction (UNISDR) defines vulnerability as the conditions determined by physical, social, economic, and environmental factors or processes that increase the susceptibility of a community to the impacts of hazards (UNISDR 2004). All communities, however, are not equally vulnerable, and there are differences both spatially and temporally even within the same community (Fraser et al. 2011; Fussel 2010). Past research has revealed that vulnerability depends on a range of factors such as the distance to the city or district headquarters (Pandey & Jha 2012), wealth and well-being and class of households (Gentle et al. 2014; Macchi et al. 2014), gender of households (Arora-Jonsson 2011; Denton 2002; Gentle et al. 2014), and dependency on natural resources and livelihood options (Thomas & Twyman 2005). Scholars have also argued that vulnerability is influenced not only by internal factors but also by external factors such as government policy (Yeh et al. 2013; Dougill et al. 2010) which makes the issue complex. Therefore, the challenge for vulnerability assessment research is to develop robust and credible measures (Adger 2006).

In recent years, vulnerability research development and consequent adaptation policy have gained top priority (Hinkel 2011). The importance of vulnerability and adaptive capacity has been frequently cited in order to explaining the societal aspects of climate change (Fussel & Klein 2006). There have been a numbers of research endeavours that have tried to assess the vulnerability and adaptive capacity of communities through the development of indices (see, for example, Hahn et al. 2009; Gbetibouo et al. 2010; Pandy & Jha 2012). All of these studies have encountered conceptual and data-related problems while selecting and aggregating relevant indicators in the respective indices. The typical feature of the concept of vulnerability measurement is the level or scale of analysis which ends with an index construction. An index generally deals with the aggregation of a series of observable contributing variables into a scalar variable (Hinkel 2011). Constructions of such indices distinguish between two major ontological approaches: data-driven and theory-driven approaches (Vincent 2007). Selection and aggregation of contributing indicators in the data-driven approach are based on expert judgment and efforts are then made to correlate these with records of previous disasters (Alberini et al. 2006; Brooks et al. 2005). However, the weakness in this approach lies with the limited objectivity of experts and assessment of contributing indicators against a bench-mark of vulnerability (Below et al. 2012).

A theory-driven approach, on the other hand, applies insights from the literature to select and aggregate contributing indicators (Vincent 2007). However, the weakness in normative selection of contributing indicators remains with the associated uncertainties (Below et al. 2012). Considering the limitations, this third group of researchers adopts both theoretical and empirical aspects to select and aggregate the contributing indicators in constructing the index (Hahn et al. 2009; Pandey & Jha 2012). However, the conceptual work on vulnerability and its related theme has yet not resolved the methodological and terminological confusion (Hinkel 2011). Over the years, various researchers have tried to develop vulnerability indices

addressing different set of parameters and their contributing indicators/subcomponents which are presented in Table 2.3.

Author(s) name and year	Name of index	Comments	
Turner et al. (2003)	Vulnerability Framework	Based on this framework many researcher developed vulnerability index	
Cutter et al. (2003)	Social vulnerability index (SoVI)	Employed principal components analysis in country level socio- economic and demographic data. Some important variables related to exposure to natural hazard were ignored due to data structure	
Vincent (2004)	Social vulnerability index (SVI)	Different weights were used for different sub-components in multi- country analysis. There was missing data problem due to usage of secondary data.	
Hahn et al. (2009)	Livelihood vulnerability index (LVI)	Problem of secondary data set was removed and diversified components were considered for vulnerability. Flexibility to consider indicators and hence applicable to any area.	
Vincent and Cull (2010)	Household social vulnerability index (HSVI)	Five composite sub-indices was used and assigned equal weight. Vulnerability might not affect equally.	
Pandey and Jha (2012)	Climate vulnerability index (CVI)	Primary data set was used. Useful tool for assessing scale differences in vulnerability.	
Ge et al. (2013)	Social vulnerability index (SVI)	Economic variables (GDP, PCI) were used to assess hazards loss. Absence of exposure indicator(s)	
Lee (2014)	Social vulnerability	All indicators showed positive direction to vulnerability. Zero-mean standardized the indicator values.	

 Table 2. 3: Development of vulnerability index.

The social vulnerability index developed by Cutter et al. (2003) and Vincent (2004) has been used by many researchers in different context. However, the method is associated with secondary data driven problem. Later, Hahn et al. (2009) developed an indicator based livelihood vulnerability index which is free from such problem. In this study, indicator based vulnerability analysis approach is adopted

which includes both climatic and non-climatic (socio-economic) indicators (see detailed in Chapter 4).

2.5.3 Resilience

Human populations are concentrated along the bank of rivers and coastal areas which make them more susceptible to the impacts of climatic hazards. The main focus of climate change research is to enhance resilience of such disaster-prone communities. The concept of resilience is aimed at sustaining and enhancing the capacity of social-ecological systems to adapt to uncertainty and surprise (Adger et al. 2005a). It reflects the degree to which a complex adaptive system is capable of self-organisation and the degree to which the system can build capacity for learning and adaptation (Carpenter et al. 2012; Folke et al. 2002). However, there is a lack of standard framework to effectively measure resilience capacity of households (Magis 2010; Manyena 2006). This study attempts to measure the resilience capacity of the households through developing an indicator based index (see detailed in Chapter 5). The findings will enable policy makers to understand the factors that limit the resilience capacity of hazard-prone households and thus effective policy and programs can be formulated.

2.5.4 Adaptation

Globally, mitigation and adaptation are the two major policy responses to climate change. Adaptation has the potential to reduce the adverse impacts of climate change (IPCC 2001). IPCC defines adaptation as the adjustment in human or natural systems in response to climatic or environmental stimuli which buffer harm or exploit beneficial opportunities (IPCC 2001). However, adaptation strategies vary from sector to sector, community to community and place to place (Malone 2009; Smit & Wandel 2006). Scholars argued that all adaptation is not good (Eriksen et al. 2011; Nyong et al. 2007). For example, the adaptation measures that deliver short-term gains and economic benefit can lead to increased vulnerability in the medium or long run (Jones & Boyd 2011). According to Smith et al. (2000), to fully understand adaptation, it is important to know three fundamentals of adaptation such as: adaptation to what, who adapt and how adaptation occurs?

There are few adaptation studies in Bangladesh that mainly focus on drought prone areas in Bangladesh (see, for example, Alam 2015; Alauddin & Sarker 2014; Sarker et al. 2013; Habiba et al. 2012). Few studies focus on low lying and salineprone areas (Rashid et al. 2014; Anik & Khan 2012). Though these studies provide important policy input, it may not be applicable in other hazards-prone areas due to the variation in socio-economic and climatic conditions. Despite recognition of the need for adaptation, so far there is no study that explores adaptation strategies and the influence of various determinants such as social capital and access to institutional facilities on adaptation decisions for marginalised riparian households which is crucial to making proper climate adaptation policies in the country (see more discussions on Chapter 7 and Chapter 8).

2.6 Food security issues

In this section the food security issues are discussed.

2.6.1 Food security concept

Keating (2013) mentioned that there had been significant revision of the concept of food security in the last 40 years by different scientists and organisations. Hoddinott (1999) reported about 200 definitions and 450 indicators of food security. World Food Summit (1996) defines food security as: "Food security exists when all people, at all times, have physical, social and economic access to sufficient, safe and nutritious food that meets their dietary needs and food preferences for an active and healthy life" (FAO 1996). Household food security is thus referred to as the application of this concept to the family level where individuals within households are the focus of concern. This definition focuses on three⁶ distinct but interrelated elements as follows:

Food availability: This refers to having sufficient quantities of food from household production, other domestic output, commercial imports, or food assistance.

⁶ Barrett (2010) argued that stability of food refers the availability and access to food.

Food access: This means households having adequate resources to obtain appropriate food for a nutritious diet, which depends in turn on available income, distribution of income in the household, and food prices.

Food utilisation: This element refers to the proper biological use of food that require a diet with sufficient energy and essential nutrients; potable water and adequate sanitation; and knowledge of food storage, processing, basic nutrition, and child care and illness management.

Although this definition mostly refers to 'food', the main concern is with calories (Heald & Lipton 1984). Other issues such as protein, micro-nutrients or more generally, food quality and safety are not considered because when caloric intake is satisfactory, then other needs are usually satisfied (Maxwell & Smith 2006).

Nutrition security, on the other hand, refers to situations in which food security is coupled with a sanitary environment, adequate health services, and proper care and feeding practices to ensure a healthy life for all household members (Horton & Shekar 2010; UNSCN 2010).

2.6.2 Impact of climate change on food security

Climate change has differential impacts on different localities and communities across the planet (Maru et al. 2014). Scholars mentioned that global food distribution system might be affected greatly due to climate change (Poppy et al. 2014; Molua 2009; Schmidhuber & Tubiello 2007; Gregory et al. 2005; Smith et al. 2000). Extreme climate events such as droughts, floods and cyclones severely affects food supplies and thereby food security. Gregory et al. (2005) argued that climate change affects all dimensions of food security directly. However, the impact of climate change on food security varies across regions and over time which depend on country's socio-economic status (Gregory et al. 2005; Stern 2006). Reports indicate that due to climate change issues, achieving/meeting household food demand in the densely deltaic countries in South Asia including Bangladesh will be at risk (Szabo et al. 2015; FAO 2013a). Being a developing country, ensuring household food security is a big challenge for Bangladesh when facing other problems such as climate change, escalating population and poverty. Within the country, food security issues

become acute for the riverine households since they are confronting riverbank erosion and other negative impacts of climate change.

2.6.3 Food security and self-sufficiency

The issue of food security came into focus during the world food crisis of 1972-74 and received momentum after the World Food Conference in 1974 where the FAO Committee on World Food Security were established and a universal declaration on the eradication of hunger and malnutrition was adopted. In the 1970s, food security was considered to be an issue of national and world food supplies, and import stabilisation policies (Maxwell & Salter 2003; Stringer 2000). In the 1980s, the publication of Sen's (1981) influential theory on entitlement (all legal sources of food)⁷ influenced food security discussions heavily. He mentioned that famine can occur even if supplies of food are adequate and markets function well. This shifted the focus of the research from the supply side to the demand side issues of food security. In the late 1990s, issues of both availability and stable access to food were also incorporated in the definition (Frankenberger & McCaston 1998).

Food insecurity therefore refers to the lack of food security and also to a state in which households do not have adequate physical, social or economic access to food for an active and healthy life. Schmidhuber and Tubiello (2007) argued that food availability is not the crucial issue of food security in the present world because food could be traded at reasonably low cost. FAO (2010) argued that the most recent increase in hunger was not the consequence of poor global harvests but was caused by the world economic crisis that had resulted in lower incomes and increased unemployment. Adequate access to food is the key to household food security which can be achieved without households being self-sufficient in food production if they have the ability to generate sufficient income or have own production which together can be used to meet food needs (Harrigan 2008; Schmidhuber & Tubiello 2007).

Schmidhuber and Tubiello (2007) asserted that national self-sufficiency is neither necessary nor sufficient to guarantee food security for all of the population, and they cited the example of Hong Kong and Singapore (food-secure without self-

⁷ Sen (1981) talked about four types of entitlement to get access to food: 'production-based entitlement' (growing food), 'trade-based entitlement' (buying food), 'own-labour entitlement' (working for food) and 'inheritance and transfer entitlement' (being given food by others).

sufficiency in food production), and India (self-sufficient but much of the population has food insecurity). Aggregate supply of food (i.e. own production plus import) may be a necessary condition but it is certainly not a sufficient condition to be a food secure country (Barrett 2010; Harrigan 2008; Titus & Adetokunbo 2007). Therefore, households or individuals are the appropriate unit of food security analysis (Sen 1981; Dreze & Sen 1989; Ravallion 1997). Scholars, however, argued that individual access to food is associated with households' income and resources (Maxwell 1994; Evans 1991). Households in the developing world also had experienced varying degrees of food insecurity. Anriquez et al. (2013) and MacFarquhar (2011) mentioned that soaring food prices since 2008 had impacted badly upon low income households both in developed and developing countries. This also dramatically changed the focus of food security research by the international organisations that incorporate the vulnerability issue as the backdrop to climate change issues (Quisumbing 2013; Hardley et al. 2011; Lang & Barling 2012; Ilbery & Maye 2010). Therefore, this study assesses the vulnerability of households and food security as well.

2.6.4 Food security situation in the world

Food is a basic human right (FAO 2010). Household food security varies across regions, agro-ecological zones and districts (Bahiigwa 1999). Countries like the USA, Canada, and Australia had also faced food security problems due to recent global soaring food price. Among the victims, female-headed households were more vulnerable to food insecurity than other types of households (Quisumbing 2013). In the USA, about 14% of households were food insecure in 2008 and about 14.9% in 2012 (Nord et al. 2008; Jensen et al. 2012). In Canada, 10% of households were food-insecure in 2002 (Che & Chen 2002) and more than 12% in 2011 (Tarasuk et al. 2013). In the case of Australia, more than 5% of the population was food insecure in 2001 (Booth & Smith 2001) and about 4.8% in 2011 (Lockie & Pietsch 2011). FAO (2010) mentioned that the food security situation was not improving in developing countries as expected. Brown et al. (2008) mentioned that the food security situation has improved a little more in the regions like East Asia, South Asia and Latin America as compared to Africa. Brown et al. (2008) stated that food inflation (i.e. increase in food prices) had struck the poor households of Vietnam and India. Krugman (2011) argued that higher prices of cereals impacted

terribly on poor people in developing countries because they spend a major proportion of their income on basic foodstuffs.

The above discussion supports the notion that access to food is the key issue for household food security. Results of the previous studies on the factors affecting household food security both in developed and developing countries are presented in Section 2.6.6.

2.6.5 Food security studies in Bangladesh

The twin objective of Bangladesh after independence in 1971 was to increase food production and eradicate poverty (GoB 2000). Due emphasis has been placed on augmenting food production and thereby ensure food security of the mass population in the country. A significant number of research have been carried out on food security, funded by the donors, particularly the WB, USAID, FAO and IFPRI (Ahmed et al. 2004 & 2007; Murgai & Zaidi 2005; WB 2006). Few studies are highly dominated by the field of economics which provides an economic explanation of food security in Bangladesh (see, for instance, Mishra et al. 2015; Islam 2014; Chowdhury et al. 2010; Faridi & Wadood 2010; Hossain 2010; Shahabuddin 2010; Deb et al. 2009; Hossain & Deb 2009; Ali et al. 2008; Ahmed et al. 2007). However, food security at household level is relatively unexplored, particularly for riverine households when they are confronting the problem of land loss and other climate change issues.

2.6.6 Determinants of household food security

There are many reasons to believe that incidences of food insecurity vary between the rural households that are affected by riverbank erosion. This study can thus hypothesise that households are not equally vulnerable to food insecurity and its different forms. Their socio-economic variables along with other variables play vital role here. Keeping these assumptions in mind, this study attempts to find out which household and community level characteristics can influence food insecurity and its different forms. In this section findings of the previous studies on the determinants of household food security are summarised:

Age of the household head was found to have a significant impact on the determinants of household food security by different studies. A study by Bashir et al. (2010) in Pakistan found that the presence of a household head aged over 35 years reduced the chances for food security by 83%. They employed the multinomial logistic regression model (MLR) for their analysis. Titus and Adetokubo (2007) in their study in Nigeria using descriptive statistics revealed that increases in the age of the household head contributed positively to food insecurity and the level was the highest for the age group of 61-70. In the USA, Onianwa and Wheelock (2006) applying binomial logistic regression model (BLR) found that the increase in household heads' age reduced the chances of becoming food secure by 2%. Family size was found to be an important factor in determining household food security by different studies (Bashir et al. 2010; Sindhuet et al. 2008; Amazaet et al. 2006). Households with large numbers of family members were likely to be more food insecure than households with small numbers of family members in Bangladesh (Alam et al. 2010). In India, Sindhu et al. (2008) revealed that one additional family member would increase the chance to be food insecure by 96%. In Nigeria, an increase of one family member would decrease the probability of food security by 1.5% (Amaza et al. 2006). Bashir et al. (2010) found that households with a nuclear (husband and wife) family system were 5 times more food insecure than households with joint family system in Pakistan. Household heads who have education levels of eight years of schooling and those who had attained graduation level increased the probabilities for their families to become food secure by 6% and 20%, respectively (Bashir et al. 2010).

Ojogho (2009) revealed by employing BLR that probability of food insecurity would decrease with the increase of education level from primary to secondary and to tertiary level by 78% and 92%, respectively, in Nigeria. Mariara et al. (2006) by using simple regression found that the level of education of the mother in the household would increase the likelihood to become food secure by 0.05 % in Kenya.

In the USA, Kaiser et al. (2003) applying BLR revealed that chances of food insecurity would decrease by 29% with the higher level of education of mother in the family. Increases in household income contributed positively towards household food security. Households' monthly income increase by 1000 Rupee contributes to

reduce food insecurity by 30% in India (Sindhu et al. 2008). Increase in households' annual income, both with and without children, would reduce the chances of becoming food insecure by 6% and 5%, respectively (Onianwa & Wheelock 2006). Che and Chen (2002) applying BLR revealed that upper middle income households were 7.29 times less food insecure than households with lower income in Canada. Alam et al. (2010) employing BLR found that probability of household food security would increase by 0.03% for a one hundred decimal increase in farm size in Bangladesh. They also revealed that the probability of household food security would increase by 0.35% with the increase of household off-farm income (income from outside agriculture) of Tk⁸ 100 thousand per year (see more on Bangladesh context in the Chapter 6). Households with no milking cow in Pakistan (Bashir et al. 2010). Amaza et al. (2006) in their study by applying BLR revealed that households having bullock would increase the likelihoods of becoming food secure by 5% in Ethiopia (see more in Chapter 6).

2.7 Measurement of food security

There are various methods or indicators for food security measurement that differ significantly (Carletto et al. 2013; Maxwell et al. 2013; Perez-Escamilla & Segall-Correa 2008). Scholars argued that a variety of measurement is required for its analysis due to the multidimensional nature of food security (Carletto et al. 2013; FAO 2013b; Maxwell et al. 2013; Coates et al. 2010; Perez-Escamilla & Segall-Correa 2008; Kennedy 2002). The most common food security measurements are – Dietary Intake assessment (DIA) (Alam 2010; Bashir et al. 2010; Feleke et al. 2005), Household Food Insecurity Access Scale (HFIAS) (Chatterjee et al. 2012; Che & Chen 2002), Anthropometry measure (Mariara et al. 2006; Sharif & Merlin 2001) and Household Expenditure Survey Method (HESM) (Titus & Adetokubo 2007; Charlton & Rose 2002).

Maxwell et al. (2013) opined that the combination of Coping Strategies Index (CSI) and Food Consumption Score (FCS) could provide reliable and acceptable results for food security analysis. Recently, Self-assessment Food Security Measures (SAFS) have been used by Heady and Oliver (2013) and Headey (2011). Maxwell et

⁸ Taka (Tk) is the Bangladesh currency, US = Tk 76.21 as on 2 September, 2015.

al. (2013) asserted that SAFS questions are well-correlated to other measures of food security. However, DIA method is commonly used for household food security measurement. The main reason is that this method often used as proxy for all nutritional requirements for health which is the last part of food security definition of FAO. Since there is no unique approach to measure food security, therefore, this study applies various methods including DIA method, CSI and HFIAS for comprehensive understanding of riverine household food security status.

2.8 Gaps in the existing literature

The discussion above indicates that climate change and hazards such as riverbank erosion have had serious impacts on food security and livelihood of the rural riverine households in Bangladesh. Notwithstanding, there have been studies to address climate change, food security and adaptation in Bangladesh. However, an assessment of vulnerability, food security and response strategy for hazards-prone people has yet to be made. This study has, therefore, made an attempt to fill this gap. The following gaps are revealed from the literature review:

- There are few studies on livelihood vulnerability on global scale. So far, there is no study that focused on index-based livelihood vulnerability measurement, especially for marginalized riparian households in Bangladesh. Therefore, applying the standard methodological framework of determining livelihood vulnerability, this study intends to bridge the gap between community necessity and priority at the policy level.
- Although resilience is an ongoing research thrust, however, there is a lack of well accepted framework to measure resilience. In order to explore the factors affecting resilience capacity of the households, this study develops and indicator based resilience capacity index.
- There are many studies on food security, however, to the best of researcher's knowledge, no study has yet examined the determinants of food security of the riverbank erosion hazard-prone rural households.

- The extent of household food security is also relatively unexplored which is important for targeted policy interventions. This study, developing an index, examines the experience of household's food insecurity over the year.
- There is hardly any study that explores the perception of climate change and hazards of marginalised households which is crucial for supporting their autonomous adaptation. Therefore, this study explores hazards-prone households' perception of climate change and their local adaptation strategies.
- There are inadequate studies on the factors affecting adaptation choices of households. This study goes beyond simply examining the determinants of adaptation, it also focuses on the influence of social capital and institutional access on adaptation choices for the resource-poor households.

2.9 Summary

This chapter provides a brief review of existing literature pertinent to the research topic under discussion. First, climate change impacts in general are discussed. Then the riverbank erosion hazard, vulnerability and food security issues are discussed. The chapter clearly sketches out some gaps in the existing literature. This research is proceed to fill up these gaps in the following chapters. The next chapter discusses the various theories and methodologies used for analysing and making a meaningful presentation of the results of this study.

CHAPTER THREE

Methodology

3.1 Chapter outline

This study employs a range of tools and techniques for analysis of the data. Thus, the purpose of this chapter is to illustrate the relevant theories and methodologies areas. employed in this study including description of the study A brief overview of some selected characteristics of the study households is also presented. However, some descriptive results and analysis techniques which are more pertinent to the specific objective are presented in the respective chapter. This chapter is divided into six sections as follows: theoretical framework is presented in Section 3.2. Section 3.3 outlines the methods of descriptive statistics. A description of the study areas followed by survey methods is presented in Section 3.4. Section 3.5 presents a description of the data including common socio-economic characteristics. Section 3.6 then contains the summary of the chapter.

3.2 Theoretical and analytical framework

This study uses a number of theories and models as per the requirements of the study objectives. These theories are the customised approach in the context of vulnerable riverine households in Bangladesh. The relevant theories and approaches are discussed in this section as follows:

3.2.1 Household food security analysis

Household food security analysis is based on the theories of consumer demand and production that is widely known as the Agricultural Household Models (AHMs) (Bashir et al. 2012; Aromolaran 2010; Shaikh 2007; Fleke et al. 2005; Straus 1983; Yotopoulos 1983; Barnum & Squire 1979). In this model, households take the decision both as a producer and a consumer. For a short-run production cycle (i.e., up to one year), the households are assumed to maximise their utility function specified as:

where, F_h = consumed goods by the household produced at home

Fm = consumed food commodities purchased from the market

I = leisure,

Dm = demographic characteristics of the household

A household, both as producer and consumer, is assumed to maximise its utility from the consumption of these goods subject to farm production, income and time constraints as follows (*Production Constraint*):

where, G is the implicit production function; Qi is the quantities of the goods produced on-farm; L is the total labour input to the farm; R^o is the farm technology fixed in short term; A^o is the household's fixed quantity of land; K^o is the fixed stock of capital. If we consider the consumption and income constraint of the households⁹, we can write the function as:

Where, Pi= price of the goods i, $(Q_i - F_h) =$ Marketed surplus of the goods i, W= Wage rate, L_f = Household labour suply for nonfarm use, N= Total non-farm income which adjusts to ensure the equation (3) is equal to zero. *Time constraint:*

T= Total time available for the household to allocate between work and leisure. The income and time constraints on household behaviour can be combined into a single equation by incorporating (4) into (3) as:

After rearranging the above equation it stands as:

The left-hand side shows the household expenditure (consumption) on food and leisure. This includes purchase of the own farm-produced goods $i(P_iF_h)$, the goods households purchased from the market (P_mF_m) and the household's purchase of its

⁹ In the short run, a household assumed to have fixed amount of land, capital and technology so these variables are considered constant.

own leisure time (wI). The right hand-side of the equation is income of the households. It consists of the value of total agricultural production (P_iQ_i), the value of the household's time (wT), the value of labour derived from both farm and hired labour (wL), and non-farm income, N. The lagrangian is:

The first order conditions of the relationship between production and consumption can be establish as:

Household being a consumer tries to maximise its utility by equating the marginal rate of substitution between food and non-food commodities to the marginal product of labour. If household has more production than consumption, it can offer to sale in the market. Similarly, if household is short supplied of labour it hires additional labour. Since it is assumed that household has no leisure time, they offer labour to other farmers and businesses if they have free time. Given the assumption of 'separability' one can derive the production side equation and consumption side equation separately (Fleke et al. 2005; Straus 1983) as:

Once the optimum level of labor is selected, the value of full income when profits have been maximised can be obtained by substituting equation (9) and (10) into the right hand side of the income constraint equation (6) as:

Where, Y^* is the full income under the assumption of maximized profit π^* . These first order conditions for consumption demand can be solved in terms of prices, wage rate, and income as follows:

Where, k = i, m. The three equations (i.e., 9, 10 and 13) give a complete picture of the economic behaviour of the farm household. They are combined through the profit effect. In semi-subsistence farming, household's income is determined by production activities that imply changes in factors influencing production which ultimately affects income, which in turn stimulates consumption behaviour. Incorporating demographic factors D_m , the demand for food indicated in equation (13) can be written as:

The logistic regression model of food security can be written as

Where, \emptyset_i is the conditional probability of food security and β_j 's parameters to be estimated and X_{ij} 's are the independent variables (Results of the analysis are discussed in Chapter 6).

3.2.2 Analysis of adoption of adaptation strategies

The econometric analysis of household adaptation is based upon the random utility theory. Households' adaptation decisions are based on their perceived utility of the different adaptation measures in response to climatic hazards. These decisions are guided by their utility (or profit) maximisation behaviour¹⁰. The households' choice of adaptation strategies is discrete and mutually exclusive. There is no natural ordering in the preferred strategies and the relationship between the underlying latent or unobservable variable is linear that justifies the use of random utility framework model (Verbeek 2004). The farmers in this study are assumed to select from the 15 alternatives those which have the highest utility.

Assuming U_h and U_k are the utility of household i, who chooses between any two alternatives, the random utility model can be written as:

¹⁰ A contrary opinion is that farmers' strategic responses are not explained entirely by profitmaximization behavior but also by farmers' attitudes and values (Marshall et al. 2012; Gasson & Errington 1993).

where, U_{ih} and U_{ik} are an individual household's utility (i) of choosing option h and k, respectively, and V_{ih} and V_{ik} imply the deterministic (observable or explainable) or systematic component of utility. Whereas, ε_{ih} and ε_{ik} represent the stochastic (random or unexplainable) element that stands for unobservable influences on individual choices and measurement error, and are assumed to be independently and identically distributed (Greene 2012). According to utility maximisation behaviour, a household will only choose an option h if $U_{ih}>U_{ik}$ for all $h \neq k$.

The deterministic components V_{ih} or V_{ik} represent an attribute vector x, i.e., $V_{ih} = x'_{ih}\beta$ or $V_{ik} = x'_{ik}\beta$. However, utility is not directly observable; rather, a household's choice of adaptation strategies can be observed. When there are many choices, the likelihood of alternative adaptations can be expressed as a probability:

where, β is a vector of unknown coefficients and x is the vector of the explanatory variables influencing the choice of adaptation and ε is a random error term. For a given x the probability that a household will choose an alternative h is given as follows:

Equation (iv) can be estimated by choice models (Greene 2012). To obtain unbiased and consistent parameters in the model, the assumption of Independence of Irrelevant Alternatives (IIA) must be fulfilled (Cameron & Trivedi 2009). It indicates that the probability of adopting a particular adaptation strategy by a given farm household requires independence from the probability of selecting another adaptation strategy.

Different choice models – multinomial probit (MNP) or multinomial logit (MNL) – can be constructed based upon the assumed distribution of the random disturbance terms. MNL provides a more precise estimation than the MNP (Kropko 2007). Moreover, estimation of MNL is simpler and interpretations of parameter estimates are easier (Cameron & Trivedi 2009; Long 1997). MNL is widely employed in climate adaptation research (e.g., Alam 2015; Alauddin & Sarker 2014; Sarker et al. 2013; Deressa et al. 2009).

The estimated parameters of MNL only show the direction of the impact of the explanatory variables on the dependent variable and do not provide the extent of change or the probabilities. Marginal effects, however, measure the impact on the probability of observing each of several outcomes rather than the impact on a single conditional mean and are more meaningful and interpretable (Cameron & Trivedi 2009; Long 1997). To compute the marginal effects of different exogenous variables, we differentiate equation (iv) with respect to N explanatory variables as follows:

Marginal effects measure the likelihood of change in the probability of the adaptation of a particular choice with respect to a unit change in an explanatory variable (Greene 2012). The MNL model can be regarded as simultaneously estimating binary logits for all possible comparisons among the outcomes. With Z outcomes, only Z-1 binary logits need to be estimated (see results of the analysis in Chapter 8).

3.2.3 Vulnerability analysis

This analysis is based on the vulnerability theories which suggest that households are particularly vulnerable if they have low adaptation capacity (IPCC 2007). According to IPCC (2007) vulnerability is a function of three dimensions as follows:

Vulnerability = f(Exposure, sensitivity, adaptive capacity)

Exposure refers to the environment/location of people that could be adversely affected by physical events and which, thereby, are subject to potential future harm, loss, or damage (Gasper 2010). This also indicates the magnitude and duration of the climate-related events such as change in precipitation or a flood. Whereas sensitivity is the degree to which the system is affected by the exposure, and adaptive capacity refers to the system's ability to withstand or recover from the exposure (Ebi et al. 2006). Exposure, sensitivity and adaptation are composed of various indicators. In recent years, index based vulnerability measurement method is widely employed in many parts of the world (see, for example, Aryal et al. 2014; Toufique & Islam 2014; Shah et al. 2013; Hahn et al. 2009). This study also adopted index based vulnerability analysis method. Detailed derivation of the approach is discussed in Chapter 4.

3.2.4 Resilience analysis

There is a growing interest among the policy makers about an understanding of how far the affected communities are resilient in order to be able to provide best support to them to cope and adapt to climate change and hazards. This analysis is based on the resilience theories which is the function of sensitivity and adaptive capacity (Malone & Brenkert 2008) as shown below:

Resilience = f(Sensitivity, adaptive capacity)

The results of the analysis will help to understand how resilient the study households are and what are the factors influencing their resilience capacity so that necessary interventions can put in place (see detailed in Chapter 5).

3.3 Descriptive statistics

Statistical analyses including descriptive statistics such as mean and percentages, moving average, standard deviation, coefficient of variation, simple t-test and posthoc are used to ensure meaningful presentation of the data. Moreover, a linear trend analysis is also employed to detect the changes in climate variables.

3.4 Description of the study areas and data collection

This section provides the information of the study areas and the procedure of data collection.

3.4.1 The approach

The study applies a mixed method approach for collection of data (quantitative and qualitative data). The methods of data collection were included the survey method (face-to-face interviews of the household head), focus group discussion and the key informants' interviews. One focus group discussion was conducted in each village with a group of 10 to 12 household heads. Moreover, discussions were also conducted with academics, agriculture officers, environmentalists and Non-Government Organisations' (NGOs) workers. The underlying purposes of the discussions were to shape the survey questionnaires and to obtain views on various climatic and socio-economic variables. These opinions were then used to cross-validate the information obtained from the surveys and the key informant interviews.

To fulfill the objectives and research questions of this study, data were gathered from both primary and secondary sources. The procedure of primary data collection (survey design) are described below:

3.4.2 Selection of the study areas

A multi-stage sampling technique was employed to collect data from riverbank erosion hazard-prone rural households¹¹ in Bangladesh. The riverbank erosion-affected districts, upazilas¹² and riverine villages were selected based on the degree of severity of erosion that was identified through a review of the literature, reports in newspapers and in consultation with experts. Within each village, respondents were selected randomly. For the field survey, the Chauhali Upazila of the Sirajgonj district and the Nagarpur Upazila of the Tangail district were selected, as they represent the most riverbank erosion-affected riparian environments in Bangladesh. The area is about 200 km north of Dhaka, the capital of Bangladesh. The Jamuna River¹³, which is reported to cause erosion of around 2,000 ha per year (CEGIS 2012), crosses the study area. Data were collected from six riverine villages, namely; Kashpukuria, Moradpur, Kairat, Datpur, Kashkawalia and Atapara (Figure 3.1). Some pictures of Jamuna riverbank erosion and livelihoods are presented in Figure 3.2.

3.4.3 Unit of analysis

Selection of the units of analysis is considered as the entry point in social science research. The unit of analysis influences greatly to the decision of research design, data collection and data analysis (Silverman & Solman 1998). In this study, the unit of analysis was households and for data collection, the household head (either male or female) was the survey participant. A household (economic agent) is a domestic unit with autonomous decision-making regarding production and consumption (Ellis 1988). In Bangladesh, household heads have the power to exercise decision-making over household's resources and setting strategies.

¹¹ Riparian (riverbank and *char*) communities.

¹² Lower administrative unit of government; below district level but above village level.

¹³ Bangladesh is composed of the floodplains and deltas of three main rivers, the Padma (Ganges in India), the Jamuna and the Meghna (Brahmaputra in India).



Figure 3. 1: The study areas: the Nagarpur and Chauhali Upazilas **Source:** http://www.banglapedia.org/httpdocs/HT/G_0144.HTM



Riverbank erosion damage

Char land under cultivation



Figure 3. 2: Riverbank erosion (Jumuna river) and livelihood in the study areas

Source: Field survey

3.4.4 Sample size

In survey research, it is a vital issue to determine the representative same size which will be yielded sound results to fulfill the research objectives. Some argued that a minimum sample size of around 30 is required for statistical analysis (Champion 1970). Perry (1998) suggested that a sample size of 350 is the optimal size for a structured interview in quantitative research. On the other hand, Bartlett et al. (2001) suggested that 5% of the population is considered to be adequate for the cross-sectional household survey.

Therefore, for this study 15% of the household heads from each village were interviewed, which gave a sample size of 380 households for the study. It is worthy to mention that there were relatively small number of households which varies across the villages. Moreover, rural households in Bangladesh virtually face analogous socio-economic, environmental and climate conditions (i.e. low education attainment and income, relatively high birth rate and high dependence on agriculture for livelihoods) which validates the use of a small sample size that can be typical of the whole population (Blaikie 2010; Gilbert 2008). To do that a complete list of riverine households of the selected villages was first collected from the Department of Agricultural Extension (DAE). The distribution of sampling is given in Table 3.1.

Village	Households	Sample size (15% of the population)
Kashkawalia	650	97
Kashpukuria	750	112
Moradpur	270	41
Kairat	315	47
Datpur	250	38
Atapara	300	45
Total	2535	380

Table 3. 1: Sample size for the survey.

3.4.5 Sampling technique and non-response

Random sampling technique is an important way to make sure that the sample is representative of the population under study (Neuman 2006). To ensure randomness

in sampling, this study employed a computer-generated random number table to the list to select 380 households.

The issue of non-response is unavoidable in cross-section household survey. Scholars, however, argued that when the sample is truly random, non-response does not represent a bias (Fowler 2009; Henry 2009). This study did not face the nonresponse issue greatly rather received good cooperation from the households. Unavailability of respondents or refusal to answer questions were mainly by the female-headed households, which cover less than 2% of the actual samples. In the case of non-response, the interviewers simply proceeded to the next household until the required number of respondents for a particular village was reached.

3.4.6 Questionnaire and data collection

The researcher developed a structured survey questionnaire containing both openended and closed-ended questions. Based on the review of the literature and discussions with the experts, a draft questionnaire was prepared to achieve the research objectives. The first part of the questionnaire contains the information on households' socio-economic condition such as household demographic information, income and expenditure. The second part covers the information that relates to household livelihood status and food security issues. The third part includes the information on their perception of climate change and hazards, and response strategies (see Appendix 1 for the complete questionnaire). Moreover, different semistructured interview schedules were also prepared for conducting FGDs and Key Informant Interviews. Since 'Bengali' was the target language, therefore, back translation of the questionnaire was done following Sperber (2004). It allowed the researcher and the enumerators to collect the precise and reliable data easily.

Prior to the final survey, the questionnaire was pilot-tested with 20 respondents to ensure adequacy of the information obtained and avoid ambiguity of questions. Expert opinion was also shared and their suggestions were incorporated to modify the questionnaire. At this stage, the questionnaire was ready to conduct the survey and the author along with three trained enumerators implemented the survey. Educated enumerators were chosen from the study areas and trained properly. Due to their known environment and familiarity with the people, the selection of households was relatively stress-free. The work of enumerators were supervised by the

researcher and where necessary, clarification of the issues and further instructions were provided.

The face-to-face interviews were conducted between January and May 2014 to collect the data. However, the observation of changes in the study areas particularly the infrastructure (road, institutions) changes was continued in time interval over the whole PhD study period: the study areas experienced massive riverbank erosion in 2015 for which the upazila complex and the only public hospital of Chauhali upazila were disappeared.

Other survey methods including mail and telephone surveys were not possible due to the low education level of the respondents and unavailability of such technology. Moreover, the in-person interview method helps to build rapport between the respondent and interviewer to get more information.

3.4.7 Validity and reliability

Validity and reliability are the two very important issues in research. These issues are related with the characteristics of measurement and its precision. In the questionnaire survey, it is hard to measure accurately which is often resulted in measurement error (Singh 2007; Williams 2003). Validity implies to how conceptually close the variables are to what it intends to measure¹⁴. In other words, validity ask the question: are we measuring what we want to measure? (Muijs 2010). Reliability, on the other hand, implies the consistency of the measures that able to provide identical results in repeated measurements (Blunch 2008; Singh 2007). Reliability of the questionnaire can be achieved by internal consistency (Williams 2003).

In this research, the validity and reliability of the survey question were ensured through adopting various techniques such as appropriate wording in the questionnaire, piloting of the questionnaire, extensive review of the questionnaire by more than six academic experts, and collection of the data through local trained enumerators under close supervision of the researcher. During face-to-face interviews, respondents were asked the questions in more than one way and tested consistency in responses which was relatively easy for the local enumerators for their

¹⁴ For instance, if child nutrition status are taken into account to measure food insecurity rather than considering the number of nights the family members have gone to bed hungry, then it will not be valid measure for this study.

known environment. These together contributed to building rigorous assessment of survey instrument's validity. Moreover, extended analysis of socio-economic characteristics of study households and drawing comparisons with local and national data, and various statistical measures also established the validity and reliability of the research which yielded sound results (for more on this, see the discussion in the proceeding chapters).

3.4.8 Data entry and cleaning

The trained enumerator thoroughly checked the filled questionnaires at the end of everyday data collection. If they found any inconsistencies, they went to the household the next day and made the required changes. They also converted local units of data into standard units. Therefore, in the first stage the filled survey questionnaires were cleaned and validated at the field level. Then the collected data were tabulated into Excel spreadsheet (Excel 13). Once the data entry was completed, the file was thoroughly edited for cleaning the data by producing frequency table for each question and checking the outliers. Thus, the data entry, cleaning and validation were done very vigilantly which made the data reliable. At this stage, the data was ready for statistical analysis. For regression analysis, the Excel data was imported in the Stata (see Chapters 6 and 8). Table 3.2 present the major statistical tools and software used in this study along with their purposes.

Statistical tools	Purpose of use	Link to research	Software
		objectives	
Descriptive statistics	To analysis the socio-	All research	Excel 2013
such as mean,	economic profile of	objectives	
percentage, frequencies	households		
Graphs such as pie	To observe the trend	Objective 3	Excel 2013
charts and line charts	in climate variables,		
	farmers' adoption		
	strategies and barriers		
	to adoption		
Pearson chi-square test	To compare the	Objective 1	SPSS 22
and independent sample	relationship between		
t-test	two categorical		
	variables and		
	continuous variables		
Logit model	To analysis the	Objectives 2 and	STATA 12

Table 3. 2: Statistical techniques, purpose of use and software.

	determinants of	3 (Research	
	household food	question (ii)	
	security and	under objective 2	
	household adaptation	and research	
	choices	question (v) under	
		objective 3)	
Breusch-Pagan test	To check the problem	-do-	STATA 12
	of heteroscedasticity		
Variance Inflation	To check the presence	-do-	STATA 12
Factor	of multicollinearity		
Hausman test	To test the assumption	-do-	STATA 12
	of IIA for the		
	Multinomial logit		
	model (MNL) for		
	adaptation		
Endogeneity test	To make sure that the	-do-	STATA 12
	model has no		
	endogeneity problem		

3.4.9 Secondary data

To support the results of the primary data, relevant secondary data were also collected and compared. This data were mainly collected from various reports of Bangladesh Bureau of Statistics (BBS), Food and Agriculture Organization (FAO), and World Bank (WB).

3.4.10 Research ethical issues

Research ethics is an important aspect in research (David & Resnik 2011). The approval of the USQ Human Research Ethic Committee is mandatory for PhD research before launching the data collection and stick to this up to 5 years after completion the research. The author enrolled in the PhD program at USQ in July 2013. After conformation of the PhD candidature, the author received the ethical approval from the University of Southern Queensland (H13REA244) in November 2013 to collect the survey data from the study areas in Bangladesh.

The ethical standard was maintained during the periods of data collection in Bangladesh by all the enumerators. Before each interview, the purpose of the research and the confidentiality of the data were described, and then their consent to provide information voluntarily were taken. Same procedure was also maintained in FGDs. Most of the respondents cooperated nicely and the author is highly grateful to all of them who provided information and suggestions for this research. Moreover, ethical issues also maintained in presented the results of this thesis in various scientific journals.

3.5 Data description

In this section, some descriptive statistics of the data are illustrated as follows:

3.5.1 Socio-economic characteristics

The information of household socio-demographic and economic characteristics are very useful to get an insight into the profile of the study households and to formulate effective policy interventions. This information can be served as the delimitation of the study so that whatever findings or outcomes derived from this study can be described within the domain of this profile. Socio-economic characteristics of the study households are presented below:

As seen in Table 3.3 that half of the household heads belong to the age group of 46 to 60 years. Average age of the household heads is around 45 years. Currently, the life expectancy at birth in Bangladesh is 70.3 years (UNESCO 2015). The majority of household heads is male (88%) as against women of 12%. The average family size of 5.21 is relatively large compared to the national average of 5.0 (BBS 2012). More than 46% of households had six members or more. The mean education level of the household was below primary level (3.17 years). More than 29% of respondents did not attend school. In Bangladesh, the estimated literacy rate was 61.5 percent in 2015 (UNESCO 2015). Majority of the household heads had education level between primary and secondary level. Only 9% had more than secondary education level.

Households' farm size is relatively low since all household had experienced loss of some of their land. Therefore, the study households were categorized as: large farm household (12%) (>2.5 acres), medium farm household (28%) (1.5–2.49 acres), small farm household (33%) (1.49–0.5 acres) and landless (27%) (<0.5 acres). Household occupation groups are classified according to the main source of income (i.e., >50%) (Table 3.3). As expected, most of the households in the study areas depend on agriculture (71%) which is relatively higher than national

statistics (BBS 2012). Service holders or affluent households usually live in nearby town or other places that are free from erosion problems.

Characteristics/Variables	Number	Percentage				
Age of HH head (Mean :45; Range:25-65)						
≤ 30 years	36	10				
31-45 years	134	35				
46 -60 years	191	50				
61-65 years	19	5				
Gender of HH head	Gender of HH head					
Male	335	88				
Female	45	12				
HHs family members (Mean :5.21,	: Range:3-11)					
3	31	8				
3-5	174	45				
\leq 6 members	175	46				
Religion						
Muslim	337	89				
Others	43	11				
Education (Mean : 3.17 ye	ears; Range: 0-16)					
Illiterate	109	29				
Primary (level 1-5)	137	36				
Secondary (level 6-10)	104	27				
Higher secondary (level 11-12)	21	6				
< Higher secondary (level 12-16)	9	2				
Employment status						
Agriculture	271	71				
Business + Agriculture	75	20				
Services + Agriculture	34	9				
HHs yearly income (Tk)	(Mean:35, 000 Tk; Std	. 38456)				
≥ 35,000	39	10				
36,000-60,000	137	36				
61,000-150,000	151	40				
≤151,000	53	14				
Farm category (Av	erage farm size: 0.56 a	cres)				
Large farm (>2.5 acres)	45	12				
Medium farm household	107	28				
(1.5–2.49 acres)						
Small farm household	127	33				
(1.49–0.5 acres)						
Landless (<0.5 acres)	101	27				

Table 3. 3: Some selected socio-economic characteristic of the study households.

Note: Household = HH

Since most of the farmers depend on agriculture, therefore their income level is also low. More than 50% of households belong to the income level of Tk 60,000

(see Gini coefficient in the next section). Road and transport communication is also inadequate in the areas. Farmers mainly use vans, bicycles, rickshaws, scooters, and tempo driven by small machines to market their products.

3.5.2 Income inequality among the households

In order to show the inequality of income among the households, we estimated the Gini coefficient and the Lorenz curve. The Gini coefficient was calculated by using the formula of Rodrigue et al. (2009):

$$G = \left| 1 - \sum_{k=1}^{k=n-1} (X_{k-1} - X_k) (Y_{k+1} + Y_k) \right|$$

Where, Xs are the proportion or share of households and Ys are the proportion of their corresponding income.

The Gini coefficients are aggregate inequality measures and can vary anywhere from 0 (perfect equality) to 1 (perfect inequality) (Todaro & Smith 2005). The estimated Gini coefficient of 0.511 is higher than Bangladesh's national income Gini coefficient of 0.458 (BBS 2010). This means that income distribution among the sample households is relatively skewed. We also draw a Lorenz curve to show the degree of inequality in income of the respondents (Figure 3.3).



Figure 3. 3: Lorenz curve of income of the study households

This indicates that the bottom 80% of the sample households had only 60% of the total income whereas the top 20% had about 40% of the total income. This

indicates that a huge income inequality exists among the households included in the study. This result is not quite unexpected in the context of rural Bangladesh. In 2010 the income share of the top 5% of the households was 24.61% whereas it was 0.78 for the bottom 5% of the households (BBS 2010).

3.6 Research design: An overview

The below diagram (Figure 3.4) links the research objectives, theories and approaches, data requirements, and major methods of analysis in this study.



Figure 3. 4: Research design

The study has set three objectives based on three close interlinked aspects (i.e., vulnerability, food security and response strategies of the households). Under the first objective – the livelihood vulnerability and resilience capacity due to riverbank erosion and other climate change issues are assessed. In the second stage, the impacts on household food security are analysed. Finally, how the households are responding in such settings is discussed.

3.7 Summary

This chapter discusses various theories and approaches to be used in this thesis. It provides descriptions of the study areas and methods of data collection. This chapter also explains how different statistical methods and software are used to achieve the objectives of the research. A brief overview of some selected characteristics of the study households are presented in this chapter.

A multistage sampling technique was employed to collect the survey data from riverbank erosion-prone households' covering six villages of the Chauhali upazila of the Sirajgonj district and the Nagarpur upazila of the Tangail district in Bangladesh. A total of 380 households head were interviewed. For better understanding of their livelihood and difficulties, discussions were also held with the key informants and other experts in all the study locations. The remaining chapters (Chapter 4 to Chapter 8) are based upon the results of the analysis of the data. The result chapters are organised in a mini thesis format where all sections such as introduction, review of the literature, methodology, results and discussions and conclusions are presented. Relevant descriptive statistics to the specific objective is presented in the respective chapters. The next chapter discusses the vulnerability of households due to riverbank erosion and other climate change issues.
CHAPTER FOUR

Vulnerability of Riverine Rural Households to Climate Change and Hazards

4.1 Chapter outline

Rural riverine households in Bangladesh are confronted with many climate-driven hazards, including riverbank erosion, which results in loss of productive land and other natural resources, and thus threatens their livelihoods and food security. To improve their situation in a sustainable way, it is crucial to have critical information on the livelihood vulnerability of these households, and this information is currently unavailable. Therefore, this chapter assesses the livelihood vulnerability of the riverbank erosion-prone rural households. The findings will allow the stakeholders to understand the complex set of factors that contribute to the vulnerability of the households. The chapter is organised as follows: the background of the study is presented in Section 4.2. Section 4.3 presents the brief methodology of the study. The results are discussed in Section 4.4; and Section 4.6 contains the conclusions of the chapter.

4.2 Introduction

Global environmental research has projected that climate change will intensify over the coming decades and emphasised the need to develop suitable adaptation strategies to address the potential impacts of unavoidable climate change (IPCC 2014). The identification of suitable adaptation strategies needs to start with an assessment of vulnerability (O'Brien et al. 2009; Ford & Smit 2004), which is the condition determined by physical, social, economic and environmental factors and processes that increase the susceptibility of a community/system to the impacts of hazards (UNISDR 2004). There are a number of benefits that vulnerability assessments can offer – for example, vulnerability indicators can be used as an instrument for evaluating development policy frameworks (Eriksen & Kelly 2007); it can provide information for developing adaptation and mitigation plans (Gbetibouo et al. 2010); it allows comparison of different contexts, and monitoring of vulnerabilities over time and space; and enables the setting of priorities in resource allocations for adaptation and mitigation (Preston et al. 2011). Therefore, it has been suggested that there is a need for place-specific and context-specific assessments of vulnerability, since it is driven by many local factors which vary with space and time (Wood et al. 2014; Fraser et al. 2011; Fussel 2010; Cutter et al. 2003). This study assesses the vulnerability of the most susceptible riparian households in Bangladesh.

Although the country experiences frequent events of extreme climatic hazards, riverbank erosion is the disaster that accounts for the largest losses (Penning-Rowsell et al. 2013; Makenron 2000). Damage from riverbank erosion occurs gradually and has long-term impacts which are naturally irrecoverable. About 8,700 hectares (ha) of land are lost each year due to riverbank erosion, which displaces approximately 200,000 people annually and pushes them into vulnerable conditions of food insecurity and poverty (IFAD 2013; Huq & Rabbani 2011; GoB 2010). Due to the dynamics of erosion and accretion in the rivers, some *char* land (sandbars/sand and silt landmasses) has emerged as islands within the river channel or as land attached to the riverbanks in Bangladesh. Households in the char areas are the poorest of the poor and are the most vulnerable (Islam & Hossain 2013; CLP 2010), and they are marginalised from the benefits of mainland people due to their poor communication networks (Sarker et al. 2003; Thompson 2000). According to the estimates of EGIS (2000), the char area covers about 5% of the total land area of the country and is home to around 6.5 million people (5% of the total population). Households in the riverine areas are also prone to frequent flooding and water logging due to their proximity to the rivers which together with erosion, create a most vulnerable environment for them.

There are several indicators which suggest that riparian (river bank and *char*) households are more vulnerable to riverbank erosion and other climate-induced hazards, however, critical information on the degree of their livelihood vulnerability is not available; this information is crucial for the development of appropriate social, economic and environmental policies. The Government of Bangladesh (2011) has also acknowledged information gap regarding the likely impacts of climate change and has highlighted the need to identify most vulnerable sectors and geographical areas. Scholars have opined that policy interventions would do little to affect poverty dynamics unless the vulnerability context is properly understood (Shah et al. 2013;

Hahn et al. 2009; IPCC 2007). This study aims to fill this gap by integrating the LVI and the CVI utilising the IPCC vulnerability framework. The study frames the following research questions to achieve this aim: (i) what are the main drivers of livelihood vulnerability of the riparian households to climate change and hazards?; (ii) are households isolated from the mainland more vulnerable to climate change than other riparian households? and (ii) does the livelihood status serve as a driver of vulnerability to climate change of these hazard-prone rural households in Bangladesh? The next section provides the methodology for vulnerability analysis.

4.3 Data analysis method

As discussed in Chapter 3, a multistage sampling technique was employed for this study to collect the data from riverbank erosion prone areas. Here the vulnerability assessment method is discussed.

The study households were divided up based on the location into two groups, namely, 'riverine mainland households' (from now on 'riverbank households') and '*char* households' (from now on '*char* dwellers'). Although riverine mainland households and *char* dwellers are both affected by riverbank erosion and other climatic hazards, they have different location identity with respect to the river and therefore suffer differently in terms of livelihood vulnerability. Households in the *char* lands are isolated from the mainland by the river and are deprived of all standard government services, whereas riverine mainland households are relatively better off, being better connected to transport and other services. The *char* villages studied were Moradpur, Datpur and Kairat, and the riverbank villages studied were Atapara, Kash Pukuria and Kash Kawalia.

4.3.1 Developing an index for vulnerability analysis

The IPCC (2007) suggests that vulnerability is characterised as a function of three dimensions – exposures, sensitivity and adaptive capacity, as follows:

Vulnerability = f(Exposure, sensitivity, adaptive capacity)

This relationship is essentially determined by the local circumstances. According to Ford and Smit (2004), vulnerability is a positive function of the system's exposure and sensitivity, and a negative function of the adaptive capacity. According to Adger

(1999), vulnerability is considered as an exposure to a group or individual stress due to a change in social and environmental conditions that disrupt livelihoods. Vulnerability assessment reflects the social process and material outcomes within the system and identifies who and what are more or less sensitive to climate risks (Ford et al. 2010; Adger 2006).

Based on the IPCC definition of vulnerability, Turner et al. (2003) developed a vulnerability framework and later Hahn et al. (2009) developed an indicator-based vulnerability assessment that has been used by many scholars in different contexts (see, for example, Panthi et al. 2015; Aryal et al. 2014; Toufique & Islam 2014; Etwire et al. 2013; Shah et al. 2013; Pandey & Jha 2012). This study adopted and extended the Livelihood Vulnerability Index (LVI) proposed by Hahn et al. (2009) and the Climate Vulnerability Index (CVI) proposed by Pandey and Jha (2012) to measure and compare livelihood vulnerability in the context of the riparian households and to assess the relative magnitude of contributing indicators within the concerned vulnerability-dimension under the IPCC framework. The LVI approach is preferred over the Sustainable Livelihood Approach (SLA), developed by Chambers and Conway (1992), which only considered five types of household assets – natural, social, financial, physical and human capital – but failed to integrate the issues of sensitivity and adaptive capacity to climate change. The LVI approach focuses on quantifying the strength of current livelihoods and health and water resource characteristics as well as the capacity of communities to alter these strategies in response to climate-related exposures (Hahn et al. 2009). The study therefore develops a weighted-balance integrated approach to the calculation of the LVI and CVI that incorporates local and indigenous knowledge into the selection of indicators.

The LVI is composed of seven major components – socio-demographic profile, livelihood strategies, social networks, health, food and water, natural disaster and climate variability. Each component comprises several sub-components or indicators. Due to the flexibility of the methods, they study included additional relevant indicators, which are based on a critical review of relevant literature, local circumstances and consultation with experts. Table 4.1 presents the major components and the indicators included in vulnerability analysis in this study along with the explanation of the reason of inclusion.

Major	Sub-components or	Explanation of sub-	Expected relationship
components	indicators	components	D :/:
Socio-	Dependency ratio	Population ratio	Positive
Demographic		under 15 and above	(It is expected that
profile (SDP)		65 years of age to	higher dependency ratio
		the population over	increases household
		15 and below 65	vulnerability)
		years of age	D :::
	Percentage of	Female member	Positive
	female headed HHs	percentage to total	(higher proportion of
		nousenoid members	nousehold female
			members increases
			vulnerability)
	Percentage of HHs	Percentage of	Positive
	where head of the	nousenoids where	(Higher education level
	HHS has not attend	the neads of	of nousenoids nead
	school	nousenoid nave	decrease vulnerability)
		zero years of	
	Average number of	Average number of	Desitive
	family mambars in	family mambars in	(higher numbers of
	the UUs	the households	family mombars
		the nousenoius	increases vulnerability)
	Dercentage of HHs	Dercentage of	Dositive (higher the
	where a women	women family	numbers higher is the
	family members are	members to total	vulnerability)
	not allowed to work	family members	vullerability)
	outside		
Livelihood	Average livelihood	The inverse of (the	Positive
strategies	diversification index	number of	(More agricultural
(LS)		agricultural	livelihoods reduce
		livelihood	vulnerability)
		activities)	
	Percentage of HHs	Percentage of	Positive (higher the
	with the family	households to total	numbers, higher is the
	member (migrate)	households	vulnerability)
	working in a		
	different community		
	Percentage of HHs	Percentage of	Positive (higher the
	solely dependent on	households to total	numbers, higher is the
	agriculture and	households	vulnerability)
	livestock as a		
	source of income		
	Ratio of non-	Ratio of non-	Inverse is considered
	agricultural income	agricultural income	(higher the ratio, lower
	to total income	to total income	the vulnerability)
Health	Average time to	Total time to reach	Positive (higher the

Table 4. 1: Major components and sub-components comprising the Livelihood Vulnerability Index (LVI) in the study areas.

	health facility (at least with MBBS	health facilities	distance, higher is the vulnerability)
	doctor) Percentage of HHs with family members with chronic ill	Percentage of households that report family members chronically ill	Positive (higher the numbers, higher is the vulnerability)
	Percentage of HHs not going to doctors (local doctor) during illness	Percentage of households to total households	Positive (higher the numbers, higher is the vulnerability)
	Percentage of household without sanitary latrine	Percentage of households to total households	Positive (higher the numbers, higher is the vulnerability)
	Percentage of households where a family member missed work or school due to illness in the past two weeks	Percentageofhouseholdsthatreportfamilymembersmissedout work or school	Positive (higher the numbers, higher is the vulnerability)
Food	Average number of months HHs struggle to find food	Average number of months households struggle for food	Positive (higher the numbers, higher is the vulnerability)
	Average crop diversity index	The inverse of (the number of crops grown by households +1)	Positive (higher the index, higher is the vulnerability)
	Percentage of households that do not get food from the family farm	Percentage of households to total households	Positive (higher the numbers, higher is the vulnerability)
	Percentage of household losing their agricultural land	Percentage of households to total households	Positive (higher the numbers, higher is the vulnerability)
	Percentage of households not practicing homestead gardening	Percentage of households to total households	Positive (higher the numbers, higher is the vulnerability)
Water	Percentage of HHs reporting water conflicts	Percentageofhouseholdsthatreport conflictsoverwater in their area	Positive (higher the numbers of conflicts, higher is the vulnerability)
	Percentage of households use unsafe drinking water (River, pond,	Percentage of households to total households	Positive (higher the numbers of households unsafe drinking water, higher is the

	hole, arsenic	vulnerability)		
	contaminated water)			
	Average time to get	Total distance to	Positive (higher the	
	safe drinking water	reach in safe	distance, higher is the	
	source	drinking source	vulnerability)	
Social	Percentage of HHs	Percentage of	Positive (more	
Network	received assistance	households that	assistance reduce	
	from social	report receiving	vulnerability)	
	networks	assistance	-	
	Percentage of HHs	Percentage of	Positive (more sources	
	provided assistance	households to total	of assistance provider	
	to others	households	reduce vulnerability)	
	Percentage of HHs	Percentage of	Positive (more sources	
	borrowing money	households to total	of borrowing money	
	from others	households	reduce vulnerability)	
	Percentage of	Percentage of	Positive (more sources	
	households lending	households to total	of lending money	
	money to others	households	reduce vulnerability)	
	Percentage of	Percentage of	Positive (more sources	
	households	households to total	of assistance reduce	
	receiving	households	vulnerability)	
	assistance/aid from			
	Government and			
NT . 1	NGOs			
Natural	Average number of	Total number of	Positive (higher the	
disaster and	flood, drought, and	disasters reported	numbers of disasters,	
chinate	the post 10 years	by the nousenoids	nigner is the	
variability	Demogratic as of LUL	Demonstrate	Vulnerability)	
	vith on injury or	households that	Positive (nigher the	
	death as a result of	reported either	higher is the	
	natural disasters in	injury or death	vulnerability)	
	the last 10 years	injury of death	vulliciaoliity)	
	Percentage of HHs	Percentage of	Positive (higher the	
	with an injury or	households that	numbers of injury or	
	death to their	reported either	death of livestock	
	livestock as a result	injury or death of	higher is the	
	of natural disasters	their livestock	vulnerability)	
	in the last 10 years			
	Percentage of HHs	Percentage of	Positive (higher the	
	with losses to	households that	numbers of losses,	
	physical assets	reported losses of	higher is the	
	(homestead/agril.	physical assets of	vulnerability)	
	equipment/	their households	-	
	machinery) due to			
	riverbank erosion			
	and other disasters			
	Percentage of HH	Percentage of	Positive (higher the	
	that do not receive a	households that did	numbers of households	
	warning before a	not receive warning	not receiving warning,	

	natural disaster	about any severe	higher is the
		disasters	vulnerability)
Climatic	Perception index of	Percentage of	Positive (changes in
variability	summer temperature	households reported	summer temperature
		change in summer	increase vulnerability)
		temperature	
	Perception index	Percentage of	Positive (changes in
	winter temperature	households reported	winter temperature
		change in winter	increase vulnerability)
		temperature	
	Total rainfall	Percentage of	Positive (changes in
	perception index	households reported	total rainfall increase
		change in total	vulnerability)
		rainfall	
	Perception index of	Percentage of	Positive(changes in
	monsoon rainfall	households reported	monsoon rainfall
		change in monsoon	increase vulnerability)
		rainfall	
	Perception index of	Percentage of	Positive (changes in
	winter months	households reported	winter rainfall increase
	rainfall	change in winter	vulnerability)
		month rainfall	
	Perception index of	Percentage of	Positive (changes in the
	frequency of	households reported	frequency of floods
	floods/riverbank	change in the	increase vulnerability)
	erosion	frequency of floods	

The interpretation of the results of the method demand some caution, since the results expressed in this methodology are in relative terms, rather than absolute, and are assessed at the scale of 0 (least vulnerable) to 1 (most vulnerable). It is very useful for cross-comparison of intra- and inter-group vulnerability and also to identify the most and least vulnerable groups.

In the LVI, each sub-component contributes equally to the overall index, even though each major component is comprised of a number of sub-components. Since all sub-components were measured on a different scale, they were required to be standardised as an index using the following equation:

where k_a is the original sub-component for an area *a*, and k_{min} and k_{max} represent the minimum and maximum values for each sub-component, respectively. These minimum and maximum values were then employed to transform this indicator into a standardized index. For variables that measure frequencies – such as 'percentage of female-headed households' and 'percentage of households where the household head has not attended school' – the minimum value was set at 0 and the maximum at 100.

The sub-components were averaged after being standardised using Equation (ii) to calculate the value of each major component:

where M_a is one of the seven major components for an area *a*, $IndexKa_i$ represents the sub-components, indexed by *i*, which make up each major component, and *n* indicates the number of sub-components in each major component.

Once the values for each of the seven major components were calculated, they were then averaged using Equation (iii) to obtain the LVI, using:

Equation (iii) can also be presented in the following way:

where LVI_a is the LVI for an area *a*, which equals the weighted average of the seven major components. The weights of each of the major components (*WMz*) were determined by the number of sub-components that make up each major component. Weights were included so that all sub-components contribute equally to the overall LVI.

The index for exposure (Exp) includes natural disaster (ND) and climate variability (CV) and was calculated as follows:

where W_{exp1} and W_{exp2} represent the weight for natural disasters and climate variability, respectively. It was equal to the number of sub-components.

The index of sensitivity (Sen) was calculated as follows:

$$Sen = \frac{W_{sen1}H_+W_{sen2}F_+W_{sen3}W_-}{W_{sen1}+W_{sen2}+W_{sen3}}\dots\dots\dots\dots\dots\dots\dots\dots\dots\dots(vi)$$

where W_{sen1} , W_{sen2} and W_{sen3} are the weight of major components health, food and water, respectively.

The adaptive capacity (AdaCap) index was calculated as follows:

where W_{ad1} , W_{ad2} and W_{ad3} represent the weight of the socio-demographic profile, livelihood strategies and social networks, respectively.

The index value for exposure, sensitivity and adaptive capacity were combined for the weightage of CVI as follows:

where N_i is the number of major components in the i^{th} dimensions of vulnerability. The value of each dimension will attain a maximum value of 1 and minimum of 0.

4.3.2 IPCC framework approach

The LVI is also calculated based on the IPCC vulnerability definition. The IPCC approach aggregates the seven major components into three dimensions – exposure, sensitivity and adaptive capacity – for vulnerability analysis. Exposure comprises natural disasters and climate variability; sensitivity comprises food, water and health; and adaptation includes socio-demographic profile, livelihood strategy and social networks. Once the value of these three dimensions was calculated, the three contributing factors were combined using the following equation:

where $LVI - IPCC_a$ is the LVI for community *a* expressed using the IPCC vulnerability framework. The minimum value for LVI - IPCC was scaled to a minimum of -1 (least vulnerable) and a maximum of 1 (most vulnerable).

4.3.3 Limitations

A major problem associated with vulnerability analysis is choosing robust and sound indicators (Etwire et al. 2013; Adger 2006). However, an extensive review of the literature and consultations with the subject experts that were carried out during this study expected to yield sound results. The important issue is that the methodology of the present study – for example, the index values – are free from the limitations of secondary data-driven methods and missing data problems. Moreover, the indicators or sub-components index value may be useful in assessing the impact of a policy or a program to see whether or not the planned activities contribute to reducing vulnerability of the households.

4.4 Discussion of the findings

This section explains the results of the vulnerability analysis in different phases. In the first stage, the results of the major components and sub-components contributing to each of the major components are presented in Table 4.2, along with their overall LVI. The spider diagram of major components of the LVI is illustrated in Figure 4.1. The second stage deals with the estimated values for the different dimensions (sensitivity, exposure and adaptive capacity) of CVI, with results presented in Table 4.3; and Figure 4.2 presents a vulnerability triangle diagram of the dimensions of the CVI.

4.4.1 Livelihood vulnerability index

Households in both the study areas were found to be vulnerable to climate change issues. However, the overall LVI of 0.478 was a little higher for the *char* dwellers compared to that of the riverbank households (0.417) (p < 0.004) (Table 4.2). This indicates that *char* dwellers are more vulnerable than riverbank households.

The study found a small difference in the estimated index value for different socio-demographic profiles for the study sites. However, there was considerable variation observed between the sub-components. As seen in the Table 4.2, the dependency ratio and percentage of female-headed households were higher for *char* dwellers; this is largely due to the fact that many of the husbands who migrate to

major cities as their seasonal coping mechanism to find employment do not return to

their families, leading to a high rate of divorce.

Table 4. 2: Indexed value of major components and sub-components comprising the
LVI (HHs = households).

Major	Indexed value of		Sub-components or	Indexed value for	
components	each component		indicator	each sub-component	
Ĩ	_			(indicato	r)
	Char	Riverbank		Char	Riverbank
	dweller	household		dweller	household
	S			s	S
Socio-	0.291	0.270	Dependency ratio	0.147	0.125
demographic	0.271	0.270	Percentage of female-	0.131	0.075
profile			headed HHs	0.131	0.075
			Percentage of HHs	0.451	0.257
			where head of HH has		
			not attended school		
			Average number of	0.456	0.429
			family members in the		
			HH		
			Percentage of HHs	0.270	0.465
			where female family		
			members are not		
			allowed to work		
			outside the home		
Livelihood	0.324	0.343	Average livelihood	0.128	0.153*
strategies			diversification index		
			Percentage of HHs	0.525	0.414
			where family		
			members migrate to		
			work in a different		
			community		
			Percentage of HHs	0.635	0.799*
			solely dependent on		
			agriculture and		
			livestock as their		
			source of income		
			Ratio of non-	0.007	0.004
			agricultural income to		
			total income		
Health	0.470	0.309	Average time to health	0.464	0.266 ^b **
			facility (at least with		
			qualified doctor)		
			Percentage of HHs	0.337	0.198 ^a **
			with family members		
			who are chronically ill		

			Percentage of HHs	0.549	0.386
			who do not attend a		
			local doctor during		
			illness		
			Percentage of HHs	0.332	0.133 a***
			without sanitary	0.000	0.100
			latrine		
			Percentage of HHs	0.343	0.217
			where a family	0.010	0.217
			member missed work		
			or school due to		
			illness in the past two		
			weeks		
Food	0.757	0.628	Average number of	0.815	0 506 ^b **
1000	0.757	0.020	months HHs struggle	0.015	0.500
			to find food		
				0.437	0.850 ^b **
			diversity index	0.437	0.050
			Percentage of HHs	0.873	0 368 ^a **
			that do not get food	0.075	0.500
			from the family farm		
			Percentage of HHs	1.000	1.000
			losing their	1.000	1.000
			agricultural land		
			Dercontage of UHs	0.662	0 412 a*
			who do not practice	0.002	0.412
			homostood gardoning		
Water	0.287	0.188	Percentage of UHs	0.251	0.117
vv ater	0.287	0.100	reporting water	0.231	0.117
			conflicts		
			Parcentage of HHs	0.555	0.374
			using unsafe drinking	0.555	0.374
			water (river, pond		
			water hole arsenic-		
			contaminated)		
			Average time to safe	0.054	0.072 a*
			drinking water source	0.034	0.072
Social	0.373	0.344	Percentage of HHs	0.572	0.357
network	0.375	0.344	who receive assistance	0.372	0.557
network			from social networks		
			Percentage of HHs	0.166	0.5/11.4**
			who have provided	0.100	0.541
			assistance to others		
			Percentage of HHg	0.742	0 314 ^a **
			horrowing money	0.742	0.514
			from others		
			Percentage of HHg	0.111	0.373
			lending money to	0.111	0.575
			others		
			others		1

			Percentage of HHs	0.273	0.136 ^a **	
			receiving			
			assistance/aid from			
			Government and			
			NGOs			
Natural	0.645	0.562	Average number of	0.821	0.743	
disaster and			reported flood,			
climate			drought and cyclone			
variability			events in the past 10			
			years			
			Percentage of HHs	0.135	0.067 ^a **	
			with an injury or death			
			as a result of natural			
			disasters in the last 10			
			years			
			Percentage of HHs	0.217	0.152	
			with an injury or death			
			to their livestock as a			
			result of natural			
			disasters in the last 10			
			years			
			Percentage of HHs	0.795	0.743	
			with losses of physical			
			assets (homestead/			
			agricultural equipment			
			and machinery) due to			
			riverbank erosion and			
			other disasters			
			Percentage of HHs	0.612	0.542	
			that do not receive a			
			warning before a			
			natural disaster			
Climatic	0.555	0.562	Perception index of	0.59	0.58	
variability			summer temperature			
			Perception index of	0.57	0.57	
			winter temperature			
			Total rainfall	0.53	0.54	
			perception index			
			Perception index of	0.56	0.58	
			monsoon rainfall			
			Perception index of	0.46	0.51	
			winter months rainfall			
			Perception index of	0.62	0.59	
			frequency of floods			
Overall livelih	nood vulne	erability ind	ex:			
Char dwellers: 0.478, Riverbank households: 0.417***						

***p<0.001, **p<0.05 and *p<0.10

^a Fisher's exact test.

^b T-statistics for mean difference test.

The index value of 'livelihood strategies' was lower for *char* dwellers (0.324) than for riverbank households (0.342). The index value for 'social network' differed slightly across sites. The index values of 'food' and 'water' were the highest for *char* dwellers, at 0.757 and 0.287, respectively, against 0.628 and 0.188, respectively, for riverbank households (Table 4.2). The index value for 'natural disasters' was the highest for *char* dwellers, whereas both sites were almost similar for 'climate variability'. The results of major components are presented in the spider diagram (Figure 4.1).



Figure 4. 1: Spider diagram of major components of the LVI

4.4.2 Climate vulnerability index

The various dimensions of vulnerability are presented in Table 4.3.

 Table 4. 3:
 Indexed dimensions of climate vulnerability of char dwellers and riverbank households.

Contributing factors to vulnerability	Char	Riverbank
	dwellers	households
Adaptive capacity	0.330	0.317
(Socio-demographic profile, livelihood strategies and		
social network)		
Sensitivity (Health, food and water)	0.538	0.403
Exposure (Natural disaster and climate variability)	0.596	0.562
Climate vulnerability index (CVI)	0.924	0.915
LVI-IPCC	0.143	0.099

The CVI was 0.924 and 0.915 for *char* dwellers and riverbank households, respectively, which indicates that households in both areas are vulnerable to climate change and variability. The vulnerability triangle that plots the scores of contributing factors for exposure, sensitivity and adaptive capacity are shown in Figure 4.2. It is evident from the figure that *char* dwellers are more exposed (0.596) to climate change than riverbank households (0.562).

In terms of sensitivity, riverbank households are less sensitive than *char* dwellers. The results of the LVI-IPCC estimation do not change the ranking of vulnerability. The *char* dwellers have come out as the most vulnerable with a score of 0.143, in contrast to riverbank households with a score of 0.099.



Figure 4. 2: Vulnerability triangle diagram of the dimensions of the CVI

4.4.3 Discussions

The following discussions are based on the information presented in Tables 4.2 and 4.3. Table 4.2 shows the components and sub-components of the LVI that are the major contributing factors for the highest and lowest vulnerability in each site. The variations in the value of both LVI and CVI across sites indicate that the vulnerability of riparian households varied both overall and in relation to the particular components and sub-components.

The most influential factors for the variation in the LVI were 'food', 'water' and 'health'. The value of the LVI for 'food', 'water' and 'health' was the highest for *char* dwellers, at 0.757, 0.287 and 0.470, respectively, than for riverine mainland

households (0.628, 0.188 and 0.309, respectively). The contribution of the 'food' component to increasing livelihood vulnerability of *char* dwellers is likely due to the reason that they have to struggle more to manage their food, and local agricultural production is limited and can feed them for only a few months of the year. Crop cultivation is inadequate due to poor soils (sandy soils) and limited land ownership. The contribution of livestock to food is also limited. Furthermore, *char* dwellers experience more natural disasters than others, which also results in loss of livestock. During the rainy season, *char* dwellers normally move to a safer place, mainly in nearby embankments or open spaces. In such a situation, they typically sell their livestock at a low price and in some cases actually lose their livestock due to poor health. The component 'water' had an even larger influence on vulnerability in the *char* area. Households in this area use unsafe drinking water mainly from the river; although there are several tube-wells, most of them were found to be contaminated with arsenic. Households are aware of the danger of drinking arsenic-contaminated water, however, they have no choice but to drink it. In order to get arsenic-free water, they would have to travel more than 1 km and sometimes become involved in conflict with others.

The *char* dwellers are more vulnerable in terms of access to health facilities. The contribution of the 'health' component in the *char* area to increasing livelihood vulnerability can be explained by the fact that they have to travel a longer distance (more than 2.5 km) to reach the health and veterinary centres. Since they are not part of the mainland and boat is the main form of transport to the mainland, it naturally takes longer to reach the canters. Besides, many of the households still believe in their traditional system to recover from sickness rather than going to the local doctor. The number of chronically ill people is the highest for the *char* area. They are deprived of all kinds of standard government services. They have a low level of education and awareness and, coupled with their traditional beliefs, make them more vulnerable. Such social barriers affect vulnerability and adaptive capacity of a particular community (Jones & Boyd 2011). Notably, in Bangladesh, overall public spending on sanitation and drinking water, and expenditure on health care are the lowest in the world (WHO 2012). Char dwellers are more dependent on agriculture and livestock-related activities, and this dependency on agriculture-based activities increases the vulnerability of the households who do not diversify (Mirza 2003).

However, it is encouraging to find that households are cultivating in the emerging *char* lands, which remained fallow previously due to the lack of crop varieties suitable for such land.

Although these major components are found to be the highest for char dwellers, the index values are also high for riverbank households. This high index values indicate that the components 'food', 'water' and 'health' facilities also act as primary drivers to increase the vulnerability of the riverbank households, coupled with natural disasters. Previous literature also showed that displaced riverbank households are subject to different hazards, notably lack of adequate housing and health facilities, and shortage of drinking water and sanitation facilities (Lein 2010; Zaman 2007; Choudhary et al. 2005; Hulton & Haque 2003). The remaining major components such as 'socio-demographic profile', 'social network' and 'climate variability' have contributed to a more or less similar extent in determining the LVI of both sites, as there was insignificant variation. However, households in both locations reported poor access to governmental or formal financial institutions, including NGOs. This is mainly due to their poor economic conditions where the possibility of financial institutions recovering their credit is somewhat uncertain. In addition, because the addresses of riverbank households often change due to changes in homestead position as a result of erosion, their social network – the key to social capital - was found to be limited. They reported to have less farmer-to-farmer contacts and the contact with the extension service providers from whom they can get advice related to agriculture and rural development.

The study households are not only vulnerable to riverbank erosion but also to other climatic hazards. The climate change vulnerability index value of the three dimensions – adaptive capacity, sensitivity and exposure – contributed to the CVI of the communities. It is evident from the results in Table 4.3 that the highest sensitivity and exposure value contributed to the highest CVI for *char* dwellers. The indicators of 'food', 'water' and 'health' were comparatively higher for *char* dwellers than for riverbank households. The riverbank households showed relatively more adaptive capacity due to many reasons, such as higher opportunities to diversify their income sources, comparatively less dependence on agriculture, low dependency ratio, higher level of education and being better connected to transport and other services. The

above discussions indicate that access to 'food', 'water' and 'health' facilities are the main drivers of the livelihood vulnerability of the study households.

4.4.4 The vicious cycle of land loss and poverty

The riverbank erosion hazard is an age-old problem in Bangladesh which causes the loss of riverbank households' entitlement every year due to the loss of productionbased entitlement (the loss of farm land) and labour-based entitlement (reduced employment opportunities in farming). It has serious consequences in the study areas where the majority of the people depend on wage earnings and other non-farm activities for their livelihoods. Thus, all levels of households are impacted severely by riverbank erosion and are forced into a low livelihood status. Good health is also very important because it enables people to pursue different farm and non-farm activities efficiently and helps them to achieve their livelihood objectives. It was observed in the study areas that farmers' physical health status influences their access to farm and non-farm activities. If the household members become sick mainly due to inadequate calorie intake and lack of proper health facilities, they will no longer be able to perform their farming jobs and ultimately become vulnerable and a burden to the family and society. Ironically, this is a common phenomenon in the study areas as an impact of recurring riverbank erosions and other climate change issues.

The study revealed that livelihood conditions of the study households follow a cycle which starts with low livelihood status and ends in poverty, as shown in Figure 4.3. Due to their existing low livelihood conditions, their opportunities to earn income from both agricultural and non-agricultural activities become limited. This limited income has resulted in low household food intake, as the main source of household food supply is either from domestic production or from market purchase. Due to loss of farmland as a result of riverbank erosion, their food production is less, and low income reduces their purchasing power to obtain necessary food items from the market. Therefore, it is inevitable that they take less food than is required, which has forced them into becoming food insecure. Lack of food means lack of required calories to remain active for doing farm and non-farm jobs which, when coupled with limited access to health services, gradually pushes household members to become susceptible to many diseases and poorer health. Diseases such as fever, cough, skin infections and diarrhoea are common in the areas. These pose another obstacle for the households through missed work and by affecting their job opportunities; this increases their vulnerability and ultimately pushes them into the vicious cycle of poverty.

It will be difficult for a developing country such as Bangladesh to bring all those inactive people into the social safety net program and thus achieve the MDGs of eradicating poverty and enhancing food security. This is mainly due to the nature of the economy, which is characterised as a poor economy confronted with many other problems such as climate change issues, natural disasters, high population growth and poverty. Therefore, the existing low livelihood status of the households is also a driver of their vulnerability in the context of future climate change issue. It is also reported that the poor households have less ability to cope with climate change (Szbo et al. 2015; Jordan 2015; Prowse & Scott 2008; Adger 2006). This demands more targeted policy interventions for improving the livelihood status of these segments of the population – it is these interventions with the aim of improving and sustaining the livelihoods of these households which are the key to alleviation of poverty (DFID 1999; UNDP 2005).



Figure 4. 3: The livelihood cycle driven by hazards in the study areas

In this vulnerability analysis, a wide range of sub-components were used where each indicator under each major component has the potential to influence that major component; in the same way, each major component under each dimension has the potential to influence that dimension. Variations in vulnerability according to location were also found in previous studies (see, for instance, Hahn et al. 2009; Aryal et al. 2014; Toufique & Islam 2014; Pandey & Jha 2012) and within male and female groups (Shah et al. 2013). The assessments of vulnerability as well as individual dimensions (exposure, sensitivity and adaptive capacity) of climate change of such hazard-prone rural households provide insights into identifying those dimensions that require interventions to reduce overall climate change vulnerability and improve livelihoods.

4.5 Summary of the chapter

In every country, there are many areas that are at a risk of being affected by various natural hazards such as floods, droughts, cyclones and erosion. People living in these areas have relatively limited capacity to cope with shocks and, consequently, natural disasters may have persistent effects on their lives and livelihood welfare. This chapter presented the results of the assessment of the vulnerability of the most vulnerable riparian households in Bangladesh: such information is crucial for enhancing vulnerable households' resilience in the face of hazards and for coping with climate change and variability. The vulnerability assessment in this study is a customised approach to the calculation of the LVI and CVI in the context of riverine households in Bangladesh. It incorporates local and indigenous knowledge into the selection of sub-components and indicators. It supports the notion that vulnerability to climate change does not exist in isolation from wider socio-economic and bio-physical attributes of the communities.

The differences in overall CVI, as well as dimensions of CVI, indicate that vulnerability to climate change differs even within the groups of people adopting similar livelihood activities. The index values of the LVI and LVI-IPCC, which are related methods, provide evidence that riparian households in both the sites are vulnerable, particularly in the areas of food, health and livelihood strategies. Furthermore, low livelihood status of the households appears to be a driver of vulnerability in the future climate change perspective, which can lead to a vicious cycle of poverty. Therefore, special attention needs to be given to these hazard-prone rural households in order to seek improvement in the areas of food availability and access to health, water and sanitation and to reduce the hazard sensitivity of these households. In order to enhance the adaptive capacity of these households, more focus needs to be given to strengthening the socio-demographic profile, social

networks and diversification of their livelihood activities. For example, women can be engaged in other income-generating activities such as tailoring, handicrafts or embroidery, which should be facilitated through proper training. Planned adaptation strategies such as access to institutions and credit facilities, and a package of technologies through agro-ecological based research, particularly for the emerging *char* land, might help to cope with challenges. Development of communication and transport networks and infrastructure is important in order to support alternative livelihoods of the households and improve their access to services. The findings of this study will help in formulating effective policies and programs to improve and sustain the site-specific coping and adaptation strategies for the resource-poor riparian households, and thus assist with incorporating these strategies into the wider climate change policies. The next chapter discusses the factors influencing the resilience capacity of the households.

CHAPTER FIVE

Resilience of the Riverbank Erosion-Prone Households

5.1 Chapter outline

Despite the increasing recognition of the need for building resilience of the poor farmers in the face of changing climate issues, there is a lack of information on the factors influencing resilience capacity. This chapter develops an indicator-based resilience capacity index to understand the factors influencing resilience capacity of resource-poor riverbank erosion-prone rural households in Bangladesh. The rest of the chapter is organised as follows: Section 5.2 contains the introduction of the study. The conceptual framework is presented in section 5.3. Section 5.4 describes the methodology for assessing the resilience capacity. The focus of the paper then shifts to the survey results followed by discussions in section 5.5. Section 5.6 concludes the chapter and provides some policy guidelines.

5.2 Introduction

Although Bangladesh is one of the most vulnerable countries due to climate change and hazards; not all communities within the country are uniformly affected due to differential livelihood options and resources for adaptation. The coastal and riverine households in Bangladesh are the most susceptible to the impacts of climate-driven hazards including riverbank erosion (GoB 2010). Moreover, due to climate change, they are expected to face projected increases in mean annual temperatures, uncertainty in rainfall, a likely reduction of cereal crop production, and surges in disease, pest and weed pressure on crops and livestock (Niang et al. 2014).

In such unavoidable circumstances, there is increasing recognition for the need for resilient agricultural practices and building resilience capacity for the poor farmers to cope with increasing climatic hazards by many government and non-government agencies globally (IPCC 2014; UN 2013; IFPRI 2010; DFID 2009; WB 2009; UNEP 2008). However, there has been a lack of information around the factors influencing such household resilience, particularly socio-economic resilience of the disaster-prone communities (Akter & Mallick 2013; Cutter et al. 2008). The

resilience of a household is understood by its capacity to absorb shocks while still maintaining its core functions. Scholar argued that the more resilient a household, the greater its ability to absorb shock and disturbances (Traerup 2012). Livelihoods of resource-poor rural households in developing countries like Bangladesh is generally dependent on natural resources and their capacity to cope and adapt with the compounding influence of climate change and hazards which are largely uncertain (WB 2013; Stokes & Howden 2009; Dessai et al. 2007; IPCC 2007). A loss of resilience of a natural resource-dependent community contributes to an increase in its vulnerability to shocks which could have previously been absorbed (Kasperson & Kasperson 2001). One of the principal objectives of disaster risk mitigation strategies is to achieve disaster-resilient communities (UN/ISDA 2005). Policy makers are interested in knowing what affected communities can do for themselves and how to best support the capacity of resource users to cope and adapt to climate change and hazards (Kulig et al. 2013; Wilson 2012; Nelson 2011; Nelson, et al. 2007; IFRC 2004; Walker et al. 2004; Gunderson et al. 1999).

It is therefore, crucial to better understand riparian households resilience strategies resulting from their long-term knowledge, experience and practices which will enable policy makers to ensure interventions are targeted to appropriate climate adaptation processes to mitigate the effects of an adverse climate and hazards in the country (Marshall 2010; Tompkins & Adger 2004). This chapter focuses on assessing the resilience of vulnerable riverine households from socio-economic perspectives through developing an indicator based resilience capacity index (RCI). The research questions used here to seek those answers are: (1) whether the riverine mainland households are more resilient than *char* households to river bank erosion and other climate change issues?; and (ii) what are the factors influencing their resilience capacity? In the next section the conception framework of this study is described.

5.3 Conceptual framework

In this section the conceptual linkages between climate change, vulnerability and resilience are first explored and then the study proceeds to answer the research questions. Resilience is an evolving concept applicable to climate adaptation which has been routinely applied in numerous disciplines, especially in the field of disaster

management (Manyena 2006). One of the important aims of employing resilience theory to empirical studies is to assess the current state of the socio-ecological system and make predictions about whether or not the system is resilient (Gilbert 2010; Marshall 2010; Cumming et al. 2005; Berkes & Folke 1998). In the climate change literature, resilience is often used to describe a system characteristic that determines how the impacts of climate change will be experienced (Adger 2000; Peterson 2000). Resilience denotes the ability of a system to return to an earlier (meta-) stable state after a perturbation (Fussel 2007) and to adapt to change or retain its essential functions irrespective of the changing conditions that it experiences (Wilson 2012; Perrings 2006). It is also described as a mechanism of self-organisation, the capacity to learn from experience, to process information and adapt accordingly (Cutter 2008; Marshall & Marshall, 2007; Gallopin 2006; Folke 2006; Klein et al. 2003). There is, however, debate surrounding the nexus between vulnerability and resilience.

The IPCC defines the term vulnerability, resilience and adaptation as follows: Vulnerability is the degree to which a system is susceptible to, and unable to cope with, adverse effects of climate change and variability and extremes. Resilience, on the other hand, is the ability of a social or ecological system to absorb disturbances while retaining the same basic structure and ways of functioning, the capacity for self-organisation, and the capacity to adapt to stress and change. Adaptive capacity is the ability of a system to adjust to climate change and variability and to moderate potential damages, to take advantage of opportunities, or to cope with the consequences (IPCC 2007). Scholars have mentioned resilience as a positive adaptive response¹⁵ to adversity (Luthar et al. 2000; Masten et al. 1990). Vulnerability, on the other hand, is often denoted as the antonym, i.e. flip side of resilience (Folke et al. 2002). Researchers referred vulnerability as a combination of sensitivity, exposure and adaptive capacity (Paavola 2008; IPCC 2007; Gallopin 2006; McCarthy 2001). Resilience, unlike vulnerability, does not include exposure to a disturbance (Gallopin 2006). Nelson et al. (2007) argued that vulnerability and resilience are considered to be linked to one another via response capacity. Some scholars (see, for example, Adger 2006; Smit & Wandel 2006) use these two terms synonymously. However, some viewed vulnerability and resilience are two distinct

¹⁵ Though some response may increase resilience in short period but can create great vulnerability in the long-term (Maru et al. 2014).

concepts with some components in common (Sapountzaki 2012; Cutter et al. 2008). Some scholars have taken a different approach, by treating resilience as an isolated concept in their disaster analysis. They focused on some specific strategies such as microfinance, and migration to rebound after disaster (Mallick & Vogt 2012; Mohapatra et al. 2012; Parvin & Shaw 2012). Perez et al. (2015) considered fostering resilience as an equivalent to building an on-going adaptive capacity of individuals and social organisations. Given the limitation of a widely accepted framework of resilience assessment, this study adopted resilience as a function of sensitivity and adaptive capacity proposed by Malone and Brenkert (2008), Brenkert and Malone (2005) and Moss et al. (2001) as follows:

Resilience = f (Sensitivity, adaptive capacity) (i)

Sensitivity and adaptive capacity are, in turn, composed of different components. Every element is composed of various indicators. In the sensitivity dimension the components included are food, water and health. The adaptation capacity consists of the components of household socio-demographic profile, livelihood strategies and social networks (Figure 5.1). Exposure includes natural disaster and climate variability (Shah et al. 2013; Pandey & Jha 2012; Hahn et al. 2009; IPCC 2007).



Figure 5. 1: Conceptual framework for resilience assessment

In resilience assessment, it is often difficult to quantify in absolute terms since the techniques are quantitative and use selected indicators or variables as proxies (Schneiderbauer & Ehrlich 2006). The quantitative indicator approach though, celebrated several criticisms including subjectivity regarding variable selection and weighting, lack of availability of certain variables, problems with aggregation to different scales, and difficulties in validating the results (de Leon & Carlos 2006; Luers et al. 2003). However, the usefulness of quantitative indicators for reducing complexity, measuring progress, mapping, and setting priorities makes them an important tool for policy makers (Cutter et. al. 2008).

Based on this discussion and the author's proposed conceptual framework for this study (Figure 5.1), the rationale of selecting indicators is the relevance of indicators to the local conditions resulting from the review of literature, consultations with the local experts, and field experience. It can be, however, argued that the indicators are not representative enough to assess the resilience. This issue is not uncommon in any indicator set; see, for example, Wellbeing Index (Prescott-Allen, 2001); Environmental Sustainability Index (YCELP & CIESIN 2002); Livelihood Vulnerability Index (Hahn et al. 2009).

5.4 Data analysis method

A description of study areas, sampling and data collection procedure have already been discussed in Chapter 3. This section mainly focuses on development of the resilience index. The study households were divided into 'riverine mainland households' and '*char* households' based on the location (see more in Section 4.3.1).

5.4.1 Developing resilience index

A resilience capacity index (RCI) is constructed, driving from previous equation (i) sensitivity (the potentially negative impact of climate change) and adaptive capacity (the capability to maintain, minimise loss of, or maximise gains in welfare) of the study households. The indicators included in the sensitivity and adaptive capacities are illustrated in Table 5.1. These indicators were selected on the basis of their relevant contribution to each component.

Table 5. 1: Components and indicators comprising the Resilience Capacity Index(RCI) developed for the study areas (HHs = Households).

Components	Indicators	Score/Values	Possible
_			limitations/Comments
Socio-	Dependency ratio?	If 1:3, then	Higher dependency will
demographic		score =1, if	increase the vulnerability.
profile		more $= 0$	One earning member can
			lead a family of 3
			members properly. It was
			considered standard after
			consultation with local
			people in the area.
	Education level?	Illiterate $= 0$,	Education level of
		Level 1-5,=	household head can
		1, 5-10= 2,	increase family income
		10-12=3,	and enhance resilience
		above $12 = 4$	capacity.
	Do you adopt any	Yes = 1, No	If no, it means chances of
	contraceptive	= 0	increase family members
	method?		which reduce their
			capacity to cope with any
			shocks.
	Does your family	Yes = 1. No	Some household heads
	send children to	= 0	(HHSs) engage their
	school?	_	children in some income
			earning activities due to
			their family need.
			However, it ultimately
			reduces his/her future
			scope to enter the informal
			iob market and thus more
			income
Health issues	Does your family	Yes = 1 No	This is related to good
ficulti issues	use sanitary toilet?	= 0	health as well as an
	use summing tomet.	-0	environmental issue. In the
			study areas the HHs
			physical health condition
			has influence in terms of
			access to farm and non-
			farm activities
	Do you have any	Ves - 1 No	HHs members with
	family members	-0	chronic disease will
	with chronic	- 0	increase the vulnerability
	illness?		increase the vulnerability.
	Access to health	Ves - 1 No	Access to health services
	services?	-0	can contribute to
	501 11005 :	- 0	remaining fit and healthy
			and thus reduce
	1		

			vulnerability.
	Current health	Good=2,	Good health is important
	condition of the	Poor $=1$,	for doing farm and non-
	household head?	Sick=0	farm jobs. If the HH head
	(based on		possesses an
	observation and		underprivileged or sick
	asking about		category of health, it may
	physical health)		mean he/she might not get
			employment and thus
			become less resilient.
Water	Does your family	Yes = 1 No	Use of unsafe drinking
	use tube-well	= 0	water acts as a source of
	water?	U U	many diseases.
	Distance the source	Within 5	A greater distance from a
	of safe drinking	minutes'	safe drinking water source
	water?	walk $= 1$	will increase vulnerability
	water.	more than= 0	HH's agreed that they
		more man-o	would walk up to five
			minutes to collect safe
			drinking water
Food	Household food	Yes = 1 No	Current food security
1000	secure or not?	= 0	status can reduce
	(self-ranked)	U U	vulnerability.
	Do you adopt zero-	Yes = 1. No	This practice can enhance
	tillage cultivation?	= 0	HHs resilience capacity.
		U U	Improved agricultural
			practices can contribute to
			a reduction in
			vulnerability.
	Do you adopt new	Yes = 1, No	This practice can increase
	cropping practice?	= 0	overall food production
	11 01		and enhance HHs
			resilience capacity.
	Do you adopt	Yes = 1, No	This practice can improve
	improved	= 0	HHs resilience capacity.
	management of		
	weeds?		
	Do you adopt	Yes = 1, No	This practice can enhance
	improved	=0	HHs resilience capacity.
	management of		This is important for the
	manure?		poor farmers in the face of
			increasing prices for
			chemical fertilizer.
	Do you adopt IPM?	Yes = 1, No	This practice can improve
		= 0	HH resilience capacity.
	Do you cultivate	Yes = 1, No	Scope and practices of
	multiple crops?	= 0	cultivating multiple crops
			will reduce vulnerability
			and improve resilience.
	How many months	No. of	Households' ability to

	can you provide food from your family farm?	months	supply food from their own produce will reduce vulnerability and enhance resilience.
	Do you practice homestead gardening?	Yes = 1, No = 0	This is expected to contribute to the increase in HHs food supply and income earnings.
Livelihood strategies	What is your main profession? (Agriculture=1)	If, Agri = 1, Agril + Livestock = 2, Agri+ Petty business = 3, Service = 4	Dependence on agriculture increases vulnerability and reduces resilience.
	Do you receive remittance from family member migrated to cities?	Yes = 1, No = 0	Remittance can help to improve the livelihood and thus resilience.
	Do you practice tree plantation?	Yes = 1, No = 0	Tree planting can contribute to an increase in family income and reduce erosion.
	Do you allow your family women members to work outside the home?	Yes = 1, No = 0	Women members having working opportunities outside of the home can contribute to an increase in household income and enhance resilience.
Social network	Are you a member of any cooperative society?	Yes = 1, No = 0	Involvement in social organizations can reduce vulnerability and improve resilience.
	Have you any saving accounts?	Yes = 1, No = 0	In the face of any disaster HHs are able to rely on this.
	Have you received any training in your profession?	Yes = 1, No = 0	It enables HHS to better manage farming activities and increase production/ income.
	Do you explore and utilize information technology for professional, health and family planning activities?	Yes = 1, No = 0	Access to and use of information technology can contribute to the reduction of the overall vulnerability and improve resilience.
	Do you get cooperation from other village people in case of your	Yes = 1, No = 0	Normally, lower economic households get less help from others. More sources of support can reduce

need?		vulnerability and enhance
		resilience.
	Yes = 1, No	This is important for
Do you allow	= 0	effective resource planning
women in decision-		within households. It can
making process		contribute to enhancing
		resilience.
Are your family	Yes = 1, No	It can increases the
members a member	= 0	potentiality of getting
of cooperative		assistance and help in case
society?		of necessity.

The next crucial issue is to allocate a score to each indicator. Eakin and Bojorquez-Tapia (2008) note that equal weighting makes an implicit judgment about the degree of influence of each indicator and propose a logic-based weighting method as a more objective approach. Vincent (2004, 2007) and Sullivan et al. (2002) suggest expert opinions and stakeholder discussions to determine weighting schemes. In this study, we assigned weight to each contributing indicator rather than any of the dimensions as a whole. This score is based on the knowledge of the local experts and scholars with an emphasis on the inductive approach¹⁶. The simple arithmetic functions such as weighted mean index and aggregated mean index were used to calculate the scores for indicators and dimension of resilience capacity, respectively (Mazummder & Lu 2015; Habiba et al. 2011). In RCI, each component contributes differently to the overall index, since each component is comprised of a different number of indicators.

First the index value of each indicator was calculated using the equation as follows:

Where, In_a is one of the indicators for an area a, $SIna_i$ represents the total score of each indicator, indexed by *i*, and n indicates the number of observations. After getting an index value of indicators, the next step was to estimate the index value of each component which was calculated using the equation as follows:

¹⁶ Please see Goddard and Melville (2004) for further details.

Where, C_a is the index value of one of the components for an area a, $index_{In_{C_i}}$ represents the value of indicators in each component, indexed by i, and S_{C_i} indicates the value of indicators in each component.

Once the index value of each component was calculated, they were then used to calculate the index value of sensitivity and adaptive capacity.

The index value of sensitivity (Sen) was calculated as follows:

$$Sen = \frac{W_{sen1}Health_{+}W_{sen2}Food + W_{sen3}Water}{W_{sen1} + W_{sen2} + W_{sen3}}\dots\dots\dots\dots\dots\dots\dots\dots\dots\dots\dots(iv)$$

Where, W_{sen1} , W_{sen2} and W_{sen3} are the weight (one for each) of the components health, food and water, respectively.

The index value of adaptive capacity (AdaCap) was calculated as follows:

Where, W_{ad1} , W_{ad2} and W_{ad3} represent the weight (one for each) of the component of socio-demographic, livelihood strategies and social networks, respectively. Each dimension will attain a maximum value of 1 and minimum of 0.

Then the index value for sensitivity and adaptive capacity are combined to calculate the RCI as follows:

The higher the value of the RCI, the higher the resilience capacity and vice versa. Some caution needed to be taken in interpreting the results, since the results are expressed in relative terms, rather than absolute, and are assessed at the scale of 1 (most resilient) to 0 (least resilient).

5.5 Results and discussions

In this section, the results are described in different phases. In the first stage, the results of the RCI components along with the indicators contributing to each of the

components are presented in Table 5.2. The spider diagram of major components of RCI is illustrated in Figure 5.2. The results are discussed in the later stage.

5.5.1 Resilience index

The overall RCI of 0.299 for riverine mainland households was found to be significantly higher as compared to the *char* households (0.204) (p< 0.04) (Table 5.2). This indicated that riverine mainland households were relatively more resilient than *char* households. However, both of them had a low index value that indicated a truncated resilient capacity of the communities to climate change and hazards.

Table 5. 2: Indexed value of components and indicators comprising the resilience capacity index.

Components	Index valu	e of each	Indicators	Indexed	value for
	component			each in	dicator
	Char	Riverine		Char	Riverine
	household	mainland		household	mainland
		household		nouschold	household
Socio-			Dependency ratio?	0.168	0.373 ^a **
demographic profile			Education level?	0.894	1.242 ^b ** *
			Do you adopt any contraceptive method?	0.905	0.941
	0.397	0.503	Does your family send children to school?	0.873	1.000 ^a *
Health issues			Does your family use sanitary toilet?	0.913	0.987
			Do you have any family members with chronic illness?	0.863	0.943 ^a *
			Access to health facilities?	0.000	0.778 ^a ** *
	0.494	0.823	Current health condition of the household head?	0.692	1.407 ^b **
Water			Does your family use tube-well water?	0.963	1.000 ^a *
	0.775	0.938	Distance the source of safe drinking water?	0.586	0.876 ^a **
Food	0.230	0.515	Household food	0.412	0.688^{a**}

			secure or not?		
			Do you adopt zero-tillage cultivation?	0.681	0.795 ^a **
			Do you adopt new cropping practice?	0.342	0.89 ^a ***
			Do you adopt improved management of weeds?	0.114	0.769 ^a **
			Do you adopt improved management of manure?	0.123	0.743 ^a **
			Do you adopt IPM?	0.045	0.623 ^a **
			Do you cultivate multiple crops?	0.051	0.813 ^a **
			How many months can you provide food from your family farm?	2.12	4.513 ^b ** *
			Do you practice homestead gardening?	0.712	0.471 ^a *
Livelihood strategies			What is your main profession?	1.178	1.663 ^b *
			Do you receive remittance from a family member who has migrated to the city?	0.167	0.375 ^a **
			Do you practice tree plantation?	0.457	0.492 ^a *
	0.379	0.430	Do you allow your family women members to work outside home?	0.854	0.482 ^a *
Social network			Are you a member of any cooperative society?	0.078	0.421 ^a ** *
			Have you any saving accounts?	0.014	0.753 ^a **
			Have you received any training in your profession?	0.047	0.218 ^a *
	0.111	0.446	Do you explore and utilize information technology for	0.034	0.337 ^a ** *

Overall resilien	nce capacity ind	ex:	0.201	0.297**
		society?		
		cooperative		
		member of	0.213	0.436 ^a **
		members a		
		Are your family		
		process?		
		decision- making	0.114	0.342 ^a **
		women in		
		Do you allow		
		your need?		
		people in case of		
		other village	0.274	0.612 ^a *
		cooperation from		
		Do you get		
		activities?		
		planning		
		health and family		
		professional,		

***p<0.001, **p<0.05 and *p<0.10

^a Fisher's exact test.

^b T-statistics for mean difference test.

The study also found quite a large difference in the estimated index value for the components of health, food, and social networks between the study sites. The index values for health, food, water and social network were the highest for mainland households which were 0.823, 0.515, 0.938 and 0.446, respectively as against 0.494, 0.775, 0.230 and 0.111 for *char* households, respectively (Table 5.2). The highest index values demonstrated relative higher resilience capacity of mainland community in those areas than the *char* community (Figure 5.2). The index value for the component of socio-demographic and livelihood strategies varied slightly across sites.



Figure 5. 2: Index value of the major components

A considerable variation was observed between the indicators. For instance, level of education (0.894, as against 1.242), access to health facilities (0, as against 0.778), adoption of new agricultural practices (0.261 as against 0.764), ability to supply food from own production (2.12, as against 4.513), access to familial and kinship networks (0.188 as against 0.499) were low for the *char* households which contributed to limiting their resilience capacity. The index value for the dimension of sensitivity and adaptation were significantly higher for the mainland households which were 0.499 and 0.759, and 0.296 and 0.460 for mainland households and *char* households, respectively (Figure 5.3).



Figure 5. 3: Index dimension of resilience
5.5.2 Discussions

The index values for the underlying components and indicators of RCI analysis provided evidence that limited the resilience capacity of the study households. The variation in RCI indicates that resilience capacity of hazard-prone households vary across locations, both overall and more radically in relation to the particular components and indicators.

Access to health facilities is one of the important limiting factors for resilience capacity of *char* households. The *char* area is isolated from the facilities of the riverine mainland. The main form of transport for them is boat which naturally takes more time to reach the health and veterinary centre. Their low income profile and the unavailability of health services are also preventing many households to visiting local doctors. Low income means cutting back their minimum consumption requirements to pay for health care services. Basically they are deprived of many standard government services. They have limited human resources in terms of formal education (below the primary level) and skills which limit their options when seeking employment opportunities in the non-farming sectors. It is reported that human capital development is an important pathway to enhance resilience capacity (Magis 2010; Alam et al. 2004).

There are a lack of primary education facilities in the *char* areas. Children in this area need to go to the mainland area to access education facilities. The confluence of low level of education and awareness coupled with their traditional belief in recovery from sickness makes *char* households more vulnerable and reduces their resilience capacity. *Char* households have limited scope to diversify their livelihood activities. The irony of this fact is that most of the *char* households are dependent on agriculture which is most vulnerable to climate change and hazards and thus limit their resilience. However, some households are found to adopt livestock and poultry rearing, engaging them in small businesses to be more resilient and risk adversity towards natural hazards. Therefore, the enabling role should be ensured by both the government and NGOs. For example, government organizations and NGOs can provide them with livestock support or credit for having livestock since lower income households lack capital. This can serve as an important source of supplementary income.

The low resilience capacity of the communities can also be explained by the limited access to food. The situation is worse for char households than mainland households. The contribution of the 'food' component to reduce resilience capacity for *char* households is likely for two reasons: they have to struggle more to manage their food and local agricultural production is limited only being able to feed themselves for a few months. The crop cultivation is limited to very few crops due to poor soil condition (sandy soils) and land ownership (small farm size mainly due to erosion). The contribution of livestock to food security is also limited. Furthermore, they experience more natural disasters than other areas resulting in a loss of livestock and poultry. During the rainy season, they used to move to a safer place mainly in nearby embankments or an open place. They then have to sell their livestock at a lower price and in some cases lose their livestock completely due to sickness. Despite the above mentioned limiting factors of resilience, the households in the char area are increasingly adopting resilient activities such as homestead gardening, tree plantation, new cropping practices, and allow women to work outside. Many of them also found to use safe drinking water and a sanitary toilet which might be regarded as a positive move to enhance their resilience capacity.

The low resilience capacity of the households is also due to the existing low level of social networks. Due to poor socio-economic conditions and inadequate road transportation facilities, their social network, the key to social capital, was also found to be limited. Communities which have strong risk-sharing informal networks have proved to be more resilient to idiosyncratic shocks since risks can be transferred across members and time (Tompkins & Adger 2004; Narayan 1997; Moser 1996). Such informal networks typically include women's groups, religious groups and cooperative farming groups which are currently lacking in the study areas. Households in the char areas have inadequate access to financial institutions. The existence of governmental or formal financial institutions including NGO activities is insufficient. Lack of such access can limit the potential to enhance resilience of the underprivileged in a range of ways. For example, this can limit their ability to obtain the resources and technologies they might need for adaptation. After any disaster, households need financial capital to meet their basic needs. Since the char households have limited access to familial and kinship networks to gain capital during hazards, they are forced to borrow money from informal money lenders with high interest rates. To pay the loan, sometimes they need to sell their economic assets such as livestock which otherwise can provide income support and food. This ultimately contributes to reducing the resilience capacity of households. Islam and Walkerden (2015) also maintained that households' links with NGOs can promote resilience.

The households in the riverine mainland area, on the other hand, have relatively more education level, more opportunities to diversify their livelihood strategies, better access to food, health and safe drinking water, and are quick in adopting new agricultural practices due to their easy access to agricultural extension services. They enjoy easy access to financial institutions owing to the existence and proximity of such service providers, have better use of information technology for various purposes and relatively strong social networks. Altogether these factors allow them to show somewhat more resilient capacity than the char households. However, it is important to mention that all riverine households have experienced loss of some of their land and other natural resources. That reduces their production potential and employment opportunities in farming consequently increasing vulnerability. They were able to supply food from their own production to feed family members only for a few months. They experienced more food insecurity during the months of Ashar to Agrahyon (mid-June to mid-November) (see more on Chapter 6). These months mainly cover the rainy seasons in Bangladesh when opportunities for both farming and non-farming activities are reduced significantly. Due to loss of many market places and inadequate road and transport facilities as a result of erosion hazards, residents are required to travel some distance places to sell their products. Moreover, traders were not able to go to local markets, which reduced their chances of obtaining the right price of products for riverine households.

5.5.3 Limitations

The main challenge in resilient assessments is to develop robust and sound indicators and providing scores for the assessment due to the multidimensional nature and complexity of the concept (Cumming et al., 2005). Which variables should be measured in a given study of resilience is a crucial question. This study attempts to overcome this issue through extensive review of the literature and multilevel consultation with the local experts which expected to yield sound results.

The methodology of this study can be applied to estimating and comparing the resilience capacity of other rural communities due to the flexibility of the method which allows the changes of indicators and provides scores according to the circumstances of a given condition of a region, sector or community. Furthermore, the method used the household level primary data which does not suffer from the limitation of secondary data-driven methods and missing data problems. The assessment can provide potential insights and guidance in the area of impacts, adaptation and societal behaviour. Socio-economic factors are a useful guidance to decision-makers about climate change. In determining the resilience capacity, the role of socio-economic changes such as switching profession to livestock and poultry rearing, tree plantation, cultivation of new crops in the emerging char land, homestead grading, and overseas labour migration are recently boosting in those areas and cannot be overlooked. Indicators representing such changes are incorporated in the index developed in this study. It can provide a better understanding on how to build resilience to adapt to climatic change hazards through the decision-making process. The information will enable policy makers to know where policies need to be directed to build resilience and reduce vulnerability of the households.

5.6 Summary of the chapter

This chapter enhances our understanding of the socio-economic factors affecting the resilience capacity of hazard-prone rural households in Bangladesh, through developing an indicator based index. The RCI is a relative measure and the value ranges between 0-1, where the higher the value the higher the resilience capacity. The assessment method includes several features of the communities such as the socio-demographic profile, livelihood strategies, social networks, and access to food, water and health facilities that helps to identify linkages of social factors and climate change impacts, and accompanying coping and adaptation responses. Although the *char* households show relatively less resilience capacity than riverine mainland households, however, both have a low RCI value indicating their inability to demonstrate resilience. In other words, this underpins the need of strong intervention and supports from both the government and NGOs to cope, and adapt with climate change and hazards. This study identifies that access to food, water, and health

facilities, livelihood strategies and level of education have contributed to limiting the resilience capacity of the households.

The strategies for improving access to food, water and health will contribute to reduce sensitivity whereas strengthening the socio-demographic profile, diversification of livelihood activities and access to social networks will enhance adaptive capacity of the communities. Interventions need to be targeted to promote community capacity development in the area of human capital, social capital and organizational capacity that are likely to contribute to enhancing the resilience of the disadvantaged communities. Moreover, development of communication and transport networks and infrastructure is also vitally important in order to support alternative livelihoods of the households and improve their access to available services. The next chapter discusses household food security conditions due to the impacts of erosion hazard and other climate change issues.

CHAPTER SIX

Policy Options for Improving Access to Food for Vulnerable Rural Households

6.1 Chapter outline

This chapter is linked to research objective two and the results are divided into two sections. In the first part, the key factors that derive vulnerable rural household food insecurity and the pathways in which these affect are discussed. In the second part, households coping strategies and the allocation of intra-households resource distribution are discussed. The rest of the chapter is organised as follows: Section 6.2 provide introduction. Section 5.3 presents methodology including theoretical and empirical model for analysis. Results and discussions are presented in section 6.4. Section 6.5 provides some policy guidelines. In the second part, the impact of riverbank erosion on different sectors is presented in Section 6.6. Technique of developing household food security index is discussed in Section 6.7. The coping strategies of the households are illustrated in Section 6.8. The issue of intrahousehold food distribution is discussed in Section 6.9. Section 6.10 discusses the vulnerability of households to food security and Section 6.11 contains the conclusions of this chapter.

6.2 Introduction

Food security is still a prime concern in Bangladesh, even though the country has made significant improvements in food production and eradicating poverty over the last 45 years. In regards of poverty reduction, the country also showed marked progress. The rate of poverty decreased from 62% in 1988 to 35% in 2011 (BBS 2012). In contrast, population growth rate reduced from 2.4% in 1970 to 1.47% in 2011 (BBS 2012). In regards of food production particularly rice, the main staple food of Bangladesh, impressive progress has also been made in spite of frequent climatic hazards. Recent statistics showed that the rice production in Bangladesh has increased more than three times: from 16 million tons in 1970 to more than 50

million tons in 2010 (FAO 2012). This gives an indication that the country is close to achieving self-sufficiency in food production.

There remains, however, an argument that self-sufficiency in food production, i.e. availability of food at national level does not essentially assure food security at household or individual level due to unaffordability of large poor households (MacFarquhar 2011; Harrigan 2008; Schmidhuber & Tubiello 2007; Cleaver 1993). To be food secured, households' monetary and nonmonetary resources should be sufficient to get access to adequate quantities of food (Barrett 2010; Schmidhuber & Tubiello 2007). The issue of food insecurity exists largely as a consequence of limited resources, a problem affecting many households globally and in Bangladesh. Therefore, better understanding of household food security dynamics from a resource-poor rural perspective is becoming more crucial in the changing global market economy.

In Bangladesh, growing concern among the policy makers is that certain groups within the country do not have access to food needed for active and healthy life (GoB 2011). This food insecurity due to lack of access has negative consequences on people's health, productivity and well-being which can deepen poverty situation (Harrigan 2008; Chavas et al. 2005). Stiglitz (1976) argued that the likelihood of getting job and efficiency wage rate depend on the job seeker's health condition. Scholars pointed out that less consumption of calories can be a key risk factor in many chorionic diseases of late life (Wichstrom et al. 2013; Telema et al. 2005). Less consumption of calories contributes to increases people's vulnerability to sickness and infectious diseases which resulted in missed out work (Rice et al. 1985). Moreover, important aspect of human development also depends on food security (Hamelin et al. 1999).

In Bangladesh, most of the food security studies have been conducted at national level (For instance, Mishra et al. 2015; Rich et al. 2015; Akter & Basher 2014; Muniruzzaman 2013; Dorosh & Rashid 2013; Ahmed et al. 2012; Alam & Khatun 2012; Alam et al. 2010; Hossain 2010; Faridi & Wadood 2010; Shahabuddin 2010; Ali et al. 2008; Yusuf et al. 2008; Ahmed et al. 2007; Talukder 2005, among many) and household context is relatively unexplored. There is a lack of information on the factors influencing household food security especially for the vulnerable population in the country. This research addresses this limitation using cross-

sectional survey data and provides new insights on the determinants of households' food security and their livelihood status. Furthermore, this study for the first time also explores how household heads' physical health status impacts on rural households' food security. Motivation comes through the field experience that physical health status has influence on access to both farm and non-farm job in the study areas where majority of the people depend on wage earnings and other non-farm activities for their livelihoods. The research questions seek to answer are: (i) what is the livelihood status of the riverine households?; (ii) what are the factors influencing households food insecurity especially the influence of household heads' physical health status? and (iii) what are the policy options to improve food security of these hazards-prone vulnerable rural households in Bangladesh?

Although designing and implementing food policy is a challenge in developing countries like Bangladesh, evidence-based food security analyses are conducive to resource allocation, equity and sustainability of household as well as national level food security. The result will provide an informed basis to identify the factors contributing to food insecurity of particular vulnerable segments of population in the country and thus enable policy makers to formulate various effective intervention plans and policies to reduce geographical disparity of household food security.

6.3 Data analysis method

In Bangladesh, riverine households are among the poorest of the poor and most vulnerable to food insecurity and poverty (IFAD 2013; GoB 2010). Therefore, riverbank erosion affected areas were selected purposively (see Chapter 3 for details). In case of econometric analysis, the study households were not divided based on location since the results of separate analysis of riverbank households and *char* households did not provide significant difference.

Food consumption data was collected at household level through asking about the quantity of different food items (approximately 35 items) consumed over the last three days¹⁷ along with their unit price and sources (home supplied and/or

¹⁷ The accuracy of food consumption data reduces with the length of recall period (Bouis 1994). Hence, we used three day recall method which is common in the literature (Alam 2010; Reddy 1997).

purchased). Several issues were taken into consideration to estimate the available per capita household calories:

- Firstly, food supply at household level was determined by both household supply and purchase. It was converted into calories using the Food Conversion Table of the FAO (Shaheen et al. 2013) to measure the available calories for each household.
- Secondly, available calories were converted into adult equivalent (AU) ratios and the values were then comparable across households of different sizes. Household family members and guest were both excluded and included in AU depending on their availability during the recall period. Household members under the age of six were considered as children and two children were considered as one adult member in this study (Alam et al. 2010; Omotesho et al. 2006).
- Thirdly, 2, 122 kcal per person per day (GoB 2000) were set as desirable and cut off point of calories requirement (demand) to enable an adult to live a healthy and moderate active life (food secure). Finally, the difference between calories available and calories demand by a household was used to determine the food security status of households. If the households' per capita calories were found to be greater than their demand then they were considered food secure and assigned a score 1. On the other hand, those households experiencing a calorie deficit were regarded as food insecure and scored 0.

6.3.1 Empirical model

This study applied calorie intake¹⁸ method to determine the household food security (Rahman et al. 2012; Aromolaran 2010; Bashir et al. 2010; Kazal et al. 2010; Alam et al. 2010; Sindhu et al. 2008). To compute the availability of calories (C_i), the Food Calorie Conversion Table was used. A household is considered to be food secure (C_i^*) if the difference between calorie consumption and recommended daily calorie needs (γ_i) is greater than or equal to 0.

¹⁸ It is often used as proxy for all nutritional requirements for health though there may be serious deficiencies in other nutrients required for health (Aromolaran 2010).

 $C_i^*=C_i - \gamma_i$, where, $C_i^* \ge 0$ indicates that the household is 'food secure' and if $C_i^* < 0$ the household is to be considered 'food insecure'. Assuming a liner function, household food security status can be written as:

Where, X_{ij} are explanatory variables and \mathcal{E}_i is the error term which is assumed to be uncorrelated with the explanatory variables. The observed variable is food security, where $Z_i = 1$ when $C_i^* \ge 0$ and $Z_i = 0$ when $C_i^* < 0$ for ith household. Since the observed dependent variable, Z_i , is binary/discrete in nature, the food security model can be framed as a response model (logit or probit) of qualitative variable, where \emptyset_i is the probability of food security specified as:

Now, the logistic regression can be applied to this model because it directly estimates the probability of an event occurring for more than one independent variable, that is, for k independent variables (Hailu & Nigatu 2007; Fleke et al. 2005; Demaris 1992). The logistic regression model of food security can be written as:

Where, \emptyset_i is the conditional probability of food security and β_j 's are parameters to be estimated and X_{ij} 's are the explanatory variables.

In equation (iii), the dependent variable – food security – is in log odds, the results of the logistic regression can be interpreted in terms of conditional probabilities instead of log odds or odds using the formula as:

However, the estimated parameters only show the direction of the impact of the explanatory variables on dependent variable and do not provide the extent of change or probabilities. Marginal effects, on the other hand, measure the impact on the probability of observing each of several outcomes rather than the impact on a single conditional mean and are more meaningful and interpretable (Cameron & Trivedi 2009; Long 1997). Therefore, the results of marginal effects are presented in the model after testing the stability and robustness of the results.

6.3.2 Specification of the variables

The selection of variables is based on the review of the literature and field experience. The author assumed household food security to be a function of households' socio-economic and farm characteristics such as household heads' age, gender and educational attainment, size of household, adoption of livestock, and access to market and safety net program. Also included were cultivated land size¹⁹ and access to non-farm income, as a proxy of household income. Due to limited agricultural land a large number of households depend on wage earnings or other non-farm income to maintain their livelihoods. Therefore, the study also included household heads' physical health status in the model as dummy since it has influence on access to farm and non-farm jobs, where good and poor health conditions are coded as 1 and 0, respectively. To provide a score, a few techniques were adopted to minimise self-reported biased since health status is unobserved or latent variable. For example, instead of asking about respondents' health status directly, we asked whether they are fit for farm and non-farm works regularly throughout the year. The answer were then checked with how many days they were absent from the work due to sickness. If it is less than one week²⁰ then the score is 1 and 0 otherwise. Detailed description of these variables and the summary statistics are presented in Table 6.1.

The specific model for household food security takes the form as follows:

$$Y_{i} = \beta_{0} + \beta_{1}(HHAg)_{i} + \beta_{2}(HHG)_{i} + \beta_{3}(HS)_{i} + \beta_{4}(HHEd)_{i} + \beta_{5}(HLS)_{i} + \beta_{6}(HNFI)_{i} + \beta_{7}(HLO)_{i} + \beta_{8}(ASN)_{i} + \beta_{9}(AMkt)_{i} + \beta_{10}(HHHelth)_{i} + \varepsilon_{i}$$

Where, Y_i = Probability of the ith household to become food secure; (Male = 1, Female 0), β_0 = Constant, β_{1-10} = Parameters to be estimated, ε_i = Error term, HHAg = Household head age, HHG = Household head gender, HS = Household size, HHEd = Household head education, HLS = Household land size, HNFI = Household non-farm income, HLO = Household livestock ownership, ASN = Access to safety net, AMkt = Access to market, HHhealth = Household head health status.

¹⁹ Many households have large farm size but practically most of the land are in the grip of river and not suitable for cultivation therefore excluded in estimation.

²⁰ Based on the consultation with local physician one week absent from the work was considered as normal. Diseases such as fever, cough, skin infection and diarrhoea are found common in the area.

Explanatory variables	Description	Mean	Std.	Expected sign
Age of household	Years (Continuous)	45.12	14.43	+/-
head				
Gender of household	Dummy, $1 = male, 0$	0.95	0.22	+
head	otherwise			
Education of	Years of schooling	3.17	4.63	+
household head	(Continuous)			
Household size	Number (Continuous)	5.21	3.35	-/+
Cultivated land size	Decimal (Continuous)	0.56	0.88	+
Adoption of	Dummy, $= 1$ if households	0.84	0.36	+
livestock	have livestock; 0 otherwise			
Access to non-farm	Dummy, $= 1$ if households	0.63	0.31	+
income	have access; 0 otherwise			
Access to safety net	Dummy, $= 1$ if households	0.04	0.20	+
	have received; 0 otherwise			
Household head's	Dummy, = 1 represent	0.57	0.49	+
physical health	good health and 0 represent			
condition	poor health			
Household food	Dummy, 1= secure, 0=	2048	975	
security	insecure)			

Table 6. 1: Summary statistics and description of model variables.

6.3.3 Econometric consideration

The issues of multicollinearity, heteroskedasticity and the effect of outliers in the variables were taken care of which are the inherent characteristics of cross-sectional survey data. Before proceeding with model estimation, the study tried to identify multicollinearity using the correlation matrix with all the explanatory variables after running an ordinary least square regression. The correlations are found relatively low, below 0.43 in all cases. Typically, correlation coefficients of 0.7 or higher are considered as high (Kennedy 1998). Thus, correlation problems between explanatory variables can be ruled out. In order to explore potential multicollinearity which presence in the model can lead to imprecise parameter estimates (Gujarati 2003), the Variance Inflation Factor (VIF) for each of the explanatory variables were also calculated. The VIFs ranges from 1.17 to 1.71 which does not reach convectional thresholds of 10 or higher used in regression diagnosis (Maddala 1992). The Breusch-Pagan/Cook-Weisberg the test confirms that model has no heteroskedasticity problem (The null hypothesis of homokedasticity is accepted; Chisquare 13, p>0.131). The Ramsey-RESET test was also performed in order to test the accuracy of the models. The result rejected the null hypothesis of incorrect functional form that indicates relevant variables have not been omitted. In order to be sure that household health status is exogenous, Hausman endogenity was employed to verify that error term is uncorrelated with household heads' health status. The test result rejects the null hypothesis that household heads' health status is endogenous (F (1, 23); p > 0.110).

6.4 Results and discussions

The results of the study are presented in two phases: riparian households' livelihood conditions and the econometric results for the determinants of household food security.

6.4.1 Livelihood conditions

Better understanding of the overall livelihood status of the households can provide information about potential policy interventions and thus make an improvement towards household livelihood and food security. The status of households' socioeconomic and livelihood conditions are summarized below:

The study revealed that 39% and 55% of households had lost their homestead more than three times and at least once, respectively, during the last 10 years. More than 93% of households had reported to loss of employment opportunities and income from agriculture due to the hazards. Due to loss of many market places and inadequate road and transport facilities, residents go to distance places to sell their products. Moreover, traders were not able to come to local markets which reduces the chance of getting right price of their products.

In case of hygienic issues, more than 21% households were without sanitary latrine facilities whereas 47% had no safe drinking water facilities although many of them have tube-well facilities but with arsenic contamination. The distance of nearest safe drinking water source was more than 1 km. The households were also found deprived from many standard government services. About 46% households were without any electricity facility. In case of health issue around 63% of household heads fall into the category of poor health condition. Poor health condition limits their opportunities to find job in farm and non-farm sectors. Availability of health facilities was also found limited in the study areas. They had to travel a longer

distance (more than 2.5 km) to reach the nearest health and veterinary canters including public hospitals where they are supposed to get free health care.

Moreover, the existence of governmental or formal financial institutions including NGOs activities in the areas were reported to have inadequate. About 69% households reported to have no access to government financial institutions and 64% have no access to NGOs from where they can get credit. This is mainly due to their poor economic conditions where the possibility of getting back the credit is somewhat uncertain. Because riverine households' address are often changed due to change in homestead position as a result of erosion. Moreover, most of the femaleheaded households (83%) in the study areas are found widowed or divorced²¹. They are vulnerable in all aspects of livelihood characteristics in rural Bangladesh (Mallick & Rafi 2010). Field experience suggest that their opportunities to work in farming and non-farming are limited and still are not well accepted in the society, inferring gender inequalities in the labour market. This contributes to an increase in the vulnerability of female-headed households to food security.

6.4.2 Status of household food security and expenditure

In case of household food security, more than half (56%) of households fall into the food insecure category with an average per capita calorie consumption of 1867 kcal/day which is about 12 percent less than the standard minimum daily requirements. Whereas food secure household exceeded the minimum calorie requirements by 5 percent (2229 kcl/day). This shortfall of 12% substantially understates the energy deficiency of the poor. The standard deviation of the calorie demand variable is fairly high that indicates a wide range of variability across sample.

Furthermore, about 70% of their total expenditure are spent on food items and the rest for non-food items including farming and livestock (15%) and house building and/or repairing (6%) (Table 6.2). Expenditure on health care is the less priority area where they spent less than 2% of their earnings. It is mainly due to their low income and the unavailability of health service facilities in the areas. Their low

²¹ This area has one of the highest rural-urban migrations in Bangladesh. Many of the husbands who migrate to major cities as their seasonal coping mechanisms to find job did not return to their families, leading to high rate of divorce.

income prevent them to cut back their minimum consumption requirements to pay for the health care services. After fulfilling consumption demand their target is to invest in farming and house building and/or repairing.

Expenditure head	Percentage	Food expenditure head	Percentage
Food	70	Rice/ Wheat	78
Farming and livestock	15	Fish and Meat	4
Children education and Clothing	7	Egg and Milk	1
Health care	>2	Pulse, species and oil	11
House building/repairing	6	Vegetables and fruits	6
Total	100	Total	100

 Table 6. 2: Household expenditure.

The market purchase of the total value of food consumed at home stands at 75%. This indicates the vulnerability of the households on price shocks. It is reported that the lower the share of household expenditure on food, the easier it is for household to cope with price increases and shocks (Economist 2015). In case of food expenditure, the households spent about 78% on rice/wheat, the main source of carbohydrate. Therefore, it is crucial from the policy perspective to keep the price of rice/wheat reasonable so that poor people can afford. Increasing adoption of livestock and poultry by the resource-poor households not only supplement their income but also provide eggs, milk and meat for their consumption.

6.4.3 Econometric results

The results of the regression analysis $(logit)^{22}$ are presented in Table 6.3. To test the stability and robustness of the results, four alternative specifications of the model have been estimated. In the first model, core variables are included and subsequently added other relevant variables in models 2 to 4. In model 3, the non-significant variables are dropped which do not increase the coefficients of the remaining variables substantially. Goodness of fit of the models (given by McFadden Pseudo R^2) does not increase substantially from models 1 to 4 and indicates a reasonable explanatory power of the model (Table 6.3). The last specification (model 4) represents all variables and shows the best model fit in terms of the expected sign

²² STATA 12 was used to estimate the model.

and significance level. The likelihood ratio statistics (Chi-square of 342.137) indicate the strong explanatory power of the model. In other words, it rejects the joint null hypothesis that all coefficients of independent variables in the model are zero (p<0.00). The signs and degree of statistical significance of the variables do not change substantially across the different estimates hence the estimated results are stable and robust (follow discussions below):

Educational attainment

This study found as expected a significant positive relationship between household heads' educational attainment and food security (1.134, p<0.001). Past result also yielded the same results (Anik 2013; Alam 2010). It is expected that household heads with more education have wider access to non-farm jobs and the capacity to adopt better adaptation strategies in their farming which in turn increase their production and contribute to become food secure. It is reported that household heads' education level is associated with adoption of modern agricultural technology, fertiliser and better agronomic management which is key to offsetting the negative effects of changing climate (Gebrehiwot & van der Veen 2013; Deressa et al. 2009; Lin 1991). The marginal effect of education implies that a one unit (year) increase in participants' level of education will increase the probability of household food security by 1.134 while the effect on the remaining options is negligible. The same interpretation holds true for other variables.

Age of household head

The study found negative association between household head's age and food security (-0.091, p<0.10). Similar results were also found in past research (Balagtas et al. 2014; Mannaf & Uddin 2012). It is mainly due to their inability to do relatively hard work in farm and non-farm sector with the increase of age. In the study areas most of the farmers particularly small and landless farmers used to migrate to cities for few months to improve their livelihood. However, this type of migration is less likely for an aged household head which increases their vulnerability.

Variables	Maximum likelihood estimates		Marginal effect of			
		(coefficient)		model 4		
	Model 1	Model 2	Model 3	Model 4	Coeff.	Std
						error
Age of	-0 217**	-0 215**	_0 211**	-0.213**		
household head	(0.103)	(0.102)	(0.098)	(0.101)	-0.091*	0.048
(years)	(0.105)	(0.102)	(0.070)	(0.101)		
Gender of		0.101		0.105	0.0 - 1	0 0 - 1
Household head		(1.402)		(1.027)	0.071	0.874
(dummy)		()		(
Household size	1.316***	1.312***	1.317**	-1.310***	1 0 1 1 1 1 1 1 1	0.070
(AE)	(0.470)	(0.463)	*	(0.461)	-1.041***	0.379
	· /	· /	(0.468)	× ,		
Education of	1.725***	1.723***	1.728**	1.721***	1 10 4 ** **	0.400
household head	(0.572)	(0.570)	* (0.575)	(0.569)	1.134***	0.402
(year)	. ,		(0.5/5)	· ´		
Cultivated land	1.197***		1.216**	1.192***	1 000***	0.271
size (decimal)	(0.411)		*	(0.402)	1.082***	0.371
A			(0.407)			
Access to non-	1.151***	1.148***	1.153**	1.150***	1 012***	0 275
(dummy)	(0.413)	(0.411)	(0.415)	(0.410)	1.015	0.375
(dummy)			1 167**			
LIVESIOCK		1.165***	*	1.163***	1 087**	0.513
(dummy)	-	(0.410)	(0.413)	(0.431)	1.007	0.515
Access to safety		0.139	(0.415)	0.102		
net (dummy)	-	(0.345)	-	(0.647)	0.074	0.023
Access to		0.023		0.016		
market (dummy)	-	(0.109)	-	(0.103)	0.010	0.093
Household head		(012.05)		(0.100)		
physical health	1.210***		1.237**	1.211***	1 110 4 4 4 4	0.001
condition	(0.371)	-	*	(0.349)	1.110***	0.391
(dummy)	· · · ·		(0.376)	· · · ·		
Constant	10.587**	11.451**	11.461	11.563**		
	*	*	***	*		
$\text{Prob} > \chi^2$	0.000	0.000	0.000	0.000	0.000	
Goodness of fit	0.721	0 727	0.720	0.720		
(Pseudo \mathbb{R}^2)	0.721	0.727	0.729	0.730		
Log likelihood	-80.129	-81.514	-81.461	81.921		
LR (chi-square)	337.07	341.142	341.512	342.137		
Degrees of	06	08	07	10		
freedom	00	00	07	10		
Number of	380	380	380	380		
observations	500	500	500	500		

Table 6. 3: Regression results for the likelihood determinants of food security.

Note: Dependent variable: Food security.

***p<0.001; **p<0.05 and *p<0.10.

Household size

The study found inverse relationship between family size and food security (-1.041, p<0.001). This result is consistent with previous findings (Feleke et al. 2005; Bashir et al. 2010). Households with more family members tend to have lower food security. However, households endowed with more earning members are more likely to be food secured. In this study, large family size includes mainly the members who are not able to earn income such as children and aged people. The young people with income sources were often found to be separated from their family. Higher numbers of children were found for the households who had less education and not adopted contraceptive methods. Despite a tremendous progress in reducing population growth in Bangladesh, this finding sought more pragmatic role of family planning activities among this vulnerable communities.

Cultivated land size

Access to land – the most important natural resources – is considered the key determinant of the livelihood strategies of the rural poor. Rural households' income mainly derives from land. Though 32% of households in the study areas are found landless. The study found significant positive relationship between cultivated land size and food security (1.082, p<0.001). In Bangladesh, positive relationships between farm size and household food security are well registered (Faridi & Wadood, 2010; Kazal et al. 2010). The important policy intervention is required for the emerging *char* lands which was fallow previously. Scientists need to respond to develop and improve crop varieties and production technologies suitable for the emerging *char* lands in the riverbank erosion affected areas.

Livestock ownership

Furthermore, the study found livestock adoption as a significant positive impact on household food security (1.087, p<0.05). The result is in line with the findings of Rahman and Poza (2010), and Amaza et al. (2006). Livestock serves as an important source of supplementing their family income. It is indeed encouraging that households in the areas are switching their profession towards livestock, poultry and duck rearing. However, many farm households were found to use animal power for agricultural purposes including cultivation of land. This indicates their backwardness as well as inability to adopting modern agricultural practices.

Access to non-farm income

Access to non-farm income offers an important pathway to food security. It indicates the income diversification opportunities of households. This study found a significant positive association between non-farm earnings and food security (1.013, p<0.001) which supports the previous findings (Murungweni et al. 2014; Reardon 1997). However, all households do not have equal access to non-farm income. It is reported that the poor and uneducated households, and others lacking social ties rarely enjoy the access to remunerative opportunities in non-farm earnings (Barrett et al. 2010). The public services such as education and credit facilities, and communication and transport infrastructure are crucial to participate in non-farm activities which found inadequate in the study areas. Households' limited access to institutional facilities coupled with limited agricultural activities due to land loss serve as substantial barriers to participate in non-farm activities.

Household heads' physical health condition

This study found significant positive impact of household heads' health status on household food security (1.110, p<0.001). Marginal effect suggests that household heads' good health would result in an improvement in the likelihood of household food security by 0.822. It is reported elsewhere that households with ill health are more likely to be food insecure (Bernell et al. 2006). The reason behind this is that health status has effects on labour supply and productivity, farm output, and earnings (Fisher & Lewin 2013; Chavas et al. 2005). This issue is curial for the households who are mainly dependent on agriculture as in the case of this study. Poor health prevents households to participate into farm and non-farm jobs. Scholars have pointed out a range of negative health outcomes due to food insecurity including less consumption of calories such as iron deficiency anemia, obesity, poor physical and mental health (Carter et al. 2010; Stuff et al. 2004; Vozoris & Tarasuk 2003; Che & Chen 2001). Therefore, it can be said that if the observed food insecurity situation (less calorie intake) prolonged, the households will lose their productive capacity and thus fall into the victim of food insecurity leading to increased vulnerability of poverty. In other words, it could lead an unfortunate vicious cycle of poverty which starts with food insecurity (low consumption of calories leading to poor health condition leading to poor access to farm and non-farm job leading to poor income leading to poor consumption and finally drop into poverty). This issue will in turn be the main hurdle to attain long-term food security in the country unless appropriate policy put in place.

It is important to note that previous research, for example, Kazal et al. (2010) indicates the effectiveness of safety net program on household food security. Estimates of this study, however, show positive but insignificant relationship even at 10% level of significance. This statistical insignificance can be attributed due to a few number of households (4%) included in the safety net program. It might have important policy implications for household food security which underpins the coverage of safety net program in the study areas. Contrary to this, Ahmed et al. (2012) argued that access to microfinance is more effective than safety nets in helping poor households to cope with the shocks. Households in the erosion-prone areas, however, reported to have limited access to financial institutions that needs attention.

6.5 The pathways

A broad range of actions are necessary to improve and sustain the food security of these particular vulnerable communities. First, since these resource-poor households have limited access to food due to loss of productive land and subsequent effects on income and other resource endowment, direct food transfer through food aid program is one mechanism to boost access to food in the short-term. The coverage of safety net program in the study areas seems to be inadequate. Therefore, a targeted food policy intervention is yet to be developed for these vulnerable communities. Households headed up by women should get priority since they have fewer opportunities to enter into farm and non-farm jobs which make them more vulnerable to food insecurity. Intervention through income generating activities such as tailoring, handicraft, embroidering where women can be engaged need to be facilitated through proper training.

The findings of this study clearly show that education has a significant impact on household food security. In the riverbank erosion-prone area, many educational institutions have been found eroded coupled with fragile road communication limit their access to education. Targeted programs are sought to boosting primary school enrolments and human capital development in the areas.

109

Livestock ownership by the resource poor households emerged as one of the important way forward to address household food insecurity. Since the crop production environment in the erosion affected areas is somewhat unfavourable, hence livestock rearing should be encouraged with enabling policy support. For instance, government organisations and NGOs can provide them with livestock support or credit for having livestock since the poor households have lack of capital. This can serve as an important source of supplementary income.

Health status of household heads critically affects household food security which could lead vicious cycle of poverty that underpins important policy implication. It is really a big challenge to the policy makers to improve the health condition of the rural households by ensuring access to food and health care. The consequences, if the farm households become sick mainly due to inadequate calorie intake and a lack of access to health service, will be for them that they develop an inability to perform farm and non-farm jobs which will, in turn, make them vulnerable and they will become a burden to the family and country as well. The question is whether the government will be able to bring all those inactive people into a social safety net program to overcome its food security challenge. The answer should definitely be negative due to the nature of the economy which is characterised as a poor economy (developing country) confronting with various other problems such as natural disasters, climatic change issues, high population growth and poverty. Therefore, access to health services should get top policy priority in parallel to access to food in order to achieve and sustain long-term solutions to the food security challenge in Bangladesh. Providing community health services which are currently not in place, is one option to ensure their access to health service. Poor households are supposed to get free health care from public hospital. Both the government and NGOs can set out mobile health (m-health) service in the study areas along with their microcredit program. It was found that most of the households (more than 89% households) have had mobile phone ownership which raises the opportunities to provide them variety of information related to agriculture and health services easily. For instance, BRAC, the largest NGO in Bangladesh, has introduced the use of smartphones in its 'Manoshi' project for improving slum dwellers' access to maternal and child health program (BRAC 2013).

Poor institutional support including lack of credit, extension services, inadequate marketing facilities, and poor transportation facilities limit vulnerable riverine households' ability to cope with the food insecurity situation. Properly targeted income transfers and credit programs along with infrastructure and human development programs in the riverbank erosion affected areas across the country may have very high payoffs to improving food security and reducing poverty in the long run. The critical issue is to generate income and employment opportunities for these vulnerable rural households in order to ensure their access to food and other basic needs which demand well-targeted policy interventions.

6.6 Sectorial impacts of riverbank erosion on food security

Riverbank erosion has impacted various sectors of the local economy in a varying degrees which are discussed below:

6.6.1 Impacts on various sectors

Based on the empirical findings, discussion with the experts and field observations, this study sketches out (Figure 6.1) that different sectors are affected in varying degrees which together impacting on food security and livelihood of the study households.

The agriculture sector and infrastructure are severely impacted by the erosion hazard. These resulted in a decrease of employment opportunities both in farming and non-farming sectors. Education, health, water and sanitation, and fisheries sectors also experienced moderate impacts. Although the livestock sector is relatively low impacted by erosion, however, it has a great impact on household food security. All these have serious consequences on overall household food security and livelihood (Figure 6.1).



Figure 6. 1: Severity of impacts of riverbank erosion on various sectors

6.6.2 Impacts on farm sizes and household food security

Riverbank erosion and other climate change issues have great impacts on household food security of the riverine households. Since most of the households depend on agriculture for their livelihood so changes in farm size obviously has great impact on household food security. Therefore, a change in farm size is discussed first and then impact on households' food security.

This study revealed that over the years, all households have experienced loss of land that has led to changes in their farming status – these changes in farming status on the basis of land ownership are presented in Table 6.4. Results indicate that out of 39 large farm households, about 21%, 35%, 31% and 13% have become medium, small, marginal and landless farm households, respectively, during the past 10 years. In addition, out of 158 medium farm households, about 34%, 27% and 15% have become small, marginal and landless farm households, respectively. Of 123

small farm households, about 49% and 36% have become marginal and landless farm households, respectively, during the same period (Table 6.4).

Land ownership (acres)	No. of households	At present	Percentage
	10 years before	(2014)	change
	(Percentage)		
Large farm (> 7.50)	39 (10)	Medium farm	21
		Small farm	35
		Marginal farm	31
		Landless	13
Medium farm (2.51 to 7.49)	158 (42)	Medium farm	24
		Small farm	34
		Marginal farm	27
		Landless	15
Small farm (1.51 to 2.49)	123 (32)	Small farm	15
		Marginal farm	49
		Landless	36
Marginal (0.50 to 1.49)	43 (11)	Marginal farm	26
		Landless	74
Landless (>0.50)	17 (4)		32

Table 6. 4: Changes in farming status due to riverbank erosion.

Average land holdings of households are 0.56 acres²³ (small farm size is a common feature in Bangladesh), and about 32% of households are registered as landless in the study areas. Results also indicate that during the past 10 years, another 28% of households have dropped into the category of landless. In terms of land loss, farmers of large and medium sized farms are the worst affected. Farmers of small and marginal plots have also been badly impacted since agriculture is their main source of livelihood. Evidence of this can be found in the low food index value discussed below:

6.7 Household food security index

For better understanding of household food security and formulate need based policy intervention a household food security index is also developed which is discussed below:

²³ Arable land is 0.123 acres/person in Bangladesh (WB 2015).

According to Pinstrup-Anderson (2009) household-level food security can be defined and measured in different ways. A household is considered to be food secure if it has the ability to acquire the food needed by its members; however, it does not necessarily mean that individual members are food or nutritionally secure (Pinstrup-Anderson 2009). In this study, a Food Security Index (FSI) was developed to understand the household food security status throughout the year. This index was based on the Household Food Insecurity Access Scale (HFIAS) – whether all household members are able to take food three times a day (food secure) or not (food insecure) (Chatterjee et al. 2012; Coates et al. 2007). The respondents were asked three Likert scale questions regarding food security status. To develop the index, the scale 'adequate', 'inadequate' and 'scarce' were given scores of 3, 2 and 1, respectively.

'Adequate' indicates households are getting a meal three times a day without any difficulties – they are considered food secure. Households are normally not worried or bothered about taking the required calories recommended for an active and healthy life; rather, the households keep themselves busy and direct all their efforts to fulfilling the food demand of their family members, typically three times a day. 'Inadequate' represents households that take food normally twice a day and sometimes three times. They find it difficult to manage sufficient food for their family members three times a day – it is assumed that the daily total calories they consume are less than their requirements. 'Scarce' represents households that take food less than twice a day. They find it very difficult to fulfil their household food requirements, and it is assumed that the daily total calories they consume are less than their requirements. This measurement of food security will enable policymakers to understand in which months the households experience more food shortage, and thus needs-based policy intervention can be formulated.

The FSI value ranges between 3 and 1, where 3 indicates food secure and 1 least food secure. The overall FSI was estimated to be 2.06, indicating households' inability/hardship to manage family food requirements throughout the year to be regarded as 'food secure'. These households can usually manage food twice per day for their family members. There were, however, large differences in the estimated FSI of various months throughout the year. As seen in Figure 6.2, on average, households experience more food insecurity for the six months from Ashar to

Agrahyon (mid-June to mid-November). These months mainly cover the rainy seasons in Bangladesh when opportunities for both farming and non-farming activities are reduced significantly. It was observed that the study areas also experiences more riverbank erosions during this period.

Moreover, the households experience severe food insecurity during the period from Bhadra (1.83) to Ashwin (1.91) (mid-August to mid-October). The situation starts improving marginally from the month of November when the cultivation of Boro rice starts and, as a result, the scope of employment in farming expands. In the month of Falgun (February–March), households begin harvesting crops that contribute to improving their food security situation further (2.81) (Figure 6.2). Most of the households, on average, are able to manage their family food demands nearly three times a day during that period. During food shortage they practice some coping strategies which are discussed in the next section:



Figure 6. 2: Household food security situation throughout the year

6.8 Coping strategy index

The frequency and severity of coping strategies can be considered as an alternative indicator of food insecurity and vulnerability. According to Berman et al. (2015) coping strategies can shape the availability of future adaptation options. Therefore,

there is a need to identify and analyse coping strategies for the most vulnerable riverine rural households in Bangladesh to mitigate the effects of climate change and hazard shocks and their effects on household nutritional status.

Households in the study areas were able to supply food for their families from their own production only for a few months: on an average 3.2 months. Most of the households depend on the market to purchase their necessities, and this is affected by their income-earning capacities. Households' income earning capacities, on the other hand, become limited at certain times due to lack of employment opportunities. These circumstances trigger them to adopt some sort of coping strategies based on their experience, knowledge and asset position to reduce the food shortage. According to Maxwell et al. (2003) there are mainly two types of coping strategies: one is consumption based which is related to food consumption today or tomorrow, and another is non-consumption which is linked to sale of assets and so on.

For this study, coping strategy questions were constructed through literature review, focus group discussions and pre-testing of the interview schedule. Finally, there were 13 questions related to coping strategy and households used at least one of these strategies. The respondents were asked about the frequency of use of these strategies based on 4–point Likert scale (Likert 1932). The scale was frequently, occasionally, rarely and not use. To rank the household coping strategies a Coping Strategies Index (CSI) was developed. The corresponding value of the scale 'frequently', 'occasionally', 'rarely' and 'not use' were 3, 2, 1 and 0, respectively. The CSI was constructed as follows:

$$CSI = \frac{X_{iF_i + \dots + X_{nF_n}}}{N}$$

Where, Xi = Scale value at the ith priority of the strategy, Fi = Frequency of responses on the strategy, n = Number of strategies in the parameter, N = Number of respondents, i = 1, 2, --n.

Considering the consequences of these strategies, this study was categories the index value into three groups, namely, consumption base, borrowing and selling. As shown in Table 6.5, the most common coping strategy was to rely upon less expensive or less preferred food (2.36), followed by migration (1.65) and rely on casual labour for food (1.42).

Sl. No.	Coping strategy	Index value	Category
1	Reduce amount of food per meal	1.31	Consumption
2	Reduce number of meals per day	0.80	based
3	Go bed without food	0.38	
4	Rely upon less expensive or less preferred		
	food	2.36	
5	Reduce buying children food (i.e. milk)		
	from market	1.18	
6	Purchase food on credit	1.40	Borrowing
7	Borrow money from NGOs/GB/money		
	lenders	1.09	
8	Borrow from relatives/friends and		
	neighbours	0.43	
9	Rely on casual labour for food	1.42	Selling
10	Sell labour in advance	0.83	
11	Sell cattle/livestock/land and other assets	1.02	
12	Spend money from deposit	0.70	
13	Migrate to city or other area	1.65	

Table 6. 5: Coping strategies to address food shortages.

The consequences of these coping practices are manifold. For instance, reduction in consumption base coping strategies might have long-term negative impacts on their health and wellbeing: they may no longer be able to keep themselves fit and healthy for farm and non-farm jobs and ultimately become a burden on society. Selling assets mainly includes selling livestock and poultry which are important sources of household income, and this can have negative impacts on sustainability of future household food security. However, selling assets is a common practice by the poor farmers in Bangladesh during times of hardship. Paul (1998) also found it to be a coping practice of drought victims in North Bengla, Bangladesh. Borrowing strategies might be conducive to meet up their food demand in short-term. However, it could be detrimental effects of borrowing money from money lenders: they usually charge higher interest rates. This trap sometimes make them compelled to sell their valuable assets such as land with low price to make a payment of loan.

6.9 Intra household food distribution

Although most of the study households had experienced food insecurity situation during a certain period of a year. However, all household members within the family were not equally treated to get access to food. This discrimination was mainly the result of long-practiced culture in the society in Bangladesh. Women in the areas are usually the last person in the family to eat and they consume the least amount so as to ensure food security for other family members in their household. These people sometimes go for days without food. As a result, they become more vulnerable to malnutrition. In household food distribution, the first priority goes to the male members, in other words, who earn income. Among women, aged women who have no income nor included in the social safety net programs become the last person to get access to food. However, some families where women are educated and have access to income sources are found to get equal access to household food.

6.10 Vulnerability to food security

Loss of land is a recurrent phenomenon in the riverine areas. Therefore, the households who depend on agriculture will be the most vulnerable to food security. Apparently, large and medium farmers are not in vulnerable group due to their land position that enables them to produce to meet up their family consumption. Considering the current trend of erosion, it can infer that all the farm households (large, medium and small) will fall into the vulnerability group within a passage of time due to erosion which is out of their control. They can also loss land for many other socio-economic reasons, but erosion is the main cause. Previous discussion indicates that all households in the study areas have experienced loss of land that has led to changes in their farming status and thus contributes to increase their vulnerability.

Another vulnerable group are the households who have little or no education and possess poor health. These qualifications prevent them to take the opportunity to seasonal migration to improve their food security and livelihoods, since there is lack of employment opportunities. The aged people who are neither able to do work nor included in the safety net programs are another vulnerable group in the society.

The women headed households are also most vulnerable groups. They have relatively limited scope to get employment both farming and non-farming. Many of them do not get any support from their absent husband to maintain their family that pushed them most vulnerable situation.

Scholars argued that the options to improve the household food security were either to create employment opportunities to increase per capita income and/ or ensure that households have the resources and capacity to produce their own food through farming (Musemwa et al. 2015). However, the latter option is not feasible for many households in the study areas due to recurrent land loss. Therefore, creating employment opportunities is crucial to improve their access to food especially during rainy seasons when there was limited scope of employment both farming and nonfarming. Government social safety net program such as food for work, vulnerable group feeding (VGF) might be appropriate options in such a situation.

6.11 Summary of the chapter

The first objective of this chapter was to determine the factors influencing vulnerable rural household food insecurity in Bangladesh. The analysis of survey data of 380 households using logit model has provided a better understanding of the key factors that derive vulnerable rural household food insecurity in Bangladesh. The study reveals that the riverbank erosion-prone areas are poor in a number of areas such as infrastructure, assess to education and health services, access to market and non-farm activities and availability of public utilities like electricity and safe-drinking water which contribute to increase their vulnerability. The study has found several factors serve as a driver of households' food insecurity such as household heads' level of education, household size and cultivated land holdings, livestock ownership and access to non-farm income. Study also found new evidence which suggest that physical health status of household head is a key significant influencing factor for household food security. The rest of the variables are not statistically significant but have the expected sign.

The second objective was to develop a household food security index based on HFIAS. The information is crucial for potential policy interventions and thus make an improvement towards household food security and livelihood. The overall FSI value of 2.06 out of 3 indicating households' inability to manage family food requirements throughout the year to be regarded as 'food secure'. The households experience severe food insecurity during the month of Bhadra (1.83) to Ashwin (1.91) (mid-August to mid-October). Important coping strategies include reduction of food consumption and household expenditure, borrowing money and selling assets. Government interventions such as a safety net program, which presently covers only 4% of households, need to be increased to improve the food gap. Therefore, in the

119

short-term, a targeted food policy intervention needs to be developed for these vulnerable communities to boost access to food, particularly in the most food insecure months. This empirical evidence enables the policymakers to formulate well-targeted food policies to improve food security and the livelihoods of vulnerable riverine households across Bangladesh. The next chapter discusses the response strategies of the households in the face of changing climate and hazards.

CHAPTER SEVEN

Climate Change Perceptions and Local Adaptation Strategies of Hazard-Prone Households

7.1 Chapter outline

This chapter is linked to research objectives number three. In this chapter, households' perceptions of climate change and variability and their adaptation responses are discussed. The information will enable policy makers to identify local adaptation strategies and to incorporate them in sectoral and other planning activities. The intervention strategies could thus assist the community in adopting effective, logical and sustainable adaptation practices that will enhance their resilience. The chapter is organised as follows: Section 7.2 describes the introduction of the chapter. Section 7.3 presents the literature review. Section 7.4 presents the methodology. The results are illustrated in Section 7.5 and Section 7.6 contains the conclusions of the chapter.

7.2 Introduction

The livelihood of resource-poor rural households in developing countries such as Bangladesh depends largely on agriculture but this is most vulnerable to climate change and variability. The households' capacity to adapt to the compounding influences of climate change, which can affect households' resources and resilience, is uncertain due to poor socio-economic conditions (Wood et al. 2014; Lobell et al. 2008; IPCC 2007; Adger & Vincent 2005). Therefore, adaptation measures are important to help the local communities to cope with extreme weather conditions and associated climatic variations (Niles et al. 2015; Gandure et al. 2013; Rosenzweig et al. 2013; Adger et al. 2003). Adaptation strategies are context specific and change over time from area to area and even within particular societies (Malone 2009; Smit & Wandel 2006). This chapter focuses on perceptions of climate change and local level knowledge of adaptation of the study households: such knowledge is important for enhancing vulnerable households' resilience in the face of hazards and for coping with climate change and variability. The IPCC has placed local knowledge of adaptation in the centre of discussions to formulate adaptation options (IPCC 2007).

The livelihoods of riparian rural households are severely affected because the erosion increases their vulnerability to food insecurity and reduces their ability to alleviate poverty (IFAD 2013; Huq & Rabbani 2011; GoB 2010; Lein 2010; Hutton & Haque 2004). Scholars have argued that the vulnerability of rural households depends on access to and use of livelihood capital, namely, natural, physical, financial, human and social (DFID 1999; Carney 1998; Chamber & Conway 1992). The linkages between households vulnerability context and livelihood capital enables policy makers to understand of which assets are most affected by the vulnerability context and how better way people can be supported to build up their livelihood assets and become more resilient. Therefore, it is crucial from policy perspective to identify and to understand vulnerability and possible adaptation strategies of marginalised riparian communities, which could mitigate the effects of an adverse climate.

This chapter addressed the following research questions: (i) what are the perceptions of hazard-prone rural households to climate change and climatic variability?; (ii) what are the perceived impacts of riverbank erosion and other climate change issues on the livelihoods of the households; and (iii) what local adaptation strategies can the resource-poor households adopt to enhance their resilience?

7.3 Review of literature

The brief outlines of the existing research on households' perception of climate change and adaptation strategies are summarised here. Adaptation to climate change and variability refers to the adjustments in human-environment systems in response to actual and/or anticipated climatic conditions to avoid or to alleviate related risks or to realise potential opportunities (Wheeler et al. 2013; Smit et al. 2000; IPCC 2001). Climate change affects countries, regions and communities in different ways and thus they differ in terms of their adaptive strategies. The factors responsible for the variation in adaptive responses across regions are the agro-ecological system, socio-economics, climatic impact, and existing infrastructure and capacity (Brulle et al. 2012; Fraser et al. 2011; Adger et al. 2009; Berry et al. 2006). The adaptation process

requires the capacity to learn from previous experience to cope with the current climate, and to apply these lessons to cope with future climate change (Adger et al. 2005b).

Perception and the adoption of adaptation strategies are the two key components of adaptation process (Maddison 2007). Farmers first need to perceive the impact of changes in the climate in order to take appropriate adaptive strategies to mitigate their vulnerability and to enhance the overall resilience of the agroecological systems (Arbuckle et al. 2013; Bryan et al. 2009; Reid et al. 2007). Misleading perceptions can induce inappropriate adjustment measures (Taylor et al. 1988). Farmers who perceive the potential consequences of climate change are more likely to support policies that aim to address it (Arbuckle et al. 2013; Niles et al. 2013; Gordon et al. 2013).

Numerous research has indicated that any attempts to elicit adaptive behaviour should come through an understanding of how climate variability is perceived by farmers and what shapes their perception (Weber 2010; Mertz et al. 2009; Slegers 2008; Shisanya & Khayesi 2007; Maddison 2007). Scholars have mentioned that adaptive capacity is influenced by factors such as knowledge of and perception about climate change, assets, and access to appropriate technology, institutions and policies (Brulle et al. 2012; Haden et al. 2012; Hisali 2011; Mertz et al. 2009; IFAD 2008; Adger et al. 2003). The significant link between the perception of climatic variability and the adaptation process has been examined in several countries; for example, in the USA (Arbuckle et al. 2011), Canada (Bryant et al. 2012), Australia (Wheeler et al. 2013), Vietnam (Dang et al. 2014; Schad et al. 2012), India (Vedwan & Rhoades 2001), Malaysia (Alam et al. 2012), Sri Lanka (Esham & Garforth 2013), The Philippines (Predo 2010), Nigeria (Apata et al. 2009), Tanzania (Slegers 2008), Tunisia (Mertz et al. 2009), Ethiopia (Deressa et al. 2011) and South Africa (Gandur et al. 2013).

Studies in Bangladesh in this context are limited. Most adaptation studies are drought focused (see, for example, Alam 2015; Alauddin & Sarker 2014; Sarker et al. 2013; Habiba et al. 2012; Shahid & Behrawan 2009; Ahmed & Chowdhury 2006; FAO 2006). Few studies focus on low lying and saline-prone areas (Rashid et al. 2014; Anik & Khan 2012; Hossain et al. 2012; Rawlani & Sovacool 2011; Ayers & Huq 2008). All these studies provide useful indicators for adaptation policy.

However, these might not be effective and applicable to other hazard-prone communities due to the heterogeneity of the impact of the various hazards and the socio-economic conditions of the households and therefore their responses vary. This is particularly important for the most vulnerable riparian communities who are poorly resourced. There are studies on displacement and the socio-economic impact of riverbank erosion in Bangladesh (see, for example, Ahmed 2015; Lein 2010; Hutton & Haque 2004, 2003; Zaman 1991, 1989; Makenro 2000; Elahi 1989; Rogge & Haque 1987; Greenberg 1986; Hossain 1984). However, there is a lack of in-depth empirical research on how the resource-poor hazard-prone households perceive climate change and variability and how their perceptions are linked to their local adaptive responses. Scholars have argued that local level adaptation knowledge is a key to promoting the resilience of vulnerable communities (Hiwasaki et al. 2014; Alexander et al. 2013; Green & Raygorodetsky 2010; Ellen 2007; Nyong et al. 2007). In terms of policy making, farmers' local knowledge of adaptation strategies will have immense significance if they are supported by relevant government organizations, NGOs and research for the overall sustainability of the adaptation process in the country.

7.4 Data analysis method

Detailed description of the study areas and sampling have been presented in Chapter 3. Here the data collection techniques of perception of climate change and adaptation strategies are discussed.

Perceptions of change in various climatic variables were collected using a 4 point Likert scale (Likert 1932): increase, decrease, remaining same and don't know. In the case of the perceived impact of climate change and of hazards, a respondent's self-elicitation status was considered: the replies to the questions ranged from high to low impact. In the case of adaptation strategies, the respondents were asked about their range of practices. The rainfall and temperature data for the observation station nearest to the study areas was obtained from the Bangladesh Meteorological Department (BMD 2014).

7.4.1 Data analysis

Statistical analyses such as descriptive analysis and a 5-year moving average for the temperature and rainfall data were conducted to compare these with household perceptions about climate change parameters. General linear regression was performed to obtain the mean rainfall and temperature trends. A non-parametric Chi-square test was performed to identify differences between the farming groups when selecting adaptation strategies.

7.5 Results and discussions

7.5.1 Households' perception of climate change and variability

The respondents indicated that they observed changes in the climate and in the extreme events over time (Table 7.1). In the case of the annual mean temperature and rainfall over time, 91% of the respondents believed that the former had increased and 89% believed that the latter had decreased (Figure 7.1). None of the respondents perceived an increase in rainfall and a decrease in temperature. They observed abnormalities in rainfall timing and distribution which has serious consequences for their production plans. These perceptions are consistent with macro-level evidence of climate change and variability in Bangladesh (WB 2013; IWFM 2012).



Figure 7. 1: Households' perception of the annual mean temperature and rainfall

As well, 83% and 81% of the household heads suggest that the frequency of flooding and cyclones respectively had increased. A similar response was also observed in regards to droughts – more than 91% of the household heads believed the frequency of droughts had increased. However, only 67% and 87% of household heads reported a decrease in the availability of groundwater and surface water,
respectively. Winter and summer periods were perceived to have increased and decreased by 90% and 93% of households, respectively. Finally, 96% of household heads said that the severity of riverbank erosion had increased (Table 7.1).

	Respondents' response						
Climate parameters	Increase No change D		Decrease	Don't know			
	(%)	(%)	(%)	(%)			
Drought	91	5	0	4			
Availability of groundwater	0	7	67	26			
Availability of surface water	0	4	87	9			
Severity of riverbank erosion	96	0	0	4			
Frequency of flood	83	10	0	7			
Frequency of cyclones	81	7	10	2			
Winter period	2	2	90	6			
Summer period	93	5	0	2			

Table 7. 1: Perception of climate change parameters.

Household perceptions of climate change and variability were also supported by the observed scientific data. An upward trend of annual mean temperature from 1980 to 2014 was found in the study areas (Figure 7.2). The Institute of Water and Flood Management reported an increasing trend in mean temperature during the same period in Bangladesh (IWFM 2012).



Figure 7. 2: Annual mean temperature in the study area

Mean rainfall data, however, showed a slightly decreasing trend over the same period which is consistent with households' perception (Figure 7.3). IWFM (2012) indicated that rainfall in the pre-monsoon and post-monsoon seasons had increased in Bangladesh, but it decreased in the monsoon season.



Figure 7. 3: Annual mean rainfall in the study areas

It should be noted that the annual mean temperature and rainfall in Bangladesh are about 26°C and 2540 mm, respectively. The maximum summer temperatures vary between 38°C and 41°C (BBS 2012). The discussion above indicates that households in the study areas are conscious of local climatic changes and variability which ultimately guide them to adopt adaptation strategies in order to minimize the adverse effects of various climatic changes and hazards, including riverbank erosion.

7.5.2 Perceived impacts of climate change and hazards

The perceived impacts of riverbank erosion hazards and other climate change issues are broadly characterized based on the capital assets on which the households' livelihood depend, namely, natural, physical, financial, human and social (Table 7.2). It is, however, important to note that successful adaptation processes to mitigate the adverse effects of climate change and climatic variability depend largely on access to and the judicial use of these capital assets.

sets				Farm category				
fo Impact/risk		Description	Large (N= 45)	Medium (N= 107)	Small (N= 127)	Landless (N= 101)		
	Food security	Food insecurity and malnutrition						
	and	increased due to low production	х	x	XXX	XXX		
	malnutrition	and income						
Dital	Unemployment	-	x	XXX	xxx			
Human ca	Disease/health condition	x	x	xx	xx			
	Migration	Induced seasonal migration to cities and other places due to a lack of employment	-	x	XX	XXX		
xial ital	Educational institutions	Many educational institutions have been damaged or eroded	xx	XX	XX	XX		
Soc	Religious institutions	Religious institutions damaged	x	х	Х	х		

Table 7. 2: Perceived impacts of climatic change and hazards.

Medical facilities		Access to health services reduced	x	х	XX	xx
	Farmers to farmers co- operationLimited co-operation among farmers' groups. Small and landless farmers hardly get help from affluent farmers		-	x	xx	xxx
	Organizational involvement	Limited involvement with different organizations	-	х	XX	XXX
apital	Credit facilities	Access to formal and non-formal sources of credit reduced	-	х	XXX	xxx
l c	Market access	Access to market reduced	XX	XX	XX	XX
Income from		Income from agriculture reduced due to loss of land, crops and yield	xxx	XXX	XXX	xxx
Fir	Savings	Reduced the ability of savings	XX	XX	XXX	XXX
al	Homestead	Loss of homestead property	XX	XX	XX	XXX
pit	Latrine facility Deteriorated latrine facilities		-	-	XX	XX
l ca	Transport	Deteriorated transport facilities		XX	XX	XX
ica	Electricity	Deteriorated electricity facilities		х	XX	XXX
ıys	Market place	Loss of market places	Х	Х	Х	Х
łd	Embankment	Damage to embankment	Х	х	х	Х
	Land	Land loss	XXX	XXX	XXX	XXX
	Water	Reduced availability of safe drinking water	х	х	XX	xx
al capital	Livestock	Shortage of fodder and poor animal health	xx	XX	XX	xx
	Fisheries	Reduced pond areas		XX	XX	XX
tur	Forestry	Loss of trees	XXX	XXX	XX	XX
Nai	Soils	Soil quality deteriorated except emerging ' <i>char</i> land' where no agriculture was possible previously.	x	x	x	x

Note: xxx = high impact, xx = medium impact, and x = low impact

Agricultural production depends mainly on physical and natural capital, which experienced varying degrees of impact. Households' human capital in terms of education and skills, social capital in terms of access to health facilities, social bonding and organisational involvement, and financial capital in terms of access to NGOs and government financial institutions were found to be affected. This could limit their coping ability and push them further into vulnerable conditions.

The impact on human capital in terms of reduced food security and a decline in employment were registered as high mainly by the small and the landless farmers as these induced them to migrate to cities and other places to improve their livelihoods. Many institutions such as schools and hospitals were found to be eroded. Residents have to travel a longer distance to reach the school and health centre. Access to education and health facilities is one of the limiting factors of enhancing resilience of households. Therefore, investing in education and health facilities in the study areas should be in top policy priority.

All farming groups perceived a high impact of riverbank erosion and other climate change issues on agriculture which is the main source of their livelihood. To increase their resilience, appropriate adaptation in agriculture is necessary. The most important perceived impacts mentioned by the small and the landless farmers were a loss of land and homestead which increase their vulnerability. They also perceived a high impact in access to credit and market. The organisational involvements by bodies such as NGOs were found to be limited. NGOs are not interested in providing them with credit because many of them had no permanent residence. All farming groups perceived that climate change impacts on crop and yield loss, causes a decline in soil fertility and creates a scarcity of safe drinking water which all affect their livelihood (Table 7.2). However, they consider the new '*char* land' as a blessing for them since the advent of new crop varieties and technology has enabled agricultural activities where such land remained fallow previously.

7.5.3 Households' adaptive responses

The study revealed that, despite the apparent difficulties of riverbank erosion and climate change issues, all the resource-poor households were attempting to sustain and to improve their livelihoods through a range of adaptation strategies. The study identified 15 farming and non-farming adaptation strategies which were practiced by the respondents based on their long-term knowledge and perceptions of climate change (Table 7.3).

Most of the households adopted more than one strategy. Based on the respondents' main choice, the most common adaptation practices are changing plantation time, cultivation of pulses, cultivation of spices and oil seeds, homestead gardening, tree plantation and migration (Table 7.3).

A .1	Responses		Farm category					
Adaptive measure	* (%)	Large	Medium	Small	Landless	Comments		
Agricultural adjustment				•				
Change planting time	8	Х	Х	Х		ILA		
Cultivation of pulses	11	Х	Х	Х		ILA		
Cultivation of wheat	1	v	V	v		ПА		
and other crops	4	Λ	Λ	Λ		ILA		
Tree plantation	6	Х	Х			ILA/PA		
Cultivation of spices	10	v	v	v	v	Π Δ/ΡΔ		
and oil seed	10	Λ	Λ	Λ	Λ	ILA/I A		
Cultivation of local	5	v	v			ΠΔ		
Aman rice	5	Λ	Λ			ILA		
Cultivation of	6	v	v	v	v	ΠΔ		
vegetables	0	Λ	Λ	Λ	Λ	ILA		
Cultivation of HYV								
rice varieties (e.g.,	8	х	Х	х		ILA		
BRRI-28, 29)								
Livestock rearing	7	Х	Х	Х	Х	ILA/PA		
Poultry rearing	5		Х	Х	Х	ILA/PA		
Duck rearing	3			х	Х	ILA/PA		
Homestead gardening	5			Х	Х	ILA/PA		
Non-agricultural								
Migration	12			Х	Х	ILA		
Off-farm work (van,								
rickshaw, tempo	7			х	Х	ILA		
driving)								
Petty business	3			Х	Х	ILA		

Table 7. 3: Adaptation strategies of the households in the study areas.

*According to main adaptation strategies although there were multiple options.

ILA = Individual level adaptation based on experience and knowledge.

PA = Individual level and planned adaptation supported by government organizations and NGOs.

Adaptation strategies were, however, shaped by farming category. A Chisquare test was conducted to see whether there were differences between the farming groups in the adaptation strategies adopted. A significant difference was found (χ^2 test, p<0.003). In the second stage, we conducted a post-hoc analysis to see the location of the difference. The result indicates that non-agricultural adaptation was practiced mostly by small and landless farm households (p<0.001) while agricultural adjustments were practiced mainly by large and medium farms. This indicates that wealthier farmers are in a better position to respond to the challenges posed by climate change and variability through adopting different strategies in agriculture whereas small and landless farmers have few choices. Households received relatively little technical and financial supports from government organisations and NGOs for some adaptation practices (Table 7.3).

Crop cultivation was found to be diversified in the study areas. In the past, farmers rarely cultivated horticultural and cereal crops, and large parts of their farmland remained fallow in the dry season. Farmers with cultivable land were found to be adopting the HYV rice and wheat varieties as part of their response to the changing climatic conditions. In particular, they were cultivating spices and oil seeds in the newly formed *char* lands which had remained fallow due to the unavailability of crop varieties suitable for such land previously. In responding to the adverse effects of climate change, households were changing the planting times of their crops. Most of the land in the *char* areas and/or near to the river is subject to water logging and flooding during the rainy seasons. But the crops are now cultivated in a way that enables harvesting to be done before a hazard can arise. This adjustment evolved from long-term local knowledge and perceptions about the climate. Vegetable cultivation appeared to be the most common adaptation strategy in the study areas: for example, small and landless farmers cultivated different types of short duration winter and summer vegetables.

The continuous loss of land through riverbank erosion is the main problem for the households. In order to ensure the sustainable use of the available land, households were practicing homestead gardening and tree plantation, particularly the small and landless farmers (Table 7.3). Homestead gardening provides a continuous supply of nutrients in the food chain and can be an important source of income. Both government organisations and NGOs play an enabling role in this regard by providing improved technologies and skills as well as encouraging households to adopt this strategy. However, the small and the landless farmers have limited access to financial institutions and extension services. They were undertaking small businesses such as grocery shops, a tea stall and retail sales that require less capital. Many of them had taken up driving as their occupation in the face of diminishing employment in farming. Government organisations and NGOs can play an enabling role in improving their livelihoods by providing training and financial support in this regard.

Migration, both seasonal and permanent, was also found to be an important adaptation strategy, especially for the small and the landless farmers. Households with limited agricultural land used to migrate in search of alternative livelihoods for a few months. This temporary migration is very common in the study areas, especially during the rainy seasons when there is limited scope for farming and nonfarming employment. However, it is encouraging to find that households are adopting different activities such as livestock, poultry and duck rearing on their land to secure their livelihood by supplementing their income. This local level knowledge of adaptation is crucial for policy makers to support and promote adaptation strategies, and to turn them into effective and sustainable action. Eriksen and Lind (2009) argued that in order to be effective, an adaptation must take place at the local level rather than being a nationally imposed option.

It is not necessary for all of the strategies to relate directly to climate change. For example, the adoption of new crop varieties might be introduced simply as a way to increase production and household income. Besides, there might be other external factors that stimulate adaptation strategies such as scientists responding with new crops and varieties suitable to local conditions like the BRRI-28, BRRI-29 and BRRI-45 rice varieties, and government and NGOs disseminating information to farmers. However, since these resource-poor hazard-prone rural households are responding to climatic events and other opportunities (e.g., cultivation in the char lands) by adopting strategies which are based on their local knowledge and experience guided by climate change perceptions, they can be treated as climate change adaptation strategies. Scientists will continue to develop crop varieties, highvalue crops and technologies suitable to local conditions, especially in the char lands, to accelerate the adaptation process. The Bangladesh Government should strengthen the existing extension network to provide information on successful adaptation strategies and other agricultural services to the grass roots stakeholders through its extension service.

7.5.4 Access to information and adaptation

Access to Information and Communication Technology (ICT) can play a vital role in providing information to the vulnerable households in the study areas where road and transportation communication system is inadequate. It is reported that access to climate information can be effective and contribute to reduce vulnerability of rural livelihoods to climate variability (Troy 2008; Patt et al. 2007; Ziervogel 2004; Patt &

Gwata 2002). Availability of climate information is important as it could contribute to improve management of climate variability and change, and thus, adaptive capacity (Matarira et al. 2012). Scholar argued that information assists farmers in deciding which agricultural technologies and adaptation mechanisms may be most useful in responding to weather variability and change (Ziervogel & Ericksen 2010). Thus, it is important for the policy makers and the development practitioners to understand the kind of ICT that rural households have access to; in order to determine appropriate ways of providing cost-effective information services to rural households.

7.5.5 Access to and uses of ICT by the study households

Exploring access to common ICT equipment by rural households in the study areas involved TV (both colour, and black and white), radio, computer, the Internet and mobile phone (Figure 7.4).



Figure 7. 4: Possession of ICT devices by rural households (multiple options)

Figure 7.4 shows that a mobile phone was the most common ICT device possessed by the rural households (86%). Only 1% of the households had access to a computer with a modem for the Internet access. The study by Ullah (2010) in the coastal areas of Bangladesh found that around 41% of households had access to a radio in 2006: this was used mainly to receive warning and forecasting information. In this study household access to a radio was only 9%, ranking it third of the ICT devices. Most strikingly, the figure shows an extreme inequality of access to ICT

when the mobile phone is excluded. In other words, this represents a digital divide for the rural households. Therefore, mobile phones contribute to reducing the digital gap (between the haves and the have nots) among the study rural households. The results indicate that more than half of the rural households (58%) had more than one mobile phone with about 2% having five or more mobile phones.

The households in the study had less contact with the agriculture extension officers and had less farmer-to-farmer contacts and less involvement with different organizations from which they can receive information and assistance. Hence, they need information and communication sources that can meet their demand for necessary information. Most households in the study areas possessed mobile phones, however, they were not receiving information related to agriculture and rural development through it. This study suggests that mobile phone can be a viable way of disseminating information easily relating to agricultural production, market prices, weather forecasts, climatic hazards and health issues for large rural farm households than other ICT where households have heterogeneous access.

7.6 Summary of the chapter

Despite the recognition of the need for policies and programs to implement and facilitate adaptation strategies, there is still a lack of information about local measures that can reduce households' vulnerability. This chapter has presented local level adaptation strategies in relation to hazard-prone resource-poor rural households' perception of climate change and climatic variability. All of the households perceived changes in the climate and extreme events, particularly abnormal rainfall in terms of timing and distribution which has serious consequences on production plans. Climate data also supported households' perception of climate change and climatic variability. The households perceived the various impacts of erosion hazard and other climate change issues on livelihood capitals, including loss of land, crops and yield, homestead, pond areas, trees, access to education and health facilities, and infrastructure resulting in an increase of livelihood vulnerability. Households respond through adopting farming and non-farming adaptation strategies to build resilience based on their local knowledge, climate change perception and farming status. Significant differences are observed among farm household groups when choosing adaptation strategies: small and landless farm households adopt non-

135

farming adaptation practices mostly. The important strategies include adopting new crop varieties, changing plantation time, homestead gardening, tree plantation, migration and changing profession to livestock, poultry and duck rearing.

This information will enable policy makers to identify local adaptation strategies and to incorporate them in sectoral and other planning activities. The intervention strategies could thus assist the households in adopting effective, logical and sustainable adaptation practices that will enhance their resilience. The next chapter describes the factors affecting adaptation strategies and the barriers to adaptation of the study households.

CHAPTER EIGHT

Determinants of and Barriers to Adaptation by Resource-Poor Households

8.1 Chapter outline

Farm level adaptation strategies are the key to reducing climate change impacts on agriculture, food production and the vulnerability of rural households. In this chapter, the determinants of and barriers to adaptation of the resource-poor households are illustrated using economic techniques. The information will enable policy makers to identify the factors that influence household adaptation choices so that effective intervention policies can be formulated to enhance their resilience. The reminder of the chapter is organised as follows: after introduction in Section 8.2, a review of relevant empirical evidence is presented in Section 8.3. Section 8.4 presents the brief methodology of the study. The results are discussed in Section 8.5 and Section 8.6 provides a conclusion of this chapter.

8.2 Introduction

Bangladesh is most vulnerable to climate change (WB 2013; IPCC 2007) which poses a major risk to the lives, livelihoods and food security of 64% of the rural population who depend on agriculture (BBS 2012). Scholars have put a high importance on the adaptation to climate change as one policy option for reducing the adverse effects of climate change so as to protect the livelihood and food security of poor farmers (IPCC 2014; Gandure et al. 2013; Wheeler et al. 2013; WB 2013; Lobell et al. 2008; Adger et al. 2005b).

Farmers' adaptation strategies and responses vary, however, between the different agro-ecological contexts and are unevenly distributed depending on socioeconomic and institutional factors, climatic impact and infrastructure (Brulle et al. 2012; Adger et al. 2009; Bryan et al. 2009; Deressa et al. 2009; Maddison 2007; Berry et al. 2006). It is suggested that adaptation options need to be assessed at household and community levels in order to meet the development goals of poverty alleviation and food security (Thornton et al. 2010). Adaptation strategies can be classified in different forms such as planned and autonomous (spontaneous), structural and non-structural, and hard and soft (IPCC 2001). Planned adaptation requires intervention by government and/or regional, national and international organisations to support and/or enhance responses by farmers and organisations (Shaw et al. 2013). Autonomous adaptation actions are those undertaken by the affected people without planned intervention (IPCC 2007; Smit et al. 2001). These generally occur through private agents such as farmer or agricultural organisations (Shaw et al. 2013; Seo 2011). Poor households' autonomous adaptation strategies are often overlooked in international and national efforts to manage the impact of climate change (Christoplos et al. 2009). But these strategies can be influenced by a range of factors and that information is crucial for identifying appropriate options for enhancing adaptation. A lack of successful adaptation will make the households more vulnerable to poverty and food insecurity.

Farmers in Bangladesh have experienced a range of climatic hazards, including riverbank erosion, and have made adaptation decisions. Some argued that adaptation research should focus on the most vulnerable groups or those with the least adaptive capacity (Hulme et al. 2011; IPCC 2007). However, the factors influencing hazard-prone resource-poor households' adaptation strategies and the barriers to adaptation have not been explored so far. These are crucial to formulating and implementing an effective and sustainable adaptation policy in Bangladesh. Moreover, recent literature has indicated that farmers' access to various institutions (Alam 2015; Alauddin & Sarker 2014) and their social capital (i.e., social connection) play crucial roles in their adaptation decisions (Wolf et al. 2010; Deressa et al. 2009). These issues have particular importance for the resource-poor rural riparian communities where the availability of institutional services and social connection among farmers seems to be limited due to the fragile infrastructure and low livelihood status. Action like government intervention is crucial in ensuring sustainability of farm-level autonomous adaptations (Stringer et al. 2009; Maddison 2007; Smit & Pilifosova 2001).

This research using cross-sectional survey data provides new insights on the determinants of the households' choice of adaptation and the barriers to their adaptation. The research questions posed to investigate this are: (i) what are the main adaptation strategies that the resource-poor households adopt?; (ii) what are the

barriers to adaptation?; and (iii) what are the determinants influencing adaptation strategies, especially the influence of institutional access and social capital of resource-poor households in the study areas?

8.3 Review of literature

This section provides a summary of the existing research on climate change adaptation and the factors influencing adaptation. Scholars mentioned that adaptation is one of the key policy options that determine the severity of the impact of climate change on agriculture (Green & Raygorodetsky 2010; Kurukulasuriya & Mendelsohn 2008; Lobell et al. 2008; Brooks et al. 2005; Adger et al. 2003). Understanding the determinants of adaptive capacity is crucial to explaining the local autonomous adaptation process. This knowledge assists policy development by strengthening adaptation through investing in these factors (Yohe & Tol 2002).

Empirical evidence from outside Bangladesh indicates that the most common adaptation strategies are using new crop varieties, diversifying crop varieties, adopting mixed crop and livestock farming systems, changing planting dates, planting trees, irrigation, soil conservation, and switching from farm to non-farm activities (Gebrehiwot & van der Veen 2013; Deressa et al. 2011; Deressa et al. 2009; Molua 2009; Kurukulasuriya & Mendelsohn 2008; Nhemachena & Hassan 2007). The ability and capacity to adapt are influenced by system characteristics (e.g., agro-ecological) that are called the 'determinants of adaptation' (Smit et al. 2000). The determinants of adaptation choices can be broadly categorized as:

- Household and farm characteristics, including household head's age, gender, education, farming experience, household income, farm size and tenure status (Gebrehiwot & van der Veen 2013; Bryan et al. 2013; Deressa et al. 2009; Hassan & Nhemachena 2008).
- Social capital encompassing farmer-to-farmer extension and organizational involvement (Deressa et al. 2009).
- Institutional variables comprising access to climate information, extension services, credit facilities, markets, irrigation, and off-farm employment opportunities (Gebrehiwot & van der Veen 2013; Bryan et al. 2009; Deressa et al. 2009; Kurukulasuriya & Mendelsohn 2008).

Although the impact of climate change and hazards in Bangladesh is not limited to the occurrence of droughts, most of the adaptation strategies are drought focused (see previous Chapter 7). Previous studies have been identified various determinants of adaptation strategies using a multinomial logit model (MNL). For example, Alam (2015) indicated that farmers with more experience of farming, better schooling, and access to electricity and institutional facilities would have an increased likelihood of adopting alternative adaptation strategies in the droughtprone Rajshahi district. Alauddin and Sarker (2014) showed a household head's education level, farm size, access to climate information, electricity for irrigation, agricultural subsidies and severity of drought were significant factors underpinning the farmers' decision to adopt adaptation strategies in drought-prone areas in Bangladesh. Sarker et al. (2013) found that the household head's gender, age, education, household income, farm size, farmer-to-farmer extension, and access to credit, subsidy and electricity were the main determinants of an adaptation strategy in the Rajshahi district.

Empirical results suggest that riverbank erosion has catastrophic impacts on the lives and livelihood of riverine households in Bangladesh (Penning-Rowsell et al. 2013; Lein 2010; Hutton & Haque 2004, 2003; Haque 1997). So far there is no indepth empirical research on the factors influencing the local adaptation of hazardprone resource-poor rural households and the barriers to adaptation. Place-based climate adaptation studies have received much theoretical discussion in recent years (Groulx et al. 2014; Fresqe-Baxter & Armitage 2012). Eisenack (2009) argued that local autonomous adaptation is not sufficient to reduce the risk of climate change. The factors that contribute to the adaptive capacity of households could allow government intervention to target the right groups of people and to formulate and implement an effective and sustainable adaptation policy in the country.

8.4 Data analysis method

As procedure of data collection and empirical model are discussed detailed in Chapter 3. Here, only the specifications of the variables are discussed.

8.4.1 Specification of variables

The selection of explanatory variables in this study is based on the review of the literature, the focus group discussions and field experience. The author has assumed household adaptation strategies are a function of a household's socio-economic and farm characteristics such as the age, gender and education of the household head, household income and farm size, access to climate information and other institutions, and social capital.

Some authors have argued that social capital and access to various institutions have crucial roles in enabling households to adjust their management practices (Wood et al. 2014; Deressa et al. 2009; Hassan & Nhemachena 2008; Smit & Wandel 2006). Jordan (2015) argued that social capital can increase a household's resilience and can be used for more forward-looking adaptations. Therefore, indices of social capital and access to various institutional facilities were constructed. The components of the institutional access index are: (i) access to market (input and output), (ii) financial institution for credit, (iii) agricultural extension services, (iv) information on climate and weather conditions, and (v) off-farm employment opportunities. The social capital index includes farmer-to-farmer extension, organisational involvement of the household heads and women members. The respondents replied 'yes' or 'no' to the questions on these components and the score as provided to make the index²⁴. The higher the index value the higher the likelihood of the adoption of that particular adaptation strategy.

The specific model for the determinants of adaptation strategies stands as follows:

 $Y_i(DCV, HG, TP, DIS and Mi) = \beta_0 + \beta_1(HHAg)_i + \beta_2(HHG)_i + \beta_3(HI)_i + \beta_4(HHEd)_i + \beta_5(LF)_i + \beta_6(MF)_i + \beta_7(SF)_i + \beta_8(LL)_i + \beta_9(IAc)_i + \beta_{10}(SCa)_i + \varepsilon_i$

Where, $Y_i(DCV, HG, TP, DIS \text{ and } Mg)$ = Probability of the ith household to adopt the adaptation strategies, β_0 = Constant, β_{1-10} = Parameters to be estimated, \mathcal{E}_i = Error term, DCV = Diversifying crops and varieties, HG = Homestead gardening, TP = Tree plantation, DIS = Diversifying income sources, Mg = Migration, HHAg = Household head age, HHG = Household head gender, HI = Household income,

²⁴ No weighting was used to treat the facilities equally. Weighting can be inherently biased (Wheeler et al. 2013; Hoffmann et al. 2009).

HHEd = Household head education, LF = Large farm, MF = Medium farm, SF = Small farm, LL = Land less, IAc = Access to Institutions, SCa = Social capital. The variables and summary statistics are presented in Table 8.1.

Explanatory variables	Description	Mean	Std.
Age	Years (continuous)	45.12	14.43
Education	Years of schooling	3.17	4.63
	(continuous)		
Gender	Dummy, $1 = male$, $0 = female$	0.95	0.22
Average household income	Bangladeshi Taka	35000	38456
	(continuous)		
Large farmer $(N = 47)$	Dummy, $1 = large farmer, 0 =$	0.23	0.32
	otherwise)		
Medium farmer ($N = 119$)	Dummy, 1 = medium farmer,	0.44	0.33
	0 = otherwise)		
Small farmer ($N = 131$)	Dummy, $1 = $ small farmer, 0	0.63	0.46
	= otherwise)		
Landless ($N = 83$)	Dummy, $1 = $ small farmer, 0	0.68	0.48
	= otherwise)		
Institutional access index	Continuous	1.36	0.89
Social capital index	Continuous	0.67	0.45

 Table 8. 1: Summary statistics.

8.4.2 Model diagnosis

The problems of multicollinearity, heteroskedasticity and the effect of outliers in the variables are usually associated with cross-sectional survey data. This study examined collinearity using the correlation matrix with all the explanatory variables. The correlations are found to be relatively low (<0.39) in all cases which is less than the typical range of 0.7 (Kennedy 1998). Thus, correlation problems between explanatory variables can be ruled out. The effects of a possible collinearity between variables were not large. In order to explore the potential multicollinearity in the model, the VIF for each of the explanatory variables were calculated which range from 1.07 to 1.53. This does not reach to the conventional thresholds of 10 or higher used in regression diagnosis (Gujarati 2003). The robust standard errors were used to tackle the problem of heteroskedasticity. The Ramsey-RESET test was also performed to test the accuracy of the models. The result rejected the null hypothesis of incorrect functional form that indicates relevant variables have not been omitted.

Endogeneity can also be a problem as its presence in the model creates bias estimates and limits the ability to make inferences about the characteristics (Wooldridge 2006). However, this issue has so far received relatively little attention in climate change adaptation studies (notable exceptions are Alam 2015; Wheeler et al. 2013; Di Falco et al. 2012). The education variable in the model could be argued to be a potential endogenous variable due to the influences of some external confounding factors, namely the Compulsory Primary Education Policy of the government of Bangladesh (Alam 2015). The endogeneity problem of the education variable in the model is examined by employing an augmented Durbin–Wu–Hausman test. Using the total educated numbers in the family as a proxy for the government policy intervention, the test result rejects the null hypothesis that the education variable is endogenous (F value 1, 1.05; Prob >0.2).

8.5 Results and discussions

8.5.1 Households' main adaptation strategies

The households were found to adopt different strategies based on their long-term knowledge, experience and perceptions in the face of the riverbank erosion hazard and other climate change issues. All of the households responded positively to undertaking adaptation measures to address these adverse effects. Households adopted at least one form of adaptation from the various adaptation options to sustain their farming and livelihood. An initial 15 adaptation strategies were identified through the focus group discussions. However, these failed to generate statistically significant parameters in the logit estimation. Therefore, following Alam (2015), Alauddin and Sarker (2014), Gebrehiwot and van der Ven (2013), and Sarker et al. (2013), the adaptation strategies were reorganized by grouping closely related choices into the same category based on the best practices in the field and expert opinions for the model estimation. Thus, diversifying crops and varieties included the cultivation of pulses, spices and oil seed, and the cultivation of wheat and HYV rice varieties (e.g., BRRI-28, BRRI-29). Adjusting planting time and techniques included the cultivation of Aman and Aus varieties of rice, as well as vegetables. Diversifying income sources included livestock, poultry and duck rearing, small business and offfarm employment. Small and landless farmers were found to adopt seasonal migration, especially during the rainy seasons when there was limited scope of both

farming and non-farming employment to improve their livelihood and food security. Tree plantation was practiced mainly by large and medium farmers who had sufficient land. The adaptation strategies of the households resulted in six main outcomes which are unordered and discrete (Figure 8.1).



Figure 8. 1: Main adaptation strategies of households

8.5.2 Barriers to adaptation

Although the households were adopting adaptation strategies, they reported some barriers that prevented them from adapting successfully. The main barriers were the lack of information about riverbank erosion and related climate issues, one's own land for cultivation, appropriate crop varieties, knowledge of appropriate adaptation and credit facilities (Table 8.2). Also mentioned were other post-production related problems such as a lack of storage facilities, marketing and transportation facilities which are crucial for policy intervention.

However, the barriers were felt heterogeneously among the farming groups. For example, the main barriers to adaptation for households with relatively less land ownership were the lack of credit, own land and knowledge about appropriate adaptation: the lower average land size among these households was highly significant (p<0.007) compared to the households who did not mention these as a main barrier (independent sample t-test). The lack of storage and marketing facilities were mentioned mainly by the large and medium farmers as these might prevent

them from getting the best price for their products (Table 8.2). Connecting the small farmers to supermarkets could be a strategic option for both government and NGOs who are working to improve the livelihoods by enabling them better access to market.

Barriers to adaptation	Response by farm category					
Darriers to adaptation	Large	Medium	Small	Landless		
Lack of information about						
riverbank erosion and related	XX	XX	XX	XX		
climatic issues						
Lack of appropriate variety of crops	XX	XX	XX	—		
Lack of knowledge concerning	x	x	xx	xx		
appropriate adaptation strategies	<u> </u>	А				
Lack of credit/money/saving	_	Х	XX	XX		
Lack of suitable land for cultivation	_	_	XX	XX		
Lack of own land	_	_	XX	XX		
Lack of storage facilities	XX	XX	_	_		
Lack of marketing facilities	XX	XX	XX	—		
Lack of transportation facilities	Х	Х	Х	—		

Table 8. 2: Perceived barriers to adaptation measures.

Where, xx = main barriers, x = barriers

They also mentioned a lack of knowledge about appropriate adaptation strategies and transport facilities as barriers. A lack of credit is appeared to be the main barrier for small and medium farmers. A lack of institutional access and credit can limit their ability to get the resources and technologies they might need for adaptation. Since the small and landless farmers have resource limitations, access to financial institutions is crucial for them to undertake adaptation.

8.5.3 Econometric results

Table 8.3 presents the results of the MNL model of estimated parameters and marginal effects.

Overall, the model offers a good fit with factors predicting the adoption of adaptation strategies by the study households. The chi-square statistics (LR–213.43) indicate the strong explanatory power of the model. In other words, the joint null hypothesis that all variables are jointly significant is accepted. Goodness of fit of the model given by the McFadden pseudo R^2 of 0.29 also indicates reasonable explanatory power of the model (Table 8.3). We also tested the IIA by employing the Hausman test. The test results failed to reject the null hypothesis of IIA at the 5% level (p value of 0.231). This indicates that the estimated model meets the asymptotic assumption of the test. Moreover, most of the explanatory variables in the model and their marginal values were found to be statistically significant with an expected sign (see discussion below).

Level of education

It is expected that household heads with more education are more likely to adopt better adaptation strategies. The study found a significant positive relationship on the adoption of diversifying crops and varieties (0.112, p<0.05), homestead gardening (0.019, p<0.10), tree plantation (0.123, p<0.05) and diversifying income sources (0.034, p<0.10). It implies that a one unit (year) increase in a respondent's level of education will increase the probability of adopting diversifying crops and varieties by 0.112 relative to the base category while the effect on the remaining options is negligible. The same interpretation holds true for the other variables. This finding supports the empirical evidence that farmers with higher educational levels were likely to adapt better to climate change in the African context (Gebrehiwot & van der Veen 2013; Deressa et al. 2009) and in Bangladesh (Alam 2015; Alauddin & Sarker 2014).

Age of household head

The age of the household head acts as a proxy for experience and so influences the adoption of adaptation strategies. The study found the household head's age was a significant positive factor on adopting diversifying crops and varieties (0.012, p<0.10) and negative factor in adopting a migration decision (-0.105, p<0.05). It may be due to the fact that experienced people have good knowledge about weather and climate variability and thus adapt to this risk-aversion strategy. Households with low income and resources tend to migrate for few months to improve their livelihood and

food security. However, temporary migration is less likely for an aged household head (negative impact) as it represents their vulnerability. This finding is consistent with previous adaptation studies (Alam 2015; Hisali et al. 2011; Deressa et al. 2009).

Gender of household head

This study found a significant relationship between adopting the strategies of diversifying crops and varieties (0.002, p<0.05) and a migration decision (-0.021, p<0.05) for male-headed households. This result is in accordance with the field experience. But the opinion that male-headed households in an African context are more likely to take up climate adaptation strategies is mixed: Deressa et al. (2009) opposed the findings of Nhemachena and Hassan (2007).

Household income

Household income was a significant positive factor in adopting the strategies of diversifying crops and varieties (0.101, p<0.05) and tree plantation (0.007, p<0.10) and a negative factor in adopting a migration decision (-0.103, p<0.001). Modern agriculture is capital intensive: more capital is required when adopting new crops and varieties, agricultural technologies and fertiliser management. This opportunity is somewhat limited for small and marginal farmers unless they get access to credit. Previous studies found a positive relationship between income and adaptation also (Alam 2015; Alauddin & Sarker 2014; Gebrehiwot & van der Veen 2013).

	Adaptation strategies (Dependent variable)									
Explanatory	Diversifying crops and varieties		Homestead gardening		Tree plantation		Diversifying income sources		Migration	
variables	Coefficient	Marginal effect	Coefficient	Marginal effect	Coefficient	Marginal effect	Coefficient	Marginal effect	Coefficient	Marginal effect
Constant	-5.31**		-3.41*		-1.75*		-1.23 **		-2.65 *	
Constant	(2.441)		(2.201)		(0.905)		(0.571)		(1.361)	
A go	0.125**	0.012*	0.141	0.025	0.130*	0.019	0.102*	0.037	-0.321***	-0.105**
Age	(0.051)	(0.013)	(0.112)	(0.017)	(0.077)	(0.031)	(0.052)	(0.025)	(0.121)	(0.047)
Education	0.313**	0.112**	0.065*	0.019*	0.071**	0.123**	0.093**	0.034*	0.071	0.006
Education	(0.124)	(0.053)	(0.037)	(0.011)	(0.033)	(0.061)	(0.043)	(0.018)	(0.032)	(0.012)
Condor	0.011**	0.002**	0.017	0.009	0.061	0.015	0.023	0.009	-0.131***	-0.021**
Gender	(0.004)	(0.001)	(0.014)	(0.021)	(0.047)	(0.012)	(0.013)	(0.011)	(0.041)	(0.01)
Average household	0.135**	0.101**	0.023	0.001	0.013*	0.007*	0.013	0.002	-0.211***	-0.103***
income	(0.061)	(0.047)	(0.021)	(0.000)	(0.007)	(0.004)	(0.006)	(0.000)	(0.056)	(0.031)
L C	1.128***	0.231***	0.017	0.005	0.193**	0.074**	0.011	0.000	-0.171***	-0.103***
Large farmers	(0.331)	(0.083)	(0.102)	(0.014)	(0.065)	(0.026)	(0.104)	(0.000)	(0.051)	(0.035)
Madium formara	0.122***	0.101***	0.023	0.007	0.103**	0.045**	0.027	0.003	-0.112***	-0.073**
Medium farmers	(0.039)	(0.029)	(0.142)	(0.105)	(0.035)	(0.022)	(0.204)	(0.093)	(0.036)	(0.026)
Small formara	0.118	0.072	0.191***	0.108**	0.076	0.012	0.213***	0.112***	0.172***	0.094**
Small farmers	(0.103)	(0.041)	(0.061)	(0.045)	(0.045)	(0.014)	(0.067)	(0.036)	(0.054)	(0.035)
Landlass formars	0.105	0.051	0.115**	0.073**	0.114	0.065	0.059**	0.023**	0.237***	0.113***
	(0.076)	(0.031)	(0.041)	(0.025)	(0.102)	(0.073)	(0.021)	(0.011)	(0.067)	(0.037)
Institutional access	0.511***	0.191***	0.130**	0.071**	0.028**	0.011**	0.106**	0.013**	0.014	0.005
index	(0.183)	(0.072)	(0.064)	(0.034)	(0.014)	(0.005)	(0.045)	(0.006)	(0.045)	(0.012)
Social capital	0.215***	0.102***	0.251**	0.127**	0.151	0.016	0.113**	0.031*	0.153***	0.119***
index	(0.073)	(0.04)	(0.097)	(0.055)	(0.312)	(0.145)	(0.051)	(0.017)	(0.053)	(0.041)
Log likelihood	-227.12									
Pseudo R ²	0.29									
I R (Chi square)	213.43									
Lik (Cill-square)	(p<0.02)									

Table 8. 3: Estimated results from MNL model.

N= 380. ***p<0.001; **p<0.05 and *p<0.10. Adjusting planting time and techniques is used as base category. Robust standard errors are indicated in parentheses.

Farm status

Land ownership plays a key role in the livelihood of most of the rural households and this was expected to be a factor in increasing adaptation in farming. Farmers of large and medium land holdings are relatively well resourced and more likely to adopt strategies earlier than farmers with small plots, and landless farmers. This study found a significant positive relationship in adopting diversifying crops and varieties (0.231, p<0.001 and 0.101, p<0.001) and tree plantation (0.074, p<0.05 and 0.045, p<0.05), and a significant negative relationship in the case of a migration decision (-0.103, p<0.001 and -0.073, p<0.05) for farmers of large and medium land holdings, respectively. It is understandable that households with sufficient land are not likely to migrate. By contrast, farmers of small plots and landless farmers migrate seasonally frequently (0.094, p>0.001 and 0.113, p>0.001 for small and landless farmers, respectively). They cannot generate enough income to sustain their livelihood mainly due to the lack of employment opportunities in farming. They are more likely to adopt homestead gardening (0.108, p>0.05 and 0.073, p>0.05 for small and landless farmers, respectively) for the effective and sustainable use of their limited land resources. This strategy provides nutrients in their food chains and is an important source of subsequent income throughout the year. The significant positive relationship between farm size and adaptation are consistent with previous studies (Alauddin & Sarker 2014; Sarker et al. 2013; Deressa et al. 2009).

Institutional access

This study found evidence that suggests a household's access to institutional facilities greatly influences the likelihood of adopting adaptation strategy. The marginal results of the probability of adopting adaptation strategies such as diversifying crops and varieties (0.191), homestead gardening (0.071), tree plantation (0.011) and diversifying income sources (0.013) were found significant at the 5% level. The availability of information can promote adaptation through better management of crops, land, fertilizer and climate variability. Access to credit has been reported to have a significant positive impact on adaptation decisions (Alauddin & Sarker 2014; Bryan et al. 2009; Deressa et al. 2009). Gebrehiwot and van der Veen (2013) mentioned that access to markets can serve as a platform for providing information for farmers. Information on climate change can create awareness among farmers and increase the probability of adopting adaptation strategies (Alam 2015;

Deressa et al. 2009; Maddison 2007). Field experience suggests that farmers with small plots and landless farmers have limited access to institutional facilities, especially in terms of access to credit and extension services, which limits their scope to adopt adaptation strategies. Access to institutional facilities was also mentioned as a main barrier to adaptation by the small and landless farmers (section 8.5.2). Strong government intervention is required to ensure these households' access to institutional facilities.

Social capital

The study results show a highly significant role of social capital on the likelihood of adaptation strategy adoption. Social capital increases the probability of implementing the strategy of diversifying crops and varieties (0.102, p<0.001), especially for large and medium farmers. Small and landless farmers benefit through adopting the strategies of migration (0.119, p<0.001), homestead gardening (0.127, p<0.05) and diversifying income sources (0.031, p<0.10). This result is consistent with the findings that the presence of a strong kinship network can increase the adaptive capacity of farmers by providing economic, managerial and psychological help (Smit & Wandel 2006). Deressa et al. (2009) found a highly significant negative relationship between social capital and no adaptation decision. Households have reported that access to farmer-to-farmer extension and government extension services stimulated them to cultivate in the new 'char land' which was fallow in the past. Households which adopted homestead gardening and changing profession towards livestock, poultry and duck rearing reported a positive contribution for adopting such strategies through their involvement in different organizations and NGOs. However, informal social networks typically include women's groups, religious groups and cooperative farming groups which is the key to form social capital is found limited existence in the study areas. This is mainly due to the fragile environment and low livelihood status of the households. Small and landless farmers expressed an opinion that sharing and exchanging information and views with each other helped them to take the seasonal migration decision to improve their livelihood and food security.

8.6 Summary of the chapter

This chapter has highlighted the factors influencing local adaptation strategies and the barriers to adoption by hazard-prone resource-poor households. The MNL model passes the assumptions of the IIA and does not suffer from multicollinearity, heteroskedastacity and endogeneity problems as confirmed by the statistical tests. The study reveals that all of the sample households have responded at least somewhat to the hazards and other climate change issues through adopting a range of adaptation strategies depending on their socio-economic and household characteristics, and access to institutional facilities and social capital. Migration appears to be an important adaptation strategies are diversifying crops and varieties, diversifying income sources, adjusting plantation time and techniques, planting trees and homestead gardening. The important barriers to adopting the adaptation strategies include a lack of information about riverbank erosion and related climatic issues, a lack of knowledge about appropriate strategies, unsuitable crop varieties, the limitations of one's own land and limited access to credit.

Analyses of marginal effects indicate that household characteristics such as household heads' level of education and age, farm status and household income have significant impacts on which adaptation strategies are decided upon. Thus, investment in education and a supply of high yielding crops and varieties suitable to local conditions can be effective options for reducing the adverse impacts of climate change and hazards, and be means to improve their livelihoods. The study also reveals that access to institutional facilities and social capital are the key factors influencing the adoption of adaptation strategies by the households. This underscores the need for strengthening the extension services in the study areas and providing rural households with better information on production techniques, agronomic and land management practices, and climate change issues. Access to financial institutions and the creation of off-farm employment opportunities in riverine rural areas are also crucial to support the households in adapting to climate change at the farm level. Government organisations and NGOs can play a greater role by helping to form social organizations/clubs with the farmers (e.g., an Integrated Pest Management club) or assisting cooperative farms in these poorly resourced

151

communities so that the adoption of adaptation strategies is likely to contribute to their successful continuation.

Adaptation strategies and intervention policies which are centralised in nature in Bangladesh need to consider local circumstances when developing new crop varieties, high-value crops and technology suitable for the emerging *char* land in order to accelerate the effective and logical autonomous adoption of adaptation processes. This will enhance the resilience of vulnerable households in riparian areas across Bangladesh. The next chapter summarises the findings of the study.

CHAPTER NINE

Conclusions and Recommendations

9.1 Chapter outline

This chapter summarises the whole thesis. After describing the brief background of the research in Section 9.2, the chapter is organised as follows: the main findings of the research as per set objectives are presented in Section 9.3. Relevant policy recommendations based on the findings are presented in Section 9.4. The contribution of this research to the literature is enlightened in Section 9.5. The acknowledgement of the limitations of this research is presented in Section 9.6. Further research directions are provided in Section 9.7 and the Section 9.8 contains the conclusions of the study.

9.2 Introduction

Bangladesh is most vulnerable to climate change due to its low-lying deltaic topography dominated by major rivers, population density, limited land area and poverty. It is situated in the interface of two environments – the Bay of Bengal to the south and the Himalayas to the north. This geographical position makes the country more exposed to the impacts of frequent events of extreme climatic hazards including riverbank erosion. A large part of Bangladesh are subject to recurrent riverbank erosion. This causes significant loss of land and displaces thousands of people annually, and pushes them into vulnerable conditions of food insecurity and poverty. Forecasted climate change impacts may also influence the frequency of flooding which accelerate the erosion.

The most vulnerable communities are those marginalised rural groups who seek to make a living along riverbanks and on sand bars in the shifting channels and sediment-laden tributaries where land is available, albeit, temporarily. People living in these areas have relatively limited capacity to cope with climate-induced shocks and consequently natural disasters are likely to have persistent effects on their lives, livelihoods, health and welfare. The crucial policy agenda of Bangladesh is to identify and to understand vulnerability and possible adaptation strategies, particularly for marginalised riparian communities which could mitigate the effects of an adverse climate change and hazards and improve food security and livelihood. Until now, no study had systematically addressed this issues. Therefore, this study has assessed the livelihood vulnerability of riverbank erosion and its impact on riparian rural households' food security and their coping and adaptation strategies.

The study uses both primary (cross-sectional survey) and secondary data. For the field survey, the Chauhali upazila of the Sirajgonj district and the Nagarpur Upazila of the Tangail district were selected. Survey data were collected randomly from 380 households using a structured survey questionnaire with face-to-face interviews. Moreover, focus group discussion was also conducted to obtain views on various climatic and socio-economic variables, and these opinions were then used to cross-validate the information obtained from the surveys and the key informant interviews.

9.3 Summary of findings

The study employs a range of standard tools and techniques to analysis the data which are expected to yield sound results. This section briefly narrated the major findings under the set objectives of this study.

9.1.1 Research objective 1

To assess the livelihood vulnerability of the riparian households.

To fulfill the objective, the following research questions are set: (i) what are the main drivers of livelihood vulnerability of riparian households to climatic change and hazards?; (ii) are households isolated from the riverine mainland more vulnerable to climate change than other riparian households?; (iii) does livelihood status serve as a driver of vulnerability?; and (iv) what are the factors influencing their resilience capacity?

Building on the IPCC framework, this study adopts a holistic approach to assessing the livelihood vulnerability of rural riparian households from *char* and riverine mainland communities in Bangladesh. Both communities are affected by the riverbank erosion and other climatic hazards; however, they have different locational identity with respect to the river and therefore, suffer differently in terms of livelihood vulnerability and have different response strategies. Two key vulnerability assessment approaches - the Livelihood Vulnerability Index (LVI) and Climate Vulnerability Index (CVI) - are customized to incorporate local and indigenous knowledge into the selection of sub-components and indicators. The LVI and CVI values are found to be different between *char* and riverbank communities. The *char* dwellers are the most vulnerable and have less adaptive capacity than riverbank mainland households who have a greater diversity in income sources, comparatively less dependence on agriculture, low dependency ratio and a higher level of education. The main drivers of vulnerability are found to be livelihood strategies and access to food, water and health facilities. However, riparian households were also found to be vulnerable due to their relative inaccessibility and low livelihood status. These coupled with climate impacts on river morphology driving erosion and loss of land with consequent decrease in economic potential, creates a vicious cycle of poverty. The vulnerability theory supports the notion that climate change vulnerability does not exist in isolation from wider socio-economic and bio-physical attributes of the communities.

Resilience theory is also applied in this study which is the function of adaptation capacity and sensitivity components of climate change. For this, an indicator based Resilience Capacity Index (RCI) is developed for better understanding of riparian households' resilience activities from socio-economic perspective which is resulting from their long-term knowledge, experience and practices. This will enable policy makers to ensure more targeted and appropriate climate adaptation policies to mitigate the effects of an adverse climate and hazards in the country. Results reveal that they have adopted a range of resilient activities such as homestead gardening, tree plantation, new cropping practices, allowing women to work outside, using safe drinking water and sanitary toilet which might be regarded as a positive move to enhance their resilience capacity. However, the RCI values in both the locations are low that infer households' inability to keep peace or cope with, and adapt to the increasing impacts of climate change and hazards. Their lower level of education, social networks and access to food, water and health services are the important limiting factors for their resilience capacity.

9.1.2 Research objective 2

To assess the food security status of the respondents.

The following research questions are sought to fulfill the objective: (i) what is the livelihoods status of the riverine households?; (ii) what are the factors influencing households food insecurity?; (iii) which months the households experience more food shortage (the extent of food insecurity)?; (iv) what are their coping strategies to address the food shortage? and (v) what are the policy options to improve food security of these hazards-prone rural households in a sustainable way in Bangladesh?

The determinants of household food security are basically based on the theories of consumer demand and production that is widely known as the Agricultural Household Models. Both calorie intake method and Household Food Insecurity Access Scale are used to determine the household food security. The study indicates that riverine households' lack of access to many basic necessities and services such as food, safe drinking water, education and health results in increased vulnerability to food insecurity. More than half (56%) of the households fall into the food insecure category, with an average per capita calorie consumption of 1,867 kcal/day, which is about 12% less than the standard minimum daily requirement. The standard deviation of the calorie demand variable is fairly high, which indicates a wide range of variability across sample. Study also reveals that about 73% of the households' total expenditure is on food items and less than 2% on health care. The total market purchase value of food consumed at home stands at 75%: this indicates the vulnerability of the households to price shocks. Employing logit model after addressing data related problems including endogeneity, the study has found several factors served as a driver of households' food insecurity such as household heads' education, household size, adoption of livestock and access to non-farm earnings. Study also found new evidence which suggests that physical health status of the household heads is a key significant factor influencing household food security.

To understand the household food security status throughout the year a household Food Security Index (FSI) is developed. The FSI value ranges between 3 and 1, where 3 indicates food secure and 1 least food secure. The overall FSI was estimated to be 2.06, indicating households' hardship to manage family food requirements throughout the year to be regarded as 'food secure'. These households

can usually manage food twice per day for their family members. Households, on an average, experience food insecurity for the six months from Ashar to Agrahyon (mid-June to mid-November). These months mainly cover the rainy seasons in Bangladesh when opportunities for both farming and non-farming activities are reduced significantly. The households experience most food insecurity situation during the month of Bhadra (1.83) to Ashwin (1.91) (August to October). The situation starts improving little bit from the month of November when the cultivation of Boro rice started i.e. scope of employment in the farming started. The households were able to supply food only for a few months to feed family members from their own production: on an average 3.2 months. To reduce the food shortage they adopt some sort of coping strategies. Therefore, a Coping Strategies Index (CSI) was developed to rank the household coping strategies. The most common coping strategies are to rely upon less expensive or less preferred food, reduce number of meals per day and purchase food on credit.

9.1.3 Research objective 3

To identify the response strategies of the vulnerable households in the face of riverbank erosion and other climate change issues.

The findings of this objective are described in two parts. The first part is concerned with the evidence of local knowledge of adaptation in response to perceived changes in the climate and the impact of riverbank erosion and other climate change issues. The second part analyse the factors that influence household adaptation choices using Multinomial Logit (MNL) model and the barriers to adaptation. Despite the recognition of the need for policies and programs to implement and facilitate adaptation strategies, there is a lack of information about local measures that can reduce households' vulnerability. The research questions set to understand the first part are: (i) what are the perceptions of hazard-prone rural households to climate change and hazards?; (ii) what are the perceived impacts of riverbank erosion and other climate change issues on the livelihoods of the households; and (iii) what local adaptation strategies can the resource-poor households adopt to enhance their resilience?

157

Using descriptive and non-parametric statistical techniques the results indicate that all the households perceived changes in the climate and extreme events, particularly abnormal rainfall in terms of timing and distribution which has serious consequences on their production plans. Climate data also supported households' perception of climate change and climatic variability. The households perceived the various impacts of erosion hazard and other climate change issues on livelihood capitals, including loss of land, crops and yield, homestead, pond areas, trees, access to education and health facilities, and infrastructure resulting in an increase of livelihood vulnerability. Agricultural production depends mainly on physical and natural capital, which experienced varying degrees of impact. Households' human capital in terms of education and skills, social capital in terms of access to health facilities, social bonding and organisational involvement, and financial capital in terms of access to NGOs and government financial institutions are found to be affected. They have to travel a longer distance to reach the school and health centre. Altogether might limit their coping ability and push them further into vulnerable conditions.

Households are found responding through adopting farming and non-farming adaptation strategies to build resilience based on their local knowledge, climate change perception and farming status. Significant differences are observed among farm household groups when choosing adaptation strategies: small and landless farm households adopt non-farming adaptation practices mostly. The important strategies include adopting new crop varieties, changing plantation time, homestead gardening, tree plantation, migration and changing to livestock, poultry and duck rearing.

The second part address the following research question: (iv) what are the barriers to adaptation?; and (v)what are the determinants influencing adaptation strategies, especially the influence of institutional access and social capital of the resource-poor households? The results of the second part indicates that all of the sample households have responded at least somewhat to the hazards and other climate change issues through adopting a range of adaptation strategies depending on their socio-economic and household characteristics, and access to institutional facilities and social capital. The important barriers to adaptation strategies include a lack of information about riverbank erosion and related climatic issues, a lack of knowledge about appropriate strategies, unsuitable crop varieties, the limitations of

158

one's own land and limited access to credit. Analyses of marginal effects indicate that household characteristics such as household heads' level of education and age, farm status and household income have a significant impact on which adaptation strategies are decided upon. The study also reveals that access to institutional facilities and social capital are the key factors influencing the adoption of adaptation strategies by the households.

9.4 Policy implications

The finding of this study has several policy implications. Based on the results, the following specific recommendations are made. Both the government and NGOs should strengthen their activities in different capacities to improving household food security and to reduce vulnerabilities in the areas.

9.4.1 Public sector role

- To ensure access to food, a targeted food policy intervention is yet to be developed for these vulnerable communities. In short-term, direct food transfer through food aid program can boost access to food, since these resource-poor households have limited access to food. The coverage of safety net program in the study areas seems to be inadequate which need to be expanded significantly.
- Development of improved communication, transportation and access to markets and services also vitally important to support existing and alternative livelihoods for individuals and households.
- Interventions need to be targeted to promote community's capacity development in the area of human capital, social capital and organisational capacity that are likely to contribute to enhance resilience of the disadvantaged communities.
- Targeted programs are sought to boosting primary school enrolment and other skills development program in the areas. Many educational institutions have been found eroded coupled with fragile road communication which limit their access to education.
- More investment in agricultural research and development is required. Scientists need to continue to develop crop varieties, high-value crops and technologies

suitable to local conditions, especially in the emerging *char* lands, to accelerate the adaptation process.

- Existence of government financial institutions are limited in the study areas. Therefore, poor farmers' access to credit should be ensured. This will enable them to obtain the resources and technologies they might need for adaptation.
- Adaptation strategies and intervention policies which are centralised in nature in Bangladesh need to consider local circumstances when developing new crop varieties, high-value crops and technology in order to accelerate the effective and logical autonomous adoption of adaptation processes. This will enhance the resilience of vulnerable households in riparian areas across Bangladesh.
- In the long run, the construction of river embankments and the protection, and further planting of riparian forests would help to reduce bank erosion. Therefore, government investment is required which often hardly possible for the government due to the magnitude of the cost. The homeless households can be relocated in the *khas* land and providing them with the facilities of house, school, hospital and other basic facilities to build long-term adaptive capacity.

9.4.2 Role of private sector and NGOs

- Activities of NGOs should be strengthen which seem to be inadequate in the study areas. They need to be extended their credit programs significantly so that poor farmers' access to credit is ensured.
- NGOs also need to expand their safety net programs in the areas. NGOs should spread their education program in the remote *char* areas. Both the NGOs and private sector can launch various training programs to improve their skills to be fit for the non-farm sectors.
- They should come forward to disseminate various information including successful adaptation among farmers and stimulates them to adopt with appropriate support such as credit and technical support.

9.4.3 Community involvement

Strengthening capacity of local institutions and communities in the riverine areas is vitally important to cope with the present and future climatic hazards. Access to resources, awareness raising, sharing knowledge and skills, access to information, and local planning process can ensure greater community participation which will ultimately contribute to raise the capacity of autonomous adaptation of the poor farmers.

9.4.4 Public-Private-NGOs partnership

Coordination between the government's various sectors and other stakeholders including NGOs is required in order to make the efforts effective to reduce vulnerabilities. Public-private-NGOs partnerships can play vital role to improve their livelihood in the following areas:

- ♦ Health status of household head critically affects household food security which call attention for important policy implication. Household members with poor health, mainly due to inadequate calorie intake and unavailability of health services, will be incapable to perform farm and non-farm jobs which in turn make them vulnerable and burden to the family and country as well. This demand for policy priority of access to health services in parallel with access to food in order to achieve and sustain long-term food security in Bangladesh. Provision of adequate community health services, which are currently lacking, is one option to ensure households' access to health care: poor households are actually supposed to get free health care from the public hospital. Both the government and NGOs could set up mobile health (m-health) services in the areas along with their microcredit programs. Since most of the households have mobile phone, which enhances the opportunity to provide them with a variety of information related to agriculture and health services. Hospital can be built in the areas through public-private partnership where option for poor people to get access to health service free and/ or stumpy cost should be in place.
- Livestock adoption by the resource poor households emerged as one of the important way forward to address household food insecurity. Therefore, livestock rearing should be encouraged with enabling policy support. For example, government organizations and NGOs could provide households with
livestock support or credit for having livestock, as the poor households suffer from a lack of capital.

- Since the households are deprived of getting right price of their products due to the lack of storage and marketing facilities in the areas. Therefore, making storage facilities available in the areas is one of the important policy tasks where private sectors can play a vital role. They can come forward to establish storage facilities through public private partnership. Connecting small farmers to supermarkets, on the other hand, might ensure right price of their products. Both private sector and NGOs who are working to improve the livelihoods by enabling them better access to market should be considered this option.
- Lack of institutional access and social capital underpins the need for strengthening the extension services in the study areas and providing rural households with better information on production techniques, agronomic and land management practices, and climate change issues. Government organisations and NGOs can play a greater role by helping to form social organisations/clubs with the farmers (e.g., an Integrated Pest Management club) or assisting cooperative farms in these poorly resourced communities so that the adoption of adaptation strategies is likely to contribute to their successful continuation.

9.5 Contribution

The contributions of this study are manifold. These are discussed below:

9.5.1 Contributions to the knowledge

The primary specific contribution of this thesis is that for the first time it generated detailed insights from the vulnerability and resilience analyses of most vulnerable riparian households in Bangladesh. The policy implication of this is that policy makers will be enable to formulate targeted social, economic and environmental policies to overcome the increasing climate change vulnerabilities to improve food security and livelihood of these marginalised communities.

This study has provided a better understanding of the key factors that derive household food insecurity and the pathways in which these factors affect vulnerable groups and households in Bangladesh. This study also explores the new dimension of how household heads' physical health status impacts on rural households' food security and found a significant impact. Policy implication of this is that health care facilities also need policy support in parallel with improved access to food to achieve and sustain long-term food security in Bangladesh. The development of food security index will contribute to understand the most food deficit months the households' experience and thus need-based policy interventions can be formulated.

This study has provided local level knowledge of adaptation which is important for enhancing vulnerable households' resilience in the face of hazards and for coping with climate change and variability. This study also explores the influence of new dimension of social capital and access to institutional facilities in adapting adaptation strategies at the farm level. This information is crucial as a way forward to support and sustain local adaptation process of these vulnerable resource-poor households.

9.5.2 Methodological contribution

The methodological contribution is that this study develops an index for assessing resilience, household food security and coping strategy. This study also modified the livelihood vulnerability index in the context of hazard-prone households. It is expected that these methods can be used in other sectors, regions or rural communities in the world for assessing and comparing vulnerability, resilience and household food security due to the flexibility of the methods. The methods allow to change or replace indicators or sub-components as per the local conditions. Furthermore, the methodologies are free from the limitations of secondary datadriven methods and missing data problems. On the other hand, indicators or subcomponents index values might be very handy in assessing the impact of a policy or a program by substituting the value of indicators which is likely to change and reestimating the overall vulnerability and resilience. Similarly, future vulnerability and resilience under some policy or program interventions could be calculated in order to see whether the planned activities contribute to reduce the vulnerability and enhance the resilience. The conceptual framework developed in this study can be used in future hazard, food security and adaptation research for Bangladesh in particular and other hazards-prone rural communities in general.

9.5.3 Contribution to the theory

The contribution in theory is that this study has adapted the theory of consumer demand and production that is widely known as the Agricultural Household Models, the theory of random utility and the theory of vulnerability to a new and important setting. This study has validated the wider application of these theories in the context of individual household responses to riverbank erosion hazard and other climate change issues and adaptation.

9.6 Limitations of the study

Although, the data of this study is reliable and representative to understand the livelihood vulnerability and food security of the riverine households. However, the results should be judged in light with the *caveat* that it is based on specific areas: relatively closest to the capital of Bangladesh which enables the households to take the opportunities of seasonal migration to the cities. Involving more areas (e.g., Padma and Meghna riverbank erosion-prone areas) would have contributed to a better understanding of their vulnerability, food insecurity and response strategies and thus strong generalization of the results. This is, however, impractical for an individual researcher due to time and funding constraints (Blaikie 2010; Gilbert 2008). Noteworthy, rural households in Bangladesh virtually face analogous socio-economic, environmental and climate conditions (i.e., low educational attainment and income, relatively high birth rate and high dependence on agriculture for livelihoods) which validates the use of a small sample size that can be typical of the whole population.

Another limitation is that one can argue that the indicators used for assessing vulnerability and resilience are not representative enough. This study overcome this issue through extensive review of the literature and multilevel consultation with the experts which expected to yield sound results.

9.7 Future research directions

This research focuses on the most severe riverbank erosion-prone areas in Bangladesh. However, the literature indicates that vulnerability is context specific, and different sectors and communities are impacted unequally due to climate change and hazards. This warrants area and sector specific research to generate a more complete picture of the impacts of climate change and hazards and their response strategies. Therefore, future research can be undertaken in the drought, flood, cyclone and salinity affected areas of Bangladesh. This then may provide an avenue for policy makers to devise area specific adaptation policies which will have potential to address the adverse effects of climate change and hazards more effectively.

Seasonal migration is a common features in the riverine areas. This issue can be investigated more in-depth for understanding the linkages of migration and riverine households' food security.

Household's hazards responses differ as per the location, time and magnitude. Therefore, more research needs to be conducted in other riverbank erosion-prone areas to better understand their resilience activities and provide support to turn them into actions.

Further research can investigate farmers' adjustment in cropping patterns based on environmental and climate variability, and livelihood strategies to better understand of their response strategies. The perceived changes and possible impacts of climate change and hazards to the riverine households need to be monitored scientifically over time.

Successful adaptation strategies of these households can help to better adapt with higher degree of climate change whereas others can be maladaptive. Therefore, the effectiveness of local adaptation strategies of vulnerable riverine households to climate change and hazards needs to be scientifically tested and prioritized.

9.8 Conclusions of the study

This study has comprehensively examined challenging issues facing rural households in Bangladesh in the form of rising water levels, land erosion near the riverbanks, and the subsequent loss of arable land. This in turn has pushed these households into food insecurity and together, these factors are major obstacles to economic and social progress in the nation of Bangladesh. Moreover, this study did field research to seek answers directly from the households themselves about the extent to which this crisis is affecting them, the extent of their food insecurity, and the nature of any adaptation strategies that they may have deployed to offset these challenges.

In bringing together this analysis, the researcher has produced credible findings employing standard tools and techniques for comprehensive understanding of the livelihood vulnerability and food security status of the riverbank erosion hazard-prone rural households in Bangladesh. It has been found that a lack of institutional support including lack of credit, extension services, inadequate marketing facilities, and poor transportation and communication facilities in combination with the effect of land loss, limit their ability to cope with the food insecurity situation and thereby increases the level of their vulnerability. Properly targeted income transfers and credit programs along with infrastructure and human development programs in the riverbank erosion affected areas across the country may have significant positive impact to improve food security and reduce poverty in the long run. What is urgently needed in Bangladesh to circumvent this slow spiral into poverty and hunger is to generate income and employment opportunities for the rural poor people in order to ensure their access to food and other basic needs. Without offsetting hunger and poverty, Bangladesh's capacity for economic and social development will remain stymied.

It is therefore recommended that new policy interventions are needed, focusing on improved access to food, health, water, sanitation and education for these vulnerable communities across Bangladesh that inhabit and make a living from the shifting rivers that comprise the Ganges-Brahmaputra-Jamuna delta, to break the vicious cycle of economic deprivation and poverty. This study further recommends that without addressing household vulnerability, interventions to improve poverty dynamics of such households will be ineffective.

REFERENCES

- Adger, WN & Vincent, K 2005, 'Uncertainty in adaptive capacity', *Geoscience*, vol. 337, no. 4, pp. 399–410.
- Adger, WN 1999, 'Social vulnerability to climate change and extremes in coastal Vietnam,' *World Development*, vol. 27, no. 2, pp. 249–69.
- Adger, WN 2000, 'Social and ecological resilience: Are they related'? *Progress in Human Geography*, vol. 24, no. 3, pp. 347–364.
- Adger, WN 2006, 'Vulnerability', *Global Environmental Change*, vol. 16, no. 3, pp. 268–81.
- Adger, WN, Arnell, NW & Tompkins, EL 2005b, 'Successful adaptation to climate change across scales', *Global Environmental Change*, vol. 15, no. 2, pp. 77– 86.
- Adger, WN, Dessai, S, Goulden, M, Hulme, M, Lorenzoni, I, Nelson, DR, Naess, LO, Wolf, J & Wreford, A 2009, 'Are there social limits to adaptation to climate change'? *Climatic Change*, vol. 93, no. 3, pp. 335–354.
- Adger, WN, Hughes, TP, Folke, C, Carpenter, SR & Rockstron, J 2005a, 'Socialecological resilience to coastal disasters', *Science*, vol. 309, no. 5737, pp. 1036–1039.
- Adger, WN, Huq, S, Brown, K, Conway, D & Hulme, M 2003, 'Adaptation to climate change in the developing world' *Progress in Development Studies*, vol. 3, no. 3, pp. 179–195.
- Agrawala, S, Ota, T, Ahmed, AU, Smith, J & van Aalst, M 2003, Development and Climate Change in Bangladesh: Focus on Coastal Flooding and the Sundarbans, Environment Directorate and Development Cooperation Directorate, Organization for Economic Cooperation and Development, Paris.
- Ahmed, A, Ahmad, K, Chou, V, Hernandez, R, Menon, P, Naeem, F, Naher, F, Quabili, W, Sraboni, E & Yu, B 2012, *The Status of Food Security in the Feed the Future Zone and other Regions of Bangladesh: Results from the* 2011–2012 Bangladesh Integrated Household Survey, International Food Policy Research Institute, Washington, DC.

- Ahmed, AK, Chowdhury, EH 2006, Study on Livelihood Systems Assessment, Vulnerable Groups Profiling and Livelihood and Long-term Climate Change in Drought-prone Areas of NW Bangladesh, Food and Agriculture Organization of the United State, Rome.
- Ahmed, AU 2005, 'Adaptation options for managing water related extreme events under climate change regime: Bangladesh perspectives', In: Mirza, MMQ & Ahmad, QK (eds), *Climate Change and Water Resources in South Asia*, Balkema Press, Leiden, pp. 255–278.
- Ahmed, AU 2006, *Bangladesh Climate Change Impacts and Vulnerability: A Synthesis,* Department of Environment, Government of Bangladesh, Dhaka.
- Ahmed, AU, Quisumbing, AR & Hoddinott, JF 2007, Relative Efficiency of Food and Cash Transfers in Improving Food Security and Livelihoods of the Ultrapoor in Bangladesh, International Food Policy Research Institute, Washington, DC.
- Ahmed, AU, Rashid, S, Sharma, M & Zohir, S 2004, Food Aid Distribution in Bangladesh: Leakage and Operational Performance, International Food Policy Research Institute, Washington, DC.
- Ahmed, I 2015, *People of Many Rivers: Tales from the Riverbanks*, The University Press Limited, Dhaka.
- Ahsan, N & Warner, J 2014, 'The socioeconomic vulnerability index: A pragmatic approach for assessing climate change led risks–A case study in the southwestern coastal Bangladesh,' *International Journal of Disaster Risk Reduction*, vol. 8, pp. 32–49.
- Ahsan, S, Ali, MS, Hoque, MR, Osman, MS, Rahman, M, Babar, MJ, Begum, SA, Rahman, DM & Islam, KR 2011, 'Agricultural and environmental changes in Bangladesh in response to global warming', In: Lal, R, Sivakumar, MVK, Faiz, SMA, Rahman, A & Islam, KR (eds), *Climate Change and Food Security in South Asia*, Springer, Netherlands, pp. 119–34.
- Akhter, R 1984, 'Causes and consequences of migration to the city: A case study of slum dwellers in Dhaka, Bangladesh', Unpublished manuscript, Centre for Urban Studies, Dhaka.

- Akinyele, IO 2009, Ensuring Food and Nutrition Security in Rural Nigeria: An Assessment of the Challenges, Information Needs, and Analytical Capacity, Nigeria Strategy Support Program (NSSP), International Food Policy Research Institute, Washington, DC.
- Akter, S & Basher, SA 2014, 'The impact of food price and income shocks on household food security and economic well-being: Evidence from rural Bangladesh', *Global Environmental Change*, vol. 25, pp. 150–162.
- Akter, S & Mallick, B 2013, 'The poverty-vulnerability-resilience nexus: Evidence from Bangladesh', *Ecological Economics*, vol. 96, pp. 114–124.
- Akter, T 2009, Climate Change and Flow of Environmental Displacement in Bangladesh, Unnayan Onneshan, Dhaka.
- Alam, E & Collins, AE 2010, 'Cyclone disaster vulnerability and response experiences in coastal Bangladesh', *Disasters*, vol. 34, no. 4, pp. 931–54.
- Alam, GMM, Rahman, MS & Mondal, MAS 2004, 'Backward and forward linkages of power tiller technology: Some empirical insights from an area of Bangladesh', *Bangladesh Journal of Political Economy*, vol. 20, no. 1, pp. 139-152.
- Alam, GMM & Khatun, MN 2012, 'Sweetening food security in Bangladesh: Present situation and future strategy', *Pakistan Sugar Journal*, vol. 27, no. 1, pp. 9-17.
- Alam, K 2015, 'Farmers' adaptation to water scarcity in drought-prone environments: A case study of Rajshahi District Bangladesh,' Agricultural Water Management, vol. 148, pp. 196–206.
- Alam, M 2003, 'Bangladesh country case study', Paper presented in the workshop on National Adaptation Program of Action (NAPA), 9-11 September 2003, Bhutan.
- Alam, MM, Siwar, C, Mollah, RI, Talib, B & Toriman, MEB 2012, 'Paddy farmers' adaptation practices to climate vulnerability in Malaysia', *Mitigation and Adaptation Strategies to Global Change*, vol. 17, no. 4, pp. 415–423.
- Alam, QM, Miah, MAM & Mohabbatullah, M 2010, Land Use Patter, Nutritional Status and Food Security of Indigenous People in Hill Areas of Bangladesh,

National Food Policy Capacity Strengthening Program (NFPCSP), Ministry of Food, Dhaka.

- Alauddin, M & Sarker, MAR 2014, 'Climate change and farm-level adaptation decisions and strategies in drought-prone and groundwater-depleted areas of Bangladesh: An empirical investigation', *Ecological Economics*, vol. 106, pp. 204–213.
- Alberini, A, Chiabai, A & Muehlenbachs, L 2006, 'Using expert judgment to assess adaptive capacity to climate change: Evidence from a conjoint choice survey,' *Global Environmental Change*, vol. 16, no. 2, pp. 123–44.
- Alexander, C, Bynum, N, Johnson, E, King U, Mustonen, T & Neofotis, P 2011, 'Linking indigenous and scientific knowledge of climate change', *BioScience*, vol. 61, no. 6, pp. 477–84.
- Ali, AKMS, Jahan, I, Ahmed, A & Rashid, S 2008, 'Public food distribution system in Bangladesh: Successful reforms and remaining challenges', In: Rashid, S, Gulati, A & Cummins, RJ (eds), *From Parastatals to Private Trade: Lesson from Asian Agriculture*, The Johns Hopkins University Press, London.
- Amaza, PS, Umeh, JC, Helsen, J & Adejobi, AO 2006, 'Determinants and measurement of food insecurity in Nigeria: Some empirical policy guide', Paper presented at International Association of Agricultural Economists Annual Meeting, August 12-18, Queensland.
- Anik, AR, Manjunatha, AV, Bauer, S 2013, 'Impact of farm level corruption on the food security of households in Bangladesh', *Food Security*, vol. 5, no. 4, pp. 565–574.
- Anik, SI & Khan, MASA 2012, 'Climate change adaptation through local knowledge in the north eastern region of Bangladesh', *Mitigation and Adaptation Strategies to Global Change*, vol. 17, no. 8, pp. 889–896.
- Anriquez, G, Daidone, S & Mane, E 2013, 'Rising food prices and undernourishment: A cross-country inquiry', *Food Policy*, vol. 38, pp. 190– 202.
- Apata, TG, Samuel, KD & Adeola, AO 2009, 'Analysis of climate change perception and adaptation among arable food crop farmers in South Western Nigeria',

Paper presented at the International Association of Agricultural Economists' 2009 Conference, 16–22 August 2009, Beijing.

- Arbuckle, JG, Morton, LW & Hobbs, J 2013, 'Farmer beliefs and concerns about climate change and attitudes toward adaptation and mitigation among farmers in the Midwestern United States', *Climatic Change*, vol. 117, no. 4, pp. 943– 950.
- Aromolaran, AB 2010, 'Does increase in women's income relative to men's income increase food calorie intake in poor households? Evidence from Nigeria', *Agricultural Economics*, vol. 41, no. 3-4, pp. 239–249.
- Arora-Jonsson, S 2011, 'Virtue and vulnerability: Discourses on women, gender and climate change', *Global Environmental Change*, vol. 21, no. 2, pp. 744–51.
- Aryal, S, Cockfield, G & Maraseni, TN 2014, 'Vulnerability of Himalayan transhumant communities to climate change', *Climatic Change*, vol. 125, no. 2, pp. 193–208.
- Ayers, MJ & Huq, S 2008, 'The value of linking mitigation and adaptation: A case study of Bangladesh', *Environmental Management*, vol. 43, no. 5, pp. 753– 764.
- Bahiigwa, BAG 1999, Household Food Security in Uganda: An Empirical Analysis, Economic Policy Research Centre, Kampala.
- Balagtas, JV, Bhandari, H, Cabrera, ER, Mohanty, S & Hossain, M 2014, 'Did the commodity price spike increase rural poverty? Evidence from a long-run panel in Bangladesh', *Agricultural Economics*, vol. 45, no. 3, pp. 303–312.
- Bardsley, DK & Wiseman, ND 2012, 'Climate change vulnerability and social development for remote indigenous communities of South Australia', *Global Environmental Change*, vol. 22, no. 3, pp. 713–23.
- Barnum, H & Squire, L 1979, 'A model of an agricultural household: Theory and evidence', World Bank occasional papers 27, Washington, DC.
- Barrett, C 2010, 'Measuring food insecurity', Science, vol. 327, no. 5967, pp. 825-828.

- Bartlett, JE, Kotrlik, JW & Higgins, CC 2001, 'Organizational research: Determining appropriate sample size in survey research', *Information Technology Learning and Performance*, vol. 19, no. 1, pp. 43–50.
- Bashir, MK, Naeem, MK & Niazi, SAK 2010, 'Rural and peri-urban food security: A case of district Faisalabad of Pakistan', *World Applied Science Journal*, vol. 9, no. 4, pp. 403–411.
- Bashir, MK, Schilizzi, S & Pandit, R 2012, 'The determinants of rural household food security in the Punjab, Pakistan: An econometric analysis,' Working Paper 1203, School of Agricultural and Resource Economics, University of Western Australia, Australia.
- BBS (Bangladesh Bureau of Statistics) 2014, *Statistical Pocketbook of Bangladesh*, Bangladesh Bureau of Statistics, Dhaka.
- BBS 2010, *Household Income and Expenditure Survey 2010*, Bangladesh Bureau of Statistics, Dhaka.
- BBS 2012, Yearbook of Agricultural Statistics, Bangladesh Bureau of Statistics, Dhaka.
- Below, TB, Mutabazi, KD, Kirschke, D, Franke, C, Sieber, S & Siebert, R 2012, 'Can farmers' adaptation to climate change be explained by socio-economic household-level variables?', *Global Environmental Change*, vol. 22, no. 1, pp. 223–35.
- Berkes, F & Folke, C (eds) 1998, Linking Social and Ecological Systems: Management Practices and Social Mechanisms for Building Resilience, Cambridge University Press, New York.
- Berman, R J, Quinn, CL & Paavola, J 2015, 'Identifying drivers of household coping strategies to multiple climate hazards in Western Uganda: Implication for adaptation to future climate change', *Climate and Development*, vol. 7, no. 1, pp. 71–84.
- Bernell, S, Edwards, M & Weber, B 2006, 'Restricted opportunities, unfortunate personal choices, and ineffective policies? What explain food insecurity in Oregon', *Journal of Agricultural and Resource Economics*, vol. 31, no. 2, pp. 193–211.

- Berry, PM, Rounsevell, MDA, Harrison, PA & Audsley, E 2006, 'Assessing the vulnerability of agricultural land use and species to climate change and the role of policy in facilitating adaptation', *Environmental Science and Policy*, vol. 9, no. 2, pp. 189–204.
- Blaikie, NWH 2010, *Designing Social Research*, Polity Press, Cambridge.
- Blunch, NJ 2008, Introduction to Structural Equation Modelling Using SPSS and AMOS, Sage Publications.
- BMD (Bangladesh Meteorological Department) 2013, Personal Interview, Dhaka.
- Booth, S & Smith, A 2001, 'Food security and poverty in Australia challenges for dieticians,' Australian Journal of Nutrition and Dietetics, vol. 58, no. 3, pp. 50–156.
- Bouis, HE 1994, 'The effect of income on demand for food in poor countries: Are our food consumption databases giving us reliable estimates?,' *Journal of Development Economics*, vol. 44, no. 1, pp. 199–226.
- BRAC (Bangladesh Rural Advancement Committee) 2013, BRAC Annual Report: 2013, Bangladesh Rural Advancement Committee, Dhaka.
- Brenkert, AL & Malone, EL 2005, 'Modeling vulnerability and resilience to climate change: A case study of India and Indian states', *Climatic Change*, vol. 72, no. 1, pp. 57–102.
- Brooks, N, Adger, WN & Kelly, PM 2005, 'The determinants of vulnerability and adaptive capacity at the national level and the implications for adaptation', *Global Environmental Change*, vol. 15, no. 2, pp. 151–63.
- Brown, N, Laffan, J & Wight, M 2008, 'High food prices, food security and the international trading system', Presented to the Informal National Food Pricing Summit, 29–30 September, Sydney.
- Brulle, R, Carmichael, J & Jenkins, J 2012, 'Shifting public opinion on climate change: An empirical assessment of factors influencing concern over climate change in the U.S., 2002–2010', *Climatic Change*, vol. 114, no. 2, pp. 169– 188.

- Bryan, E, Deressa, TT, Gbetibouo, GA & Ringler, C 2009, 'Adaptation to climate change in Ethiopia and South Africa: Options and constraints', *Environmental Science & Policy*, vol. 12, no. 4, pp. 413–426.
- Bryan, E, Ringler, C, Okoba, B, Roncoli, C, Silvestri, S & Herrero, M 2013, 'Adapting agriculture to climate change in Kenya: Household strategies and determinants', *Journal of Environmental Management*, vol. 114, pp. 26–35.
- Bryant, RC, Smit, B, Brklacich, M, Johnston, R, Smithers, J, Chiotti, Q & Singh, B 2000, 'Adaptation in Canadian agriculture to climate variability and change', *Climatic Change*, vol. 45, no. 1, pp. 181–201.
- Cameron, AC & Trivedi, PK 2009, Micro Econometrics Using STATA, Stata Press, Texas.
- Carletto, C, Zezza, A & Banerjee, R 2013, 'Towards better measurement of household food security: Harmonizing indicators and the role of household surveys,' *Global Food Security*, vol. 2, no. 1, pp. 30–40.
- Carney, D 1998, 'Implementing the sustainable rural livelihoods approach', In: Carney, D. (ed), Sustainable Rural Livelihoods: What Contribution Can We Make? Department of International Development, UK.
- Carpenter, SR, Arrow, KJ, Barrett, S, Biggs, R, Brock, WA, Crepin, AS & Zeeuw, A 2012, 'General resilience to cope with extreme events', *Sustainability*, vol. 4, no. 12, pp. 3248–3259.
- Carter, KN, Lanumata, T, Kruse, K & Gorton, D 2010, 'What are the determinants of food insecurity in New Zealand and does this differ for males and females?', *Australian and New Zealand Journal of Public Health*, vol. 34, no. 6, pp. 602–608.
- CEGIS (Centre for Environment and Geographic Information Services) 2006, Impacts of Sea Level Rise in the Southwest Region of Bangladesh, Center for Environmental and Geographic Information Services, Dhaka.
- CEGIS 2012, Prediction of River Bank Erosion along the Jamuna, the Ganges the Padma and the Lower Meghna Rivers in 2012, Centre for Environment and Geographic Information Services, Dhaka.

- Chamber, R & Conway, G 1992, *Sustainable Rural Livelihoods: Practical Concepts* for the 21st Century, Institute of Development Studies, UK.
- Champion, DJ 1970, *Basic Statistics for Social Research*, Chandler Publishing Company, Scranton.
- Charlton, KE & Rose, D 2002, 'Prevalence of household food poverty in South Africa: Results from a large, nationally representative survey,' *Public Health Nutrition*, vol. 5, no. 3, pp. 383–389.
- Chatterjee, N, Fernandes, G & Hernandez, M 2012, 'Food insecurity in urban poor households in Mumbai, India,' *Food Security*, vol. 4, no. 4, pp. 619–632.
- Chavas, JP, Petrie, R & Roth, M 2005, 'Farm household production efficiency: Evidence from the Gambia', *American Journal of Agricultural Economics*, vol. 87, no. 1, pp. 160–79.
- Che, J & Chen, J 2002, 'Food insecurity in Canadian households', *Health Reports*, vol. 12, no. 4, pp. 11–22.
- Choudhury, AM, Neelormi, S, Quadir, DA, Mallick, S & Ahmed, AU 2005, 'Socioeconomic and physical perspectives of water related vulnerability to climate change: Results of field study in Bangladesh', *Science and Culture*, vol. 71, no. 7-8, pp. 225–238.
- Chowdhury, N, Farid, N & Roy, D 2010, 'Food market liberalization in Bangladesh: How well did the government and the markets deliver?', In: Kumar, AG, Roy, D & Gulati, A (eds), *Liberalizing Food Grains Markets: Experiences, Impact, and Lesson from South Asia*, Oxford University Press, Oxford.
- Chowdury, MR 2000, 'An assessment of flood forecasting in Bangladesh: The experience of the 1998 flood', *Natural Hazards*, vol. 22, no. 2, pp. 139–163.
- Christoplos, I, Anderson, S, Arnold, M, Galaz, V, Hedger, M, Klein, RJT, Goulven, K 2009, *The Human Dimension of Climate Adaptation: The Importance of Local And Institutional Issues*, Commission on Climate Change and Development, Stockholm Resilience Center.
- Cleaver, K 1993, 'A strategy to develop agriculture in Sub-Saharan Africa and a focus for the World Bank', World Bank Technical Paper 203, Washington, DC.

- CLP (Chars Livelihoods Program) 2010, Impact of Earlier Recruitment into the CLP on Mother and Child Nutritional Status Among Chars Dwellers, Chars Livelihoods Program in Bangladesh, Rural Development Academy Campus, Bogra.
- Coates, J, Swindale, A & Bilinsky, P 2007, Household Food Insecurity Access Scale (HFIAS) for Measurement of Household Food Access: Indicator Guide, Washington, DC, Food and Nutrition Technical Assistance Project, Academy for Educational Development, UK.
- Coates, JC, Webb, P, Houser, RF, Rogers, BL & Wilde, P 2010, 'He said, she said': who should speak for households about experiences of food insecurity in Bangladesh?,' *Food Security*, vol. 2, no. 1, pp. 81–95.
- Cumming, GS, Barnes, G, Perz, S, Schmink, M, Sieving, KE, Southworth, J, Binford, M, Holt, RD, Stickler, C & Van, HT 2005, 'An exploratory framework for the empirical measurement of resilience', *Ecosystems*, vol. 8, no. 8, pp. 975–987.
- Cutter, SL, Barnes, L, Berry, M, Burton, C, Evans, E, Tate, E & Webb, J 2008, 'A place based model for understanding community resilience to natural disasters', *Global Environmental Change*, vol. 18, no. 4, pp. 598–606.
- Cutter, SL, Boruff, BJ & Shirley, WL 2003, 'Social vulnerability to environmental hazards', *Social Science Quarterly*, vol. 84, pp. 242–61.
- Dang, HL, Li, E, Bruwer, J & Nuberg, I 2014, 'Farmers' perceptions of climate variability and barriers to adaptation: Lessons learned from an exploratory study in Vietnam', *Mitigation and Adaptation Strategies to Global Change*, vol. 19, no. 5, pp. 531–548.
- Das, TK, Halder, SK, Gupta, ID & Sen, S 2014, 'Riverbank erosion induced human displacement and its consequences,' *Living Review in Landscape Research*, vol. 8, no.3, pp. 1–34.
- David, B & Resnik, JD 2011, *What is Ethics in Research and Why is it Important*? National Institute of Environmental Health Sciences, North Carolina.
- de Leon, V & Carlos, J 2006, Vulnerability: A Conceptual and Methodological Review, In: University, UN (eds), Studies of the University: Research

Counsel Education, Institute for Environment and Human Security, Bornheim.

- Deb, U, Hossain, M & Jones, S 2009, Rethinking Food Security Strategy: Self-Sufficiency or Self-reliance, Bangladesh Institute of Development Studies, Dhaka.
- Demaris, A 1992, *Logit Modelling: Practical Applications*, International Educational and Professional Publisher, London.
- Denton, F 2002, 'Climate change vulnerability, impacts, and adaptation: Why does gender matter?', *Gender & Development*, vol. 10, no. 2, pp. 10–20.
- Deressa, TD, Hassan, DRM & Ringler, C 2011, 'Perception of and adaptation to climate change by farmers in the Nile basin of Ethiopia', *Journal of Agricultural Science*, vol. 149, pp. 23–31.
- Deressa, TT, Hassan, RM, Ringler, C, Alemu, T & Yesuf, M 2009, 'Determinants of farmers' choice of adaptation methods to climate change in the Nile Basin of Ethiopia', *Global Environmental Change*, vol. 19, no. 2, pp. 248–255.
- Dessai, S, O'Brien, K & Hulme, M 2007, 'Editorial: On uncertainty and climate change', *Global Environmental Change*, vol. 17, pp. 1–3.
- DFID (Department of International Development) 1999, Sustainable Livelihoods Framework Guideline Sheets, Department of International Development, UK.
- DFID 2009, *Eliminating World Poverty: Building Our Common Future*, Department of International Development, UK.
- Di Falco, S, Yesuf, M, Kohlin, G & Ringler, C 2012, 'Estimating the impact of climate change on agriculture in low-income countries: Household level evidence from the Nile Basin, Ethiopia', *Environmental Resource Economics*, vol. 52, no. 4, pp. 457–478.
- Dorosh, PA & Rashid, S 2013, 'Trade subsidies, export bans and price stabilization: Lessons of Bangladesh–India rice trade in the 2000s', *Food Policy*, vol. 41, pp. 103–111.

- Dougill, AJ, Fraser, ED & Reed, MS 2010, 'Anticipating vulnerability to climate change in dryland pastoral systems: Using dynamic systems models for the Kalahari', *Ecology and Society*, vol. 15, no. 2, pp. 1–17.
- Douglas, I 2009, 'Climate change, flooding and food security in south Asia', *Food Security*, vol. 1, no. 2, pp. 127–136.
- Dreze, J & Sen, A 1989, Hunger and Public Action, Oxford University Press, Oxford.
- Drinkwater, M & Rusinow, T 1999, 'Application of CARE's livelihood approach', Paper presented at DFID Natural Resources Advisers' Conference, 1 June, London.
- Eakin, H & Bojorquez-Tapia, LA 2008, 'Insights into the composition of household vulnerability from multicriteria decision analysis', *Global Environmental Change*, vol. 18, no. 1, pp. 112–127.
- Ebi, K, Kovats, RS & Menne, B 2006, 'An approach for assessing human health vulnerability and public health interventions to adapt to climate change', *Environmental Health Perspectives*, vol. 114, pp. 1930–1934.
- Economist 2015, *Global Food Security Index 2015*, An assessment of food affordability, availability and quality, The Economist Intelligence Unit, The Economist, London.
- EGIS (Environment and Geographic Information Services) 2000, *Riverine Chars in Bangladesh-Environmental Dynamics and Management Issues*, Environment and GIS support project for Water Sector Planning, Dhaka.
- Eisenack, K 2009, 'Autonomous adaptation to climate change is inefficient', Paper presented at the European Association of Environmental and Resource Economists, February 2, Amsterdam.
- Elahi, KM 1989, 'Population displacement due to riverbank of the Jumana in Bangladesh', In: Clarke et al. (eds), *Population and Disasters*, Basil Blackwell, Oxford.
- Elahi, KM, Ahmed, KS & Malizuddin, M (eds) 1991, *Riverbank Erosion, Flood and Population Displacement in Bangladesh*, Riverbank Erosion Study, Jahangirnagar University, Dhaka.

- Ellen, R 2007, *Modern Crises and Traditional Strategies: Local Ecological Knowledge in Island Southeast Asia*, Studies in Environmental Anthropology and Ethno Biology, Oxford.
- Ellis, F 1988, *Peasant Economics*, Cambridge University Press, Cambridge.
- Ellis, F 2000, *Rural Livelihoods and Diversity in Developing Countries*, Oxford University Press, New York.
- Emel, J & Peet, R 1989, 'Resource management and natural hazards', In: Peet, R & Thrift, N (eds), *New Models in Geography*, pp. 49–76.
- Eregno, FE, Xu, CH & Kitterød, NO 2013, 'Modelling hydrological impacts of climate change in different climatic zones', *International Journal of Climate Change Strategies and Management*, vol. 5, no. 3, pp. 344 365.
- Eriksen, S & Kelly, P 2007, 'Developing credible vulnerability indicators for climate adaptation policy assessment', *Mitigation and Adaptation for Global Change*, vol. 12, no. 4, pp. 495–524.
- Eriksen, S & Lind, J 2009, 'Adaptation as a political process: Adjusting to drought and conflict in Keney's dry lands', *Environmental Management*, vol. 43, no. 5, pp. 453–461.
- Eriksen, S, Aldunce, P, Bahinipati, CS, Martins, RDA, Molefe, JI, Nhemachena, C, O'brien, K, Olorunfemi, F, Park, J & Sygna, L 2011, 'When not every response to climate change is a good one: Identifying principles for sustainable adaptation', *Climate and Development*, vol. 3, no. 1, pp. 7–20.
- Esham, M & Garforth, C 2013, 'Agricultural adaptation to climate change: Insights from a farming community in Sri Lanka', *Mitigation and Adaptation for Global Change*, vol. 18, no. 5, pp. 535–549.
- Etwire, PM, Al-Hassan, RM, Kuwornu, JKM & Osei-Owusu, Y 2013, 'Application of livelihood vulnerability index in assessing vulnerability to climate change and variability in Northern Ghana', *Journal of Environment and Earth Science*, vol. 3, no. 2, pp. 157–170.
- Evans, A 1991, 'Gender issues in rural household economics,' *IDS Bulletin*, vol. 22, no. 1, pp.1–10.

- FAO (Food and Agriculture Organization of the United Nations) 2006, *Livelihood Adaptation to Climate Change Variability and Change in Drought Prone Areas of Bangladesh: Developing Institutions and Options*, Food and Agriculture Organization of the United Nations, Rome.
- FAO 1996, *Rome Declaration on World Food Security and World Food Summit Plan* of Action, World Food Summit, 13-17 November 1996, Rome.
- FAO 2010, The State of Food Insecurity in the World: Addressing Food Insecurity in Protracted Crises, Food and Agriculture Organization of the United Nations, Rome.
- FAO 2011, The State of Food Insecurity in the World: How does International Price Volatility Affect Domestic Economies and Food Security?, Food and Agriculture Organization of the United Nations, Rome.
- FAO 2012, *The State of Food Insecurity in the World 2012*, Economic growth is necessary but not sufficient to accelerate reduction of hunger and malnutrition, Food and Agriculture Organization of the United Nations, Rome.
- FAO 2013a, *The State of Food Insecurity in the World 2013*, The multiple dimensions of food security, Food and Agriculture Organization of the United Nations, Rome.
- FAO 2013b, 'From valid measurement to effective decision making', Proceedings of the international scientific symposium on food and nutrition security information, Food and Agriculture Organization of the United Nations, Rome.
- Faridi R & Wadood, SN 2010, 'An econometric assessment of household food security in Bangladesh', *The Bangladesh Development Studies*, vol. 30, no. 3 & 4, pp. 35–62.
- Feleke, S, Kilmer, RL & Gladwin, C 2005, 'Determinants of food security in Southern Ethiopia', Agricultural Economics, vol. 33, no. 3, pp. 351–363.
- Fisher, M & Lewin, PA 2013, 'Household, community, and policy determinants of food insecurity in rural Malawi', *Development Southern Africa*, vol. 30, no. 4–5, pp. 451-467.

- Folke, C 2006, 'Resilience: The emergence of a perspective for social-ecological systems analyses', *Global Environmental Change*, vol. 16, no. 3, pp. 253–267.
- Folke, C, Carpenter, S, Elmqvist, T, Gunderson, L, Holling, CS & Walker, B 2002, *Resilience and Sustainable Development: Building Adaptive Capacity in a World of Transformations*, Report for the Swedish Environmental Advisory Council 2002, Ministry of the Environment, Stockholm.
- Ford, JD & Smit, B 2004, 'A framework for assessing the vulnerability of communities in the Canadian Arctic to risks associated with climate change', *Arct*, vol. 57, no. 4, pp. 389–400.
- Ford, JD, Keskitalo, E, Smith, T, Pearce, T, Berrang, L, Duerden, F & Smit, B 2010, 'Case study and analogue methodologies in climate change vulnerability research', *Wiley Interdisciplinary Reviews: Climate Change*, vol. 1, no. 3, pp. 374–392.
- Fowler, FJ 2009, Survey Research Methods, 4th Edition, Sage Publications.
- Frankenberger, TR & McCaston, MK 1998, 'The household livelihood security concept', Food Nutrition and Agriculture, Discussion Paper 22, Food and Agriculture Organization of the United Nations, Rome.
- Fraser, ED, Dougill, AJ, Hubacek, K, Quinn, CH, Sendzimir, J & Termansen, M 2011, 'Assessing vulnerability to climate change in dryland livelihood systems: Conceptual challenges and interdisciplinary solutions', *Ecological Sociology*, vol. 16, no. 3, pp. 33–42.
- Fresque-Baxter, JA & Armitage, D 2012, 'Place identity and climate change adaptation: A synthesis and framework for understanding', Willey Interdisciplinary Reviews: Climate Change, vol. 3, no. 2, pp. 251–266.
- Fussel, HM & Klein, RJT 2006, 'Climate change vulnerability assessments: An evolution of conceptual thinking', *Climatic Change*, vol. 75, no. 3, pp. 301– 29.
- Fussel, HM 2007, 'Vulnerability: A generally applicable conceptual framework for climate change research', *Global Environmental Change*, vol. 17, no. 2, pp. 155–167.

- Fussel, HM 2010, 'How inequitable is the global distribution of responsibility, capability, and vulnerability to climate change: A comprehensive indicatorbased assessment', *Global Environmental Change*, vol. 20, no. 4, pp. 597– 611.
- Gallopin, GC 2006, 'Linkages between vulnerability, resilience and adaptive capacity', *Global Environmental Chang*, vol. 16, no. 3, pp. 293–303.
- Gandure, S, Walker, S & Botha, JJ 2013, 'Farmers' perceptions of adaptation to climate change and water stress in a South African rural Community', *Environment and Development*, vol. 5, pp. 39-53.
- Gasper, D 2010, 'The idea of human security', In: O'Brien, KA, Clair, LS & Kristoffersen, B (eds), *Climate Change, Ethics and Human Security*, Cambridge University Press, Cambridge, UK, pp. 23-46.
- Gasson, R & Errington, A 1993, *The Farm Family Business*, CAB International, Wallingford.
- Gbetibouo, GA, Ringler, C & Hassan, R 2010, 'Vulnerability of the South African farming sector to climate change and variability: An indicator approach', *Natural Resource Forum*, vol. 34, no. 3, pp. 175–187.
- Ge, Y, Dou, W, Gu, Z, Qian, X, Wang, J & Xu, W, Shi, P, Ming, X, Zhou, X & Chen, Y 2013, 'Assessment of social vulnerability to natural hazards in the Yangtze River Delta, China,' *Stochastic Environmental Research and Risk Assessment*, vol. 27, no. 8, pp. 899–908.
- Gebrehiwot, T & van der Veen, A 2013, 'Farm level adaptation to climate change: The case of farmer's in the Ethiopian highlands', *Environmental Management*, vol. 52, no. 1, pp. 29–44.
- Gentle, P, Thwaites, R, Race, D & Alexander, K 2014, 'Differential impacts of climate change on communities in the middle hills regions of Nepal', *Natural Hazards*, vol. 74, no. 2, pp. 815–836.
- Gilbert, GN 2008, Researching Social Life, Sage Publications.
- Gilbert, SW 2010, Disaster Resilience: A Guide to the Literature, Special Publication 1117, National Institute of Standards and Technology (NIST), US Department of Commerce, USA.

- GoB (Government of Bangladesh) 2010, *Comprehensive Disaster Management Programme*', Phase II (2010-14), Ministry of Food and Disaster, Dhaka.
- GoB 2000, Report of the Task Force on Comprehensive Food Security Policy for Bangladesh, Government of the Peoples' Republic of Bangladesh, Dhaka.
- GoB 2011, *The Sixth Five-year Plan, 2011-2015*, Ministry of Planning, Government of the People's Republic of Bangladesh, Dhaka.
- Goddard, W & Melville, S 2004, *Research Methodology: An Introduction*, 2nd Edition, Blackwell Publishing.
- Gordon, J, Arbuckle, Jr, Prokopy, L, Haigh, T, Hobbs, J, Knoot, T, Knutson, C, Loy,
 A, Mase, AS, McGuire, J, Morton, LW, Tyndall, J & Widhalm, M 2013,
 'Climate change beliefs, concerns, and attitudes toward adaptation and mitigation among farmers in the Midwestern United States', *Climatic Change*, vol. 117, no. 4, pp. 943–950.
- Green, D & Raygorodetsky, G 2010, 'Indigenous knowledge of a changing climate', *Climatic Change*, vol. 100, no. 2, pp. 239–42.
- Greenberg, C 1986, 'The adaptation process of riverbank erosion displaces in an urban environment: A case study of squatters in Sirajgonj, Bangladesh', Unpublished thesis, University of Manitoba, Canada.
- Greene, WH 2012, Econometric Analysis, Seventh Edition, Prentice Hall, NJ.
- Gregory, PJ, Ingram, JSI & Brklacich, M 2005, 'Climate change and food security', *Philosophical Transactions of the Royal Society B: Biological Sciences*, vol. 360, no. 1463, pp. 2139–48.
- Groulx, M, Lewis, J, Lemieux, C & Dawson, J 2014, 'Place-based climate change adaptation: A critical case study of climate change messaging and collective action in Churchill, Manitoba', *Landscape and Urban Planning*, vol. 132, pp. 136–147.
- Gujarati, DN 2003, Basic Econometrics, McGraw-Hill, New York.
- Gunderson, L 1999, 'Resilience, flexibility and adaptive management antidotes for spurious certitude?,' *Conservation Ecology*, vol. 13, no. 1, pp. 1–7.

- Habiba, U, Shaw, R & Takeuchi, Y 2011, 'Drought risk reduction through a socioeconomic, institutional and physical approach in the north western region of Bangladesh', *Environmental Hazards*, vol. 10, no. 2, pp. 121–138.
- Habiba, U, Shaw, R & Takeuchi, Y 2012, 'Farmer's perception and adaptation practices to cope with drought: Perspectives from North Western Bangladesh', *International Journal of Disaster Risk Reduction*, vol. 1, pp. 72–84.
- Haden, VR, Niles, MT, Lubell, M, Perlman, J & Jackson, L 2012, 'Global and local concerns: What attitudes motivate farmers to mitigate and adapt to climate change', *PLoS ONE*, vol. 7, no. 12, pp. 52882–90.
- Hahn, MB, Riederer, AM & Foster, SO 2009, 'The livelihood vulnerability index: A pragmatic approach to assessing risks from climate variability and change-A case study in Mozambique', *Global Environmental Change*, vol. 19, no. 1, pp. 74–88.
- Hailu, A & Nigatu, R 2007, 'Correlates of household food security in densely populated areas of southern Ethiopia: Does the household structure matter', *Studies on Home Community Science*, vol. 1, no. 2, pp. 85–91.
- Halcrow & Associates 2001, Options for the Ganges Dependent Area: Main Report,
 Sir William Halcrow & Associates, for Water Resources Planning
 Organization (WARPO), Ministry of Water Resources (MOWR),
 Government of the People's Republic of Bangladesh, Dhaka.
- Hall, D & Bouapao, L 2010, Social Impact Monitoring and Vulnerability Assessment: Regional Report, Mekong River Commission, Vientiane.
- Hamelin, AM, Habicht, JP & Beaudry, M 1999, 'Food insecurity: Consequences for the household and border social implications', *Journal of Nutrition*, vol. 129, no. 2, pp. 525–528.
- Haque, C 1997, *Hazards in a Fickle Environment in Bangladesh*, Kluwer Academic Publishers.
- Haque, CE 1988, 'Impacts of riverbank erosion hazard in Brahmaputra Jamuna floodplain: A study of population displacement and response strategies', PhD thesis, Department of Geography, University of Mintoba, Canada.

- Hardley, C, Linzer, DA, Belachew, T, Mariam, AG, Tessema, F & Lindstrom, D 2011, 'Household capacities, vulnerabilities and food insecurity: Shifts in food insecurity in urban and rural Ethiopia during the 2008 food crisis,' *Social Science and Medicine*, vol. 73, no. 10, pp. 1534–1542.
- Harrigan, J 2008, 'Food insecurity, poverty and the Malawian starter pack: Fresh start or false start?' *Food Policy*, vol. 33, no. 3, pp. 237–249.
- Hassan, R & Nhemachena, C 2008, 'Determinants of African farmers' strategies for adapting to climate change: Multinomial choice analysis', *African Journal of Agriculture and Resource Economics*, vol. 2, no. 1, pp. 83–104.
- Headey, D & Olivier, E 2013, 'Rethinking the measurement of food security: From first principles to best practice', *Food Security*, vol. 5, no. 3, pp. 327–343.
- Headey, D 2011, 'Was the global food crisis really a crisis? Simulations versus selfreporting', International Food Policy Research Institute, Discussion Paper, Washington, DC.
- Heald, C & Lipton, M 1984, African Food Strategies and the EEC's Role: An Interim Review, Commissioned Study 6, Institute of Development Studies, Brighton.
- Henry, GT 2009, 'Practical Sampling', In: Bickman, L & Rog, DJ (eds), Applied Social Research Methods, 2nd Edition, Sage Publications.
- Hinkel, J 2011, 'Indicators of vulnerability and adaptive capacity: Towards a clarification of the science-policy interface,' *Global Environmental Change*, vol. 21, no. 1, pp. 198–208.
- Hisali, E, Birungi, P & Buyinza, F 2011, 'Adaptation to climate change in Uganda: Evidence from micro level data', *Global Environmental Change*, vol. 21, no. 4, pp. 1245–1261.
- Hiwasaki, L, Luna, E, Syamsidik & Shaw, R 2014, 'Process for integrating local and indigenous knowledge with science for hydro-meteorological disaster risk reduction and climate change adaptation in coastal and small island communities', *International Journal of Dis Risk Reduction*, vol. 10, pp. 15– 27.

- Hoddinott, J 1999, 'Choosing Outcome Indicators of Household Food Security, International Food Policy Research Institute, Washington, DC.
- Hoffmann, V, Sprengel, D, Ziegler, A, Kolb, M & Abegg, B 2009, 'Determinants of corporate adaptation to climate change in winter tourism: An econometric analysis', *Global Environmental Change*, vol. 19, no. 2, pp. 256–264.
- Horton, S & Shekar, 2010, *Scaling up Nutrition: What will it Cost?*, World Bank Publications, Washington, DC.
- Hossain, M & Deb, U 2009, 'Food Security and Containing Price Escalation: Facts and Implication for Policy', Paper presented in CPD; Development of Bangladesh with Equity and Justice: Immediate Tasks for the New Government, Centre for Policy Dialogue, March 28, Dhaka.
- Hossain, M 1993, 'Economic effects of riverbank erosion: Some evidence from Bangladesh', *Disasters*, vol. 17, no. 1, pp. 25–32.
- Hossain, M 2010, 'The 2007-08 surge in rice prices: The case of Bangladesh', In: Ahmed, S & Jansen, HGP (eds), *Managing Food Price Inflation in South Asia*, University Press Limited, Dhaka.
- Hossain, MA, Reza, MI, Rahman, S & Kayes 2012, 'Climate change and its impacts on the livelihoods of the vulnerable people in the south western coastal zone in Bangladesh', In: Leal, FW (eds) *Climate Change and the Sustainable Use* of Water Resources, Climate Change Management, Part 2, pp. 237–259.
- Hossain, MZ 1984, 'Riverbank erosion and population displacement: A case of Kazipur in Pabna', Unpublished thesis, Jahangirnagar University, Dhaka.
- Hulme, M, O'Neill, SJ & Dessai, S 2011, 'Is weather event attribution necessary for adaptation funding?' *Science*, vol. 334, no. 6057, pp.764–765.
- Huq, S & Ayers, J 2008, Climate Change Impacts and Responses in Bangladesh, Briefing note prepared for the European Parliament, International Institute for Environment and Development, London; and Policy Department Economic and Scientific Policy, DG Internal Policies of the Union, Brussels, pp. 1–20.
- Huq, S & Rabbani, G 2011, 'Climate change and Bangladesh: Policy and institutional development to reduce vulnerability', *Journal of Bangladesh Studies*, vol. 13, pp. 1–10.

- Huq, S, Ahmed, AU & Koudstaal, R 1996, 'Vulnerability of Bangladesh to Climate Change and Sea Level Rise', In: Downing, TE (eds), *Climate Change and World Food Security*, NATO ASI Series, I 37, Springer-Verlag, Berlin, pp. 347-379.
- Huq, S, Ali, SI & Rahman, AA 1995, 'Sea-level rise and Bangladesh: A preliminary analysis', *Journal of Coastal Research*, vol. 14, pp. 44-53.
- Huq, S, Karim, Z, Asaduzzaman, M & Mahtab, F (eds) 1998, *Vulnerability and Adaptation to Climate Change for Bangladesh*, Kluwer Academic Publishers.
- Hutton, D & Haque, E 2003, 'Pattern of coping and adaptation among erosion induced displacees in Bangladesh: Implications for hazard analysis and migration', *Natural Hazards*, vol. 29, no. 3, pp. 20–29.
- Hutton, D & Haque, E 2004, 'Human vulnerability, dislocation and resettlement: Adaptation process of riverbank erosion-induced displacement in Bangladesh', *Disasters*, vol. 28, no. 1, pp. 41–62.
- IFAD (International Fund for Agricultural Development) 2013, *Rural Poverty in Bangladesh*, International Fund for Agricultural Development, Rome.
- IFAD 2008, 'Climate change and future of smallholder agriculture: How can rural poor people be part of the solution to climate change?', Discussion paper prepared for the Round Table on Climate Change at the Thirty-first Session of IFAD's Governing Council, Rome.
- IFPRI (International Food Policy Research Institute) 2010, Food Security, Farming, and Climate Change: Scenarios, Results, Policy Options, International Food Policy Research Institute, Washington, DC.
- IFRC (International Federation of Red Cross and Red Crescent Societies) 2004, World Disasters Report 2004: Focus on Community Resilience, IFRC, Geneva.
- IIbery, B & Maye, D 2010, 'Agricultural restructuring and changing food networks in the UK', In: Coe, N & Jones, A (eds), *Reading the Economy: The UK in the 21st Centuary*, Sage Publications.

- IPCC 2001, *Climate Change 2001: Synthesis Report*, Contribution of Working Group I, II and III to the Third Assessment Report of the Intergovernmental Panel on Climate Change, Geneva.
- IPCC 2007, Climate Change 2007: Impacts, Adaptation and Vulnerability, Contribution of Working Group II to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change, Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge.
- IPCC 2014, Climate Change 2014: Impacts, Adaptation and Vulnerability, Part A: Global and Sectoral Aspects, Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge University Press, Cambridge.
- Islam, M & Islam, A 1985, 'A brief account of bank erosion, model studies and bank protective works in Bangladesh', *REIS Newsletter, vol.* 2, pp. 11–13.
- Islam, MM 2014, 'The contending image of food security in Bangladesh', *Journal of International Development*, vol. 26, no. 7, pp. 1054–1070.
- Islam, MR & Hossain, D 2013, 'Island *char* resource mobilization (ICRM): Changes of livelihoods of vulnerable people in Bangladesh', *Social Indicator Research*, vol. 117, no. 3, pp. 1033–1054.
- Islam, R & Walkerden, G 2015, 'How do links between households and NGOs promote disaster resilience and recovery?: A case study of linking social networks on the Bangladeshi coast', *Natural Hazards*, vol. 78, no. 3, pp. 1707–1727.
- IWFM 2012, Spatial and Temporal Distribution of Temperature, Rainfall, Sunshine and Humidity in Context of Crop Agriculture, Institute of Water and Flood Management, Bangladesh University of Engineering and Technology, Dhaka.
- Jakobsen, F, Hoque, AKMZ, Paudyal, GN & Bhuiyan, MS 2005, 'Evaluation of the short-term processes forcing the monsoon river floods in Bangladesh', *Water International*, vol. 30, no. 3, pp. 389–399.
- Jensen, AC, Mark, N, Margaret, A & Steven, C 2012, Household Food Security in the United States in 2011, Economic Research Report 141, United States Department of Agriculture, Washington, DC.

- Jones, L & Boyd, E 2011, Exploring social barriers to adaptation: Insights from Western Nepal, *Global Environmental Change*, vol. 21, no. 4, pp. 1262–1274.
- Jones, W, Eldridge, J, Silva, P, & Schiessler, N 2007, *LIFE and Europe Rivers: Protecting and Improving our Water Resources*, Life Focus, European Commission, Luxembourg.
- Jordan, JC 2015, 'Swimming alone? The role of social capital in enhancing local resilience to climate stress: A case study from Bangladesh', *Climate and Development*, vol. 7, no. 2, pp. 110–123.
- Kaiser, LL, Quinonez, HM, Townsend, M, Nicholson, Y, Fujii, ML, Martin, AC & Lamp, CL 2003, 'Food insecurity and food supplies in Latino households with young children', *Journal of Nutrition Education and Behavior*, vol. 35, no. 3, pp. 148–153.
- Karim, Z, Hussain, SG & Ahmed, M 1990, Salinity Problems and Crop Intensification in the Coastal Regions of Bangladesh, Bangladesh Agricultural Research Council (BARC), Dhaka.
- Kasperson, JX & Kasperson, RE (eds) 2001, *Global Environmental Risk*, United Nations University Press, UK.
- Kates, RW, Ausubel, JH & Berberian, M 1985, *Climate Impact Assessment*, Wiley, New York.
- Kazal, MMH, Villinueva, CC, Hossain, MZ & Das, TK 2010, Food Security Strategies of the People Living in Haor Areas: Status and Prospects, NFPCSP Final Report, National Food Policy Capacity Strengthening Program, Ministry of Food, Dhaka.
- Keating, A 2013, Food Security in Australia: The Logistics of Vulnerability, Food security in Australia: Challenges and prospects for the future, Springer science, New York.
- Kennedy, E 2002, 'Qualitative measures of food insecurity and hunger', Paper presented at International Scientific Symposium on Measurement and Assessment of Food Deprivation and Undernutrition, 26-28 June, 2002, Rome.

Kennedy, P 1998, A Guide to Econometrics, 4th Edition, The MIT press, Cambridge.

- Klein, RJT, Nicholls, RJ & Thomalla, F 2003, 'Resilience to natural hazards: How useful is this concept?', *Environmental Hazards*, vol. 5, no. 1, pp. 35–45.
- Kropko, J 2007, 'Choosing between multinomial logit and multinomial probit models for analysis of unordered choice data', Masters thesis, The University of North Carolina. USA.
- Krugman, P 2011, *Droughts, Floods and Food*, The New York Times, February 13, 2011.
- Kulig, JC, Edge, DS, Townshend, I, Lightfoot, N & Reimer, W 2013, 'Community resiliency: Emerging theoretical insights,' *Journal of Community Psychology*, vol. 41, no. 6, pp. 758–775.
- Kurukulasuriya, P & Mendelsohn, R 2008, 'A Ricardian analysis of the impact of climate change on African crop land', *African Journal of Agriculture and Resource Economics*, vol. 2, no. 1, pp. 1–23.
- Lang, T & Barling, D 2012, 'Food security and food sustainability: Reformulating the debate,' *The Geographical Journal*, vol. 178, no. 4, pp. 313–326.
- Lee, YJ 2014, 'Social vulnerability indicators as a sustainable planning tool', Environmental Impact Assess Review, vol. 44, pp. 31–42.
- Lein, H 2010, 'Hazards and force migration in Bangladesh', Norwegian Journal of Geography, vol. 54, no. 3, pp. 122–127.
- Likert, AR 1932, "A technique for the measurement of attitudes', Archives of Psychology, vol. 22, no. 140, pp. 1–55.
- Lin, J 1991, 'Education and innovation adoption in agriculture: Evidence from hybrid rice in China', American Journal of Agricultural Economics, vol. 73, no. 3, pp. 713–723.
- Lobell, BD, Burke, MB, Tebaldi, C, Mastrandream, MD, Falcon, WP & Naylor, RL 2008, 'Prioritizing climate change adaptation needs for food security in 2030', *Science*, vol. 319, no. 5863, pp. 607–610.

- Lockie, S & Pietsch, J 2011, 'Household food security in Australia', Paper presented at the XVIII Meeting of the Australian and New Zealand Agri-food Research Network, Australian National University, 5-8 December, Canberra.
- Long, JS 1997, Regression Models for Categorical and Limited Dependent Variables, Sage Publications.
- Løvendal, CR, Knowles, M & Horii, N 2004, 'Understanding vulnerability to food insecurity, lesson from vulnerable livelihood profiling', ESA working paper 04-18, Food and Agriculture Organization of the United Nations, Rome.
- Luers, AL, Lobell, DB, Sklar, LS, Addams, CL & Matson, PA 2003, 'A method for quantifying vulnerability, applied to the agricultural system of the Yaqui Valley, Mexico', *Global Environmental Change*, vol. 13, no. 4, pp. 255–267.
- Luthar, SS, Cicchetti, D & Becker, B 2000, 'The construct of resilience: A critical evaluation and guidelines for future work', *Child Development*, vol. 71, no. 3, pp. 543–562.
- Macchi, M, Gurung, AM & Hoermann, B 2014, 'Community perceptions and responses to climate variability and change in the Himalayas', *Climate and Development*, vol. 7, no. 5, pp. 1–12.
- MacFarquhar, N 2011, Food Prices Worldwide Hit Record Levels, Fuelled by Uncertainty, UN Says, The New York Times, February 3, 2011.
- Maddala, GS 1992, Introduction to Econometrics, Macmillan Publishing Company, New York.
- Maddison, D 2007, 'The perception of and adaptation to climate change in Africa', World Bank Policy Research Working Paper 4308, World Bank, Washington, DC.
- Magis, K 2010, 'Community resilience: An indicator of social sustainability', *Society* & *Natural Resources*, vol. 23, no. 5, pp. 401–416.
- Makenro, MB 2000, *World Disaster Report 2001*, International Federation of the Red Cross and Red Crescent Societies, Geneva.

- Mallick, B & Vogt, J 2012, 'Cyclone, coastal society and migration: Empirical evidence from Bangladesh', *International Development Planning Review*, vol. 34, no. 3, pp. 217–240.
- Mallick, D & Rafi, M 2010, 'Are female-headed households are more food insecure? Evidence from Bangladesh', *World Development*, vol. 38, no. 4, pp. 593–605.
- Malone, EL & Brenkert, AL 2008, 'Uncertainty in resilience to climate change in India and Indian states', *Climatic Change*, vol. 91, no. 3, pp. 451–476.
- Malone, EL 2009, Vulnerability and Resilience in the Face of Climate Change, Current Research and Needs for Population Information, Population Action International, Washington, DC.
- Mannaf, M & Uddin, MT 2012, 'Socioeconomic factors influencing food security status of maize growing households in selected areas of Bogra district', *Bangladesh Journal of Agricultural Economics*, vol. XXXV, no. 1&2, pp. 177–187.
- Manyena, SB 2006, 'The concept of resilience revisited', *Disasters*, vol. 30, no. 4, pp. 433-450.
- Mariara, JK, Ndenge, GK & Kirii, DM 2006, 'Determinants of children's nutritional status in Kenya: Evidence from demographic and health surveys', Paper presented at the conference on Reducing poverty and inequality: How can Africa be included? Centre for the Study of African Economies (CSAE), Oxford.
- Marshall, N, Park, S, Adger, W, Brown, K & Howden, S 2012, 'Transformational capacity and the influence of place and identity', *Environmental Research Letters*, vol. 7, no. 3, pp. 1–9.
- Marshall, NA & Marshall, PA 2007, 'Conceptualizing and operationalizing social resilience within commercial fisheries in northern Australia', *Ecological Sociology*, vol. 12, no. 1, pp. 1–10.
- Marshall, NA 2010, 'Understanding social resilience to climate variability in primary enterprises and industries', *Global Environmental Change*, vol. 20, no. 1, pp. 36–43.

- Maru, YT, Smith, MS, Sparrow, A, Pinho, P & Dube, OP 2014, 'A linked vulnerability and resilience framework for adaptation pathways in remote disadvantaged communities', *Global Environmental Change*, vol. 28, pp. 337–350.
- Masten, AS, Best, KM & Garmezy, N 1990, 'Resilience and development: Contributions from the study of children who overcome adversity', *Development and Psychopathology*, vol. 2, no. 4, pp. 425–444.
- Matarira, CH, Pullanikkatil, D, Shava, E, Kaseke, T & Manatsa, D 2012, 'Socioeconomic impacts of climate change on subsistence communities: Some observation from Lesotho', *International Journal of Climate Change Strategies and Management*, vol. 5, no. 4, pp. 404–417.
- Maxwell, D, Coates, J & Vaitla, B 2013, How do Different Indicators of Household Food Security Compare? Empirical Evidence from Tigray', Feinstein International Center, Tufts University, Medford, USA.
- Maxwell, D, Watkins, B, Wheeler, R & Collins, G 2003, The Coping Strategies Index: A tool for Rapidly Measuring Food Security and the Impact of food Aid Programs in Emergencies, Field Methods Manual, CARE and WFP Vulnerability Assessment Mapping.
- Maxwell, S & Slater, R 2003, 'Food policy old and new', *Development Policy Review*, vol. 21, no. 5-6, pp. 531–553.
- Maxwell, S & Smith, M 2006, *Household Food Insecurity: A Conceptual Review*, Institute of Development Studies, University of Sussex.
- Maxwell, S 1994, 'Food security: A post-modern perspective', Working Paper 9, Institute of Development Studies, University of Sussex.
- Mazumder, MSU & Lu, W 2015, 'What impacts does microcredit have on rural livelihood? A comparison of governmental and non-governmental microfinance programs in Bangladesh', *World Development*, vol. 68, pp. 336–354.
- McCarthy, JJ 2001, *Climate Change 2001: Impacts, Adaptation, and Vulnerability,* Contribution of working group II to the third assessment report of the

intergovernmental panel on climate change (IPCC), Cambridge University Press, Cambridge.

- McDowell, G, Ford, J, Lehner, B, Berrang-Ford, L & Sherpa, A 2013, 'Climaterelated hydrological change and human vulnerability in remote mountain regions: A case study from Khumbu, Nepal', *Regional Environmental Change*, vol. 13, no. 2, pp. 299–310.
- Mertz, O, Mbow, C, Reenberg, A & Diouf, A 2009, 'Farmers' perceptions of climate change and agricultural adaptation strategies in rural Sahel', *Environment Management*, vol. 3, no. 5, pp. 804–816.
- Mirza, MMQ 2003, 'Climate change and extreme weather events: Can developing countries adapt?', *Climate Policy*, vol. 3, no. 3, pp. 233–248.
- Mishra, AK, Mottaleb, KA & Mohanty, S 2015, 'Impact of off-farm income on food expenditures in rural Bangladesh: An unconditional quantile regression approach', *Agricultural Economics*, vol. 46, no. 2, pp. 139–148.
- MOEF (Ministry of Environment and Forest) 2005, National Adaptation Program of Action (NAPA), Ministry of Environment and Forest, Dhaka.
- Mohapatra, S, Joseph, G & Ratha, D 2012, 'Remittances and natural disasters: Expost response and contribution to ex-ante preparedness', *Environmental Development Sustainability*, vol. 14, no. 3, pp. 1–23.
- Molua, EL 2009, 'An empirical assessment of the impact of climate change on smallholder agriculture in Cameroon', *Global Planetary Change*, vol. 67, no. 3-4, pp. 205–208.
- Muniruzzaman, ANM 2013, 'Food security in Bangladesh: A comprehensive analysis', *Peace and Security Review*, vol. 5, no. 10, pp. 46-73.
- Moser, C 1996, Confronting Crisis: A Comparative Study of Household Responses to Poverty and Vulnerability in Four Poor Urban Communities, Environmentally Sustainable Development Studies and Monographs Series 8, World Bank, Washington, DC.
- Moss, RH, Brenkert, AL & Malone, EL 2001, Vulnerability to Climate Change: A Quantitative Approach, Pacific Northwest National Laboratory, Washington, DC.

- Muijs, D 2010, *Doing Quantitative Research in Education with SPSS*, Sage Publications.
- Murgai, R & Zaidi, S 2005, 'Effectiveness of food assistance programs in Bangladesh', *Journal of Developing Societies*, vol. 21, no. 1-2, pp. 121–142.
- Murungweni, C, van Wijk, MT, Giller, KE, Anderson, JA & Smaling, EMA 2014, 'Adaptive livelihood strategies employed by farmers to close the food gap in semi-arid south eastern Zimbabwe', *Food Security*, vol. 6, no. 3, pp. 313–326.
- Musemwa, L, Muchenje, V, Mushunje, A, Aghdasi, F & Zhou, L 2015, 'Household food insecurity in the poorest province of South Africa: Level, causes and coping strategies', *Food Security*, vol. 7, no. 3, pp. 647–655.
- Narayan, D 1997, Voices of the Poor: Poverty and Social Capital in Tanzania. Environmentally Sustainable Development Monograph 20, World Bank, Washington, DC.
- Nelson, DR 2011, 'Adaptation and resilience: Responding to a changing climate', *Wiley Interdisciplinary Review: Climate Change*, vol. 2, no. 1, pp. 113–20.
- Nelson, DR, Adger, WN & Brown, K 2007, 'Adaptation to environmental change: Contributions of a resilience framework,' *Annual Review of Environmental Resources*, vol. 32, pp. 395–419.
- Neuman, WL 2006, Social Research Methods: Qualitative and Quantitative Approaches, 6th Edition, Pearson Education Inc., Boston.
- Nhemachena, C & Hassan, R 2007, 'Micro-level analysis of farmers' adaptation to climate change in Southern Africa', IFPRI Discussion Paper 00714, International Food Policy Research Institute, Washington, DC.
- Niang, I, Ruppel, OS, Abdrabo, MA, Essel, A, Lennard, C, Padgham, J & Urquhart, P 2014, *Climate Change 2014: Impacts, Adaptation and Vulnerability*, Cambridge University Press, Cambridge.
- Niles, MT, Lubell, M & Brown, M 2015, 'How limiting factors drive agricultural adaptation to climate change', *Agriculture, Ecosystem and Environment*, vol. 200, pp. 178–185.

- Niles, MT, Lubell, M & Haden, VR 2013, 'Perceptions and responses to climate policy risks among California farmers', *Global Environment Change*, vol. 23, no. 6, pp. 1752–1760.
- Nord, M, Andrews, M & Carlson, S 2008, *Household Food Security in the United States*, 2008, Research Report 83, United State Department of Agriculture.
- Nyong, A, Adesina, F & Elasha, BO 2007, 'The value of indigenous knowledge in climate change mitigation and adaptation strategies in the African Sahel', *Mitigation and Adaptation Strategies for Global Change*, vol. 12, no. 5, pp. 787–97.
- O'Brien, KL, Eriksen, S, Schjolden, A & Nygaard, L 2009, *What's in a Word? Conflicting Interpretations of Vulnerability in Climate Change Research*, Centre for International Climate and Environmental Research, Oslo.
- Ojogho, O 2009, 'Determinants of food insecurity among arable farmers in Oredo and Egor local government areas of Edo State, Nigeria: Policy for mitigating food crises,' *Social Science Research Network*, vol. 05, no. 3, pp. 12–22.
- Omotesho, OA, Adewumi, LM, & Ayinde, O 2006, 'Analysis of food security among rural households in Kwara State, Nigeria', *African Journal of General Agriculture*, vol. 2, no. 1, pp. 7–15.
- Onianwa, OO & Wheelock, GC 2006, 'An analysis of the determinants of food insecurity with severe hunger in selected southern states', *Southern Rural Sociology*, vol. 21, no. 1, pp. 80–96.
- Paavola, J 2008, 'Livelihoods, vulnerability and adaptation to climate change in Morogoro, Tanzania', *Environment Science & Policy*, vol. 11, no. 7, pp. 642– 654.
- Pandey, R & Jha, S 2012, 'Climate vulnerability index-measure of climate change vulnerability to communities: A case of rural lower Himalaya, India', *Mitigation and Adaptation Strategies for Global Change*, vol. 17, no. 5, pp. 487–506.
- Panthi, J, Aryal, S, Dahal, P, Bhandari, P, Krakauer, NY & Pandey, VP 2015, 'Livelihood vulnerability approach to assessing climate change impacts on

mixed agro-livestock smallholders around the Gandaki River Basin in Nepal', *Regional Environmental Change* (In Press).

- Parvin, GA & Shaw, R 2012, 'Microfinance institutions and a coastal community's disaster risk reduction, response, and recovery process: A case study of Hatiya, Bangladesh', *Disasters*, vol. 37, no. 1, pp. 165–184.
- Parvin, GA, Takahashi, F & Shaw, R 2008, 'Coastal hazards and community coping methods in Bangladesh', *Journal of Coast Conservation*, vol. 12, no. 4, pp. 181–93.
- Patt, AG & Gwata, C 2002, 'Effective seasonal climate forecast applications examining constraints for subsistence farmers in Zimbabwe', *Global Environmental Change*, vol. 12, pp. 185–195.
- Patt, AG, Ogallo, L & Hellmuth, M 2007, 'Learning from 10 years of climate outlook forums in Africa', *Science*, vol. 318, no. 3, pp. 49–50.
- Paul, BK 1998, 'Coping mechanisms practised by drought victims (1994/5) in North Bengal, Bangladesh', *Applied Geography*, vol. 18, no. 4, pp. 355–373.
- Paul, BK 2010, 'Human injuries caused by Bangladesh's cyclone Sidr: An empirical study', *Natural Hazards*, vol. 54, pp. 483–95.
- Penning-Rowsell, EC, Sultana, P & Thompson, PM 2013, 'The 'last resort'? Population movement in response to climate-related hazards in Bangladesh', *Environ Science & Policy*, vol. 27, no. 1, pp. 44–59.
- Perez, C, Jones, EM, Kristjanson, P, Cramer, L, Thornton, PK, Forch, W & Barahona, C 2015, 'How resilient are farming households and communities to a changing climate in Africa? A gender-based perspective', *Global Environmental Change*, vol. 34, pp. 95–107.
- Perez-Escamilla, R & Segall-Correa, AM 2008, 'Food security measurement and indicators', *Review of Nutrition Campions*, vol. 21, pp. 15–26.
- Perrings, C 2006, 'Resilience and sustainable development', *Environment and Development Economics*, vol. 11, no. 4, pp. 417–427.
- Perry, C 1998, 'A structured approach for presenting theses', Australian Marketing Journal, vol. 6, no. 1, pp. 63–85.
- Peterson, G 2000, 'Political ecology and ecological resilience: An integration of human and ecological dynamics', *Ecological Economics*, vol. 35, no. 3, pp. 323–36.
- Pinstrup-Anderson, P 2009, 'Food security: Definition and measurement', *Food Security*, vol. 1, no. 1, pp. 5–7.
- Poppy, GM, Chiotha, S, Eigenbrod, F, Harvey, CA, Honzak, M, Hudson, MD, Jarvis, A, Madise, NJ, Schreckenberg, K, Shackleton, CM, Villa, F & Dawson, TP 2014, 'Food security in a perfect storm: Using the ecosystem services framework to increase understanding', *Philosophical Transactions of the Royal Society B: Biological Sciences*, vol. 369, no. 1639, pp. 1–13.
- Pouliotte, J, Smit, B & Westerhoff, L 2009, 'Adaptation and development: Livelihoods and climate change in Subarnabad, Bangladesh', *Climate and Development*, vol. 1, no. 1, pp. 31–46.
- Predo, C 2010, Adaptation of Community and Households to Climate-Related Disaster: The Case of Storm Surge and Flooding Experience in Ormoc and Cabalian Bay, Philippines, Climate Change Technical Reports, Economy and Environment Program for Southeast Asia (EEPSEA).
- Prescott-Allen, R (2001) *The Wellbeing of Nations: A Country-by-Country Index of Quality of Life and the Environment*, Island Press.
- Preston, BL, Yuen, EJ & Westaway, RM 2011, 'Putting vulnerability to climate change on the map: A review of approaches, benefits, and risks', *Sustainability Science*. vol. 6, no. 2, pp. 177–202.
- Prowse, M & Scott, L 2008, 'Assets and adaptation: An emerging debate', *IDS Bulletin*, vol. 39, no. 4, pp. 42–52.
- Quisumbing, AR 2013, 'Generating evidence on individuals' experience of food insecurity and vulnerability', *Global Food Security*, vol. 2, no. 1, pp. 50–55.
- Rahman, KMM & Poza, AS 2010, 'Food consumption and nutritional status of vulnerable people rearing livestock in Bangladesh', *Bangladesh Journal of Agricultural Economics*, vol. XXXIII, no. 1 & 2, pp. 41–56.

- Rahman, RI, Begum, A & Bhuyan, HR 2012, Household Food Insecurity in Bangladesh: Concepts, Estimates and Determinants, Bangladesh Institute of Development Studies (BIDS), Dhaka.
- Rashid, MH, Afroz, S, Gaydon, D, Muttaleb, A, Poulton, P, Roth, C & Abedin, Z 2014, 'Climate change perception and adaptation options for agriculture in Southern Khulna of Bangladesh', *Applied Ecology and Environment Science*, vol. 2, no. 1, pp. 25–31.
- Ravallion, M 1997, 'Famines and Economics', *Journal of Economic Literature*, vol. 37, no. 3, pp. 1205–1242.
- Ravi, A 2008, 'Climate change risk: An adaptation and mitigation agenda for Indian cities', *Environment and Urbanization*, vol. 20, no. 1, pp. 207–229.
- Rawlani, AK & Sovacool, BK 2011, 'Building responsiveness to climate change through community based adaptation in Bangladesh', *Mitigation and Adaptation Strategies of Global Change*, vol. 16, no. 3, pp. 845–863.
- Ray-Bennett, NS, Collins, A, Bhuiya, A, Edgeworth, R, Nahar, P & Alamgir, F 2010, 'Exploring the meaning of health security for disaster resilience through people's perspectives in Bangladesh', *Health Place*, vol. 10, no. 16, pp. 581–9.
- Reardon, T 1997, 'Using evidence of household income diversification to inform study of the rural nonfarm labor market in Africa', *World Development*, vol. 25, no. 5, pp. 735–747.
- Reddy, V 1997, Dietary Assessment Methodologies Relevant to the Development of Food-based Interventions, Published by the Food and Agricultural Organization, Regional Office for Asia and the Pacific.
- Reid, S, Smit, B, Caldwell, W & Belliveau, S 2007, 'Vulnerability and adaptation to climate risks in Ontario agriculture', *Mitigation and Adaptation Strategies of Global Change*, vol. 12, no. 4, pp. 609–637.
- Reji, EM 2013, 'Community grain banks and food security of the tribal poor in India', *Development in Practice*, vol. 23, no. 7, pp. 920–933.
- Rice, PD & Thomas, A 1985, 'The economic costs of illness: A replication and update', *Health Care Financing Review*, vol. 7, no. 1, pp. 61–80.

- Rich, M, Lesley, O, Kavita, S, Golam, K & Setara, R 2015, Food Security Country Framework for Bangladesh (FY 2015–2019), USAID Office of Food for Peace, Washington, DC.
- Rinaldi, M 2003, 'Recent channel adjustments in alluvial rivers of Tuscany, central Italy', *Earth Surface Processes and Landforms*, vol. 28, no. 6, pp. 587–608.
- Rodrigue, JP, Comtois, C & Slack, B 2009, *The Geography of Transport Systems*, 2nd Edition, Routledge.
- Rogge, J & Elahi, K 1989, *The Riverbank Impact Study: Bangladesh*, University of Manitoba, Canada.
- Rogge, J & Haque, C 1987, 'Riverbank erosion hazard, rural population displacement, and institutional responses and policies in Bangladesh', Paper presented at the annual meeting of the Association of American Geographers, Portland, Oregon.
- Romanowski, J 1987, 'Impact of riverbank erosion on agricultural productivity: A case study of Bangladesh', 21st Bengal Studies Conference, Osaka, USA.
- Rosenzweig, C, Elliott, J, Deryng, D, Ruane, AC, Müller, C, Arneth, A, Boote, KJ, Folberth, C, Glotter, M, Khabarov, N, Neumann, K, Piontek, F, Pugh, TAM, Schmid, E, Stehfest, E, Yang, H & Jones, JW 2013, 'Assessing agricultural risks of climate change in the 21st century in a global gridded crop model intercomparison', *Proceedings of National Academy of Science*, USA, vol. 111, no. 9, pp. 3268–3273.
- Salick, J, Fang, Z & Byg, A 2009, 'Eastern Himalayan alpine plant ecology, Tibetan ethnobotany, and climate change', *Global Environmental Change*, vol. 19, no. 2, pp. 147–155.
- Sapountzaki, K 2012, 'Vulnerability management by means of resilience', *Natural Hazards*, vol. 60, no. 3, pp. 1267–1285.
- Sarker, MAR, Alam, K & Gow, J 2013, 'Assessing the determinants of rice farmers' adaptation to climate change in Bangladesh,' *International Journal of Climate Strategies and Management*, vol. 5, no. 4, pp. 382–403.

- Sarker, MH, Haque, I, Alam, M & Koudstal, R 2003, Rivers, chars and char dwellers of Bangladesh, *International Journal of River Basin Management*, vol. 1, no. 1, pp. 61–80.
- Schad, I, Schmitter, P, Saint-Macary, C, Neef, A, Lamers, M, Nguyen, L, Hilger, T
 & Hoffmann, V 2012, 'Why do people not learn from flood disasters? Evidence from Vietnam's north western mountains', *Natural Hazards*, vol. 62, no. 2, pp. 221–241.
- Schmidhuber, J & Tubiello, FN 2007, 'Global food security under climate change', Proceedings of National Academic Science, USA, vol. 104, no. 50, pp. 19703–8.
- Schneiderbauer, S & Ehrlich, D 2006, 'Social levels and hazard independence in determining vulnerability', In: Birkmann, J (eds), *Measuring Vulnerability to Natural Hazards: Towards Disaster Resilient Societies*, United Nations University Press, Tokyo.
- Sen, A 1981, Poverty and Famines: An Essay on Entitlement and Deprivation, Oxford University Press, Oxford.
- Seo, SN 2011, 'An analysis of public adaptation to climate change using agricultural water schemes in South America', *Ecological Economics*, vol. 70, no. 4, pp. 825–834.
- Shah, KU, Dulal, HB, Johnson, C & Baptiste, A 2013, 'Understanding livelihood vulnerability to climate change: Applying the livelihood vulnerability index in Trinidad and Tobago', *Geoforum*, vol. 47, pp. 125–137.
- Shahabuddin, Q 2010, 'The right to food: Bangladesh perspectives', *The Bangladesh Development Studies*, vol. XXXIII, no. 1&2, pp. 91–134.
- Shaheen, N, Rahim, ATMA, Mohiduzzaman, M, Banu, CP, Bari, ML, Tukun, AB, Mannan, MA, Bhattacharjee, L & Stadlmayr, B 2013, *Food Composition Table for Bangladesh*, NFPCSP Final Report, National Food Policy Capacity Strengthening Program, Ministry of Food, Dhaka.
- Shahid, S & Behrawan, H 2008, 'Drought risk assessment in the western part of Bangladesh,' *Natural Hazard*, vol. 46, no. 3, pp. 391–413.

- Shaikh, FM 2007, 'Determinants of household food security and consumption pattern in rural sindh: non-separable agricultural household model', *IUB Journal of Social Sciences and Humanities*, vol. 5, no. 2, pp. 18–39.
- Sharif, ZM & Merlin, A 2001, 'Assessment of food insecurity among low income households in Kuala Lumpur using the Cornell food insecurity instrument- A validity study', *Malaysian Journal of Nutrition*, vol. 7, no. 1, pp. 15–32.
- Shaw R, Mallick F, Islam, A 2013, 'Climate change: Global perspectives,' In: Shaw, R, Mallick, F & Islam, A (eds), *Climate Change Adaptation Actions in Bangladesh*, Springer, Tokyo, pp. 3-14.
- Shisanya, CA & Khayesi, M 2007, 'How is climate change perceived in relation to other socioeconomic and environmental threats in Nairobi, Kenya?', *Climatic Change*, vol. 85, no. 3, pp. 271–284.
- Silverman, S & Solman, M 1998, 'The unit of analysis in field research: issues and approaches to design and data analysis', *Journal of Teaching in Physical Education*, vol. 17, no. 3, pp. 270-284.
- Sindhu, RS, Kaur, I & Vatta, K 2008, 'Food and nutritional insecurity and its determinants in food surplus areas: The case study of Punjab state', *Agricultural Economics Research Review*, vol. 21, no. 1, pp. 91–98.
- Singh, K 2007, Quantitative Social Research Methods, Sage Publications.
- Slegers, MFW 2008, 'If only it would rain: Farmers' perceptions of rainfall and drought in semi-arid central Tanzania', *Journal of Arid Environment*, vol. 72, no. 11, pp. 2106–2123.
- Smit, B & Pilifosova, O 2001, Adaptation to Climate Change in the Context of Sustainable Development and Equity, Contribution of the Working Group to the Third Assessment Report of the Intergovernmental Panel on Climate Change, Cambridge University Press, Cambridge.
- Smit, B & Wandel, J 2006, 'Adaptation, adaptive capacity and vulnerability', Mitigation and Adaptation Strategies of Global Change, vol. 16, no. 3, pp. 282–292.
- Smit, B, Burton, I, Klein, R & Wandel, J 2000, 'An anatomy of adaptation to climate change and variability', *Climatic Change*, vol. 45, no. 1, pp. 223–251.

- Sperber, AD 2004, 'Translation and validation of study instruments for cross-cultural research', *Gastroenterology*, vol. 126, no. 1, pp. 124–128.
- Stern, N 2006, 'What is the economics of climate change?', World Economics, vol. 7, no. 2, pp. 1–10.
- Stiglitz, JE 1976, 'The efficiency wage hypothesis, surplus labor, and distribution income in LDCs', *Oxford Economic Papers*, vol. 28, no. 2, pp. 185–207.
- Stokes, CJ & Howden, SM (eds) 2009, Adapting Agriculture to Climate Change: Preparing Australian Agriculture, Forestry and Fisheries for the Future, The Commonwealth Scientific and Industrial Research Organization (CSIRO), Canberra.
- Strauss, J 1983, 'Socioeconomic determinants of food consumption and production in rural Sierra Leone: Application of an agricultural household model with several commodities', MSU International Development Papers, Department of Agricultural Economics, Michigan State University, East Lansing, Michigan.
- Stringer, LC, Dyer, JC, Reed, MS, Dougill, AJ, Twyman, C & Mkwambisi, D 2009, 'Adaptations to climate change, drought and desertification: Local insights to enhance policy in southern Africa', *Environmental Science & Policy*, vol. 12, no. 7, pp. 748–765.
- Stringer, R 2000, 'Food security in developing countries', Policy Discussion Paper 0011, Center for International Economic Studies, University of Adelaide, Australia.
- Stuff, JE, Casey, PH, Szeto, K & Gossett, JM 2004, 'Household food insecurity is associated with adult health status', *Journal of Nutrition*, vol. 134, no. 9, pp. 2330–35.
- Sullivan, C, Meigh, JR & Fediw, TS 2002, Derivation and Testing of the Water Poverty Index Phase 1, Final Report, Department for International Development, UK.
- Swinnen, J & Squicciarini, P 2012, 'Mixed messages on prices and food security', *Science*, vol. 335, no. 6067, pp. 405–406.

- Szabo, S, Hossain, MS, Adger, WN, Matthews, Z, Ahmed, S, Lazar, AN & Ahmad, S 2015, 'Soil salinity, household wealth and food insecurity in tropical deltas: Evidence from south-west coastal of Bangladesh', *Sustainability Science*, (In Press).
- Talukder, RK 2005, 'Food security, self-sufficiency and nutrition gap in Bangladesh', *The Bangladesh Development Studies*, vol. 30, no. 3 & 4, pp. 35–62.
- Tarasuk, V, Mitchell, A & Dachner, N 2013, Household Food Insecurity in Canada 2011, Research to identify policy options to reduce food insecurity, PROOF, Canadian Institutes of Health Research (CIHR), Canada.
- Taylor, JG, Stewart, TR & Downton, M 1988, 'Perceptions of drought in the Ogallala Aquifer region', *Environment and Behavior*, vol. 20, no. 2, pp. 150– 175.
- Telema, R, Yang, X, Viikari, J, Valimaki, I, Wanne, O & Raitakari, O 2005, 'Physical activity from childhood to adulthood: A 21-year tracking study', *American Journal of Prevent Medicine*, vol. 28, pp. 267–273.
- Thomas, DS & Twyman, C 2005, 'Equity and justice in climate change adaptation amongst natural-resource-dependent societies', *Global Environmental Change*, vol. 15, no. 2, pp. 115–24.
- Thomas, TS, Mainuddin, K, Chiang, C, Rahman, A, Haque, A, Islam, N & Sun, Y 2013, 'Agriculture and adaptation in Bangladesh: Current and projected impacts of climate change', Discussion Paper 01281, International Food Policy Research Institute, Washington, DC.
- Thompson, PM 2000, Bangladesh Char Lands: A Review of Assets and Change, International Centre for Living Aquatic Resource Management (ICLARM), Dhaka.
- Thornton, PK, Jones, PG, Alagarswamy, G, Andersen, J & Herrero, M 2010, 'Adaptation to climate change: Agricultural system and household impacts in east Africa', *Agricultural System*, vol. 103, no. 2, pp. 73–82.

- Titus, OB & Adetokunbo, AG 2007, 'An analysis of food security situation among Nigerian urban households: Evidence from Lagos State, Nigeria', *Journal of Central European Agriculture*, vol. 8, no. 3, pp. 397–406.
- Todaro, MP & Smith, SC 2005, *Economic Development*, 8th Edition, Pearson Education Pre. Ltd.
- Tompkins, EL & Adger, WN 2004, 'Does adaptive management of natural resources enhance resilience to climate change?' *Ecology and Society*, vol. 9, no. 2, pp. 1–14.
- Toufique, KA & Islam, M 2014, 'Assessing risk from climate variability and change for disaster-prone zones in Bangladesh', *International Journal of Disaster Risk Reduction*, vol. 10, pp. 236–249.
- Traerup, SLM 2012, 'Informal networks and resilience to climate change impacts: A collective approach to index insurance', *Global Environmental Change*, vol. 22, no. 1, pp. 255–267.
- Troy, DA, Carson, A, Vanderbeek, J & Hutton, A 2008, 'Enhancing communitybased disaster preparedness with information technology', *Disasters*, vol. 32, no. 1, pp. 149–165.
- Turner, BL, Kasperson, RE, Matsone, PA, McCarthy, JJ, Corell, RW, Christensen, L, Eckleyg, N, Kasperson, ZX, Luers, A, Martello, ML, Polsky, C, Pulsipher, A & Schiller, A 2003, 'A framework for vulnerability analysis in sustainability science', *Proceedings of National Academy of Science*, vol. 100, no. 14, pp. 8074–8079.
- Ullah, MS 2010, 'Community radio movement in Bangladesh: Will they work as citizens' media?', *Global Media Journal*, vol. Summer, pp. 1–11.
- UN (United Nations) 2013, *The Millennium Development Goals Report 2013*, United Nations, New York.
- UN/ISDR 2005, Hyogo Framework for Action 2005–2015: Building the Resilience of Nations and Communities to Disaster, United Nations Office for Disaster Risk Reduction (UNISDR), New York.

- UNDP (United Nations Development Program) 2005, Sustainable Livelihoods: Concepts, Principles and Approaches to Indicator Development, United Nations Development Program, New York.
- UNEP (United Nations Environment Program) 2001, *Global Environment Outlook Report 2000*, United Nations Environment Program, Kenya.
- UNEP (United Nations Environment Program) 2008, *Climate Change Strategy*, United Nations Environment Program, Nairobi.
- UNESCO (The United Nations Educational, Scientific and Cultural Organization) 2015, *Literacy Statistics Metadata Information Table*, Institute for Statistics, September 2015, UNESCO, Geneva.
- UNISDR (United Nations International Strategy for Disaster Reduction) 2004, Living With Risk: A Global Review of Disaster Reduction Initiatives, Inter-Agency Secretariat of the United Nations International Strategy for Disaster Reduction, Geneva.
- UNSCN 2010, United Nations Standing Committee on Nutrition 16th United Nations conference of the Parties (COP16) Cancun, November 29-December 10, 2010, Cancun.
- USDA (United States Department of Agriculture) 2010, *Bangladesh Grain and Feed Annual*, United States Department of Agriculture, Washington, DC.
- Vedwan, N & Rhoades, RE 2001, 'Climate change in the Western Himalayas of India: A study of local perception and response', *Climate Research*, vol. 19, no. 2, pp. 109–117.
- Verbeek, M 2004, A Guide to Modern Econometrics, 2nd Edition, John Wiley & Sons, Inc.
- Vincent, K & Cull, T 2010, 'A household social vulnerability index (HSVI) for evaluating adaptation projects in developing countries', Proceedings of the Policies to Foster and Sustain Equitable Development in Times of Crises, 17-18 August, Midrand.
- Vincent, K 2004, 'Creating an index of social vulnerability to climate change for Africa', Working Paper 56, Tyndall Centre for Climate Change Research and School of Environmental Sciences, University of East Anglia.

- Vincent, K 2007, 'Uncertainty in adaptive capacity and the importance of scale', *Global Environmental Change*, vol. 17, no. 1, pp. 12–24.
- Vozoris, NT & Tarasuk, VS 2003, 'Household food insufficiency is associated with poorer health', *Journal of Nutrition*, vol. 133, no. 1, pp. 120–6.
- Walker, B, Holling, CS, Carpenter, SR & Kinzig, A 2004, 'Resilience, adaptability and transformability in social-ecological systems', *Ecology and Society*, vol. 9, no. 2, pp. 5–15.
- Warner, RF & Paterson, KW 2008, 'Bank erosion in the Bellinger Valley, New South Wales: Definition and management', *Australian Geographical Studies*, vol. 25, no. 2, pp. 3–14.
- Warrick, RA & Ahmad, QK (eds) 1996, '*The Implications of Climate and Sea-Level Change for Bangladesh*', Kluwer Academic Publishers.
- WB (World Bank) 2015, World Development Indicators, The World Bank, Washington, DC.
- WB 2000, 'Bangladesh: Climate Change and Sustainable Development. Report No. 21104-BD', Rural Development Unit, South Asia Region, The World Bank, Dhaka.
- WB 2006, Social Safety Nets in Bangladesh: An Assessment, The World Bank, Dhaka.
- WB 2009, Making Development Climate Resilient: A World Bank Strategy for Sub-Saharan Africa (Report No. 46947-AFR), World Bank, Washington, DC.
- WB 2013, Turn Down the Heat: Climate Extreme Regional Impact and Case for Resilience, A Report for the World Bank by the Potsdam Institute for Climate Impact Research and Climate Analytics, Washington, DC.
- WDR (World Disaster Report) 2001, *World Disaster Report 2001*, International Federation of Red Cross and Red Crescent Societies, Geneva.
- Weber, EU 2010, 'What shapes perceptions of climate change?' *Wiley Interdisciplinary Reviews: Climate Change*, vol. 1, no. 3, pp. 332–342.
- WFP (World Food Programme) 2002, *Emergency Needs Assessments*, Report on the Proceeding of the Expert Consultation, World Food Program, Rome.

- WFP 2009, World Hunger Series: Hunger and Markets, World Food Program, Rome.
- Wheeler, S, Zuo, A & Bjornlund, H 2013, 'Climate change beliefs and irrigator adaptability in the Southern Murray-Darling Basin', *Global Environmental Change*, vol. 23, no. 2, pp. 537–547.
- WHO (World Health Organization) 2012, UN Water Global Annual Assessment of Sanitation and Drinking Water (GlAAS) 2012 Report, The challenge of extending and sustaining services, World Health Organization, Geneva.
- Wichstrom, L, von Soest, T & Kvalem, IL 2013, 'Predictors of growth and decline in leisure time in physical activity from adolescence to adulthood', *Health Psychology*, vol. 32, no. 7, pp. 775–784.
- Williams, A 2003, 'How to write and analyze a questionnaire', *Journal of Orthodontics*, vol. 30, no. 3, pp. 245–52.
- Wilson, GA 2012, 'Community resilience, globalization, and transitional pathways of decision-making', *Geoforum*, vol. 43, no. 6, pp. 1218–1231.
- Wolf, J, Adger, WN, Loronzoni, I, Abrahamson, V & Raine, R 2010, 'Social capital, individual responses to heat waves and climate change adaptation: An empirical study of two UK cities', *Global Environmental Change*, vol. 20, no. 1, pp. 44–52.
- Wood, SA, Jina, AS, Jain, M, Kristjanson, P & DeFries, RS 2014, 'Small holder farmer cropping decisions related to climate variability across multiple regions', *Global Environmental Change*, vol. 25, pp. 163–172.
- Wooldridge, JM 2006, *Introductory Econometrics: A Modern Approach*, 3rd Edition, Thomson South-Western, Australia.
- YCELP & CIESIN 2002, *Environmental Sustainability Index*, An Initiative of the Global Leaders of Tomorrow Environment Task Force, Yale Centre for Environmental Law and Policy and Centre for International Earth Science Information Network, Columbia University.
- Yeh, ET, Nyima, Y, Hopping, KA & Klein, JA 2013, 'Tibetan pastoralists' vulnerability to climate change: A political ecology analysis of snowstorm coping capacity', *Human Ecology*, vol. 42, no. 1, pp. 61–74.

- Yohe, G & Tol, RSJ 2002, 'Indicators for social and economic coping capacity moving toward a working definition of adaptive capacity', *Global Environmental Change*, vol. 12, no. 1, pp. 25–40.
- Yotopoulos, P 1983, 'A micro economic-demographic model of the agricultural household in the Philippines', FAO Economic and Social Development Paper, Rome.
- Yusuf, HKM, Dasgupta, S, Khan, MAH 2008, *Climate Change: An Emerging Threat to Agriculture and Food Security in Bangladesh*, FAO, Dhaka.
- Zaman, MQ 1989, 'The socioeconomic and political context of adjustment to riverbank erosion hazard and population resettlement in Bangladesh', *Human Organization*, vol. 48, no. 3, pp. 196–205.
- Zaman, MQ 1991, 'The displaced poor and resettlement policies in Bangladesh', *Disasters*, vol. 15, no. 2, pp. 117–125.
- Zaman, MQ 2007, Flood and Erosion Disasters in Bangladesh: The Human Dimensions, Vancouver.
- Ziervogel, G & Ericksen, JP 2010, 'Adapting to climate change to sustain food security', Wiley Interdisciplinary Reviews: Climate Change, vol. 1, no. 4, pp. 525–540.
- Ziervogel, G 2004, 'Targeting seasonal climate forecasts for integration into household level decisions: The case of smallholder farmers in Lesotho', *The Geographical Journal*, vol. 170, no. 1, pp. 6–21.

APPENDICES

Appendix 1: Types of disasters and their impacts in specific disaster prone areas.

Types of	Areas Affected	Impact
Disaster		
Flood	Floodplains of the Brahmaputra- Jamuna, the Ganges-Padma and the Meghna river system	Loss of agricultural production, disruption of communication and livelihood system, injury, damage and destruction of immobile infrastructure, disruption to essential services, national economic loss, evacuation, and loss of human lives and biodiversity, displacement and sufferings of human population and biodiversity
Cyclone and Storm Surge	Coastal areas and offshore islands	Loss of agricultural production, disruption of communication and livelihood system, damage and destruction of immobile infrastructure, injury, national economic loss, loss of biodiversity and human lives, need for evacuation and temporary shelter
Tornado	Scattered areas of the country	Loss of human life and biodiversity, injury, damage and destruction of property, damage of cash crops, disruption in lifestyle, damage to essential services, national economic loss and loss of livelihood
Drought	Almost all areas, especially the northwest region of the country	Loss of agricultural production, stress on national economy and disruption in life style
Flash Flood	Haor Basins of the north-east region and south- eastern hilly areas	Damage of standing crops, disruption in life style, evacuation and destruction of properties
Hail Storm and Lightning	Any part of the country	Damage and destruction of property, damage and destruction of subsistence and cash crops and loss of livelihood
Riverbank Erosion	River banks of the Brahmaputra- Jamuna, the Ganges-Padma and the Meghna	Loss of land, displacement of human population and livestock, disruption of production, evacuation and loss of property
Landslide	Chittagong and Chittagong Hill Tracts	Loss of land, displacement of human population and livestock, evacuation, damage of property and loss of life
Earthquake	Northern and central parts of the country	Damage and destruction of property, loss of life and change in geomorphology

Source: UNEP 2001

Appendix 2: Questionnaire (Back translation of the questionnaire was used)

1. General information:	ID	

Upazila: Union: Village:

Name of the household head:

Total Population: Male- Female- Children- Old (65)- Earning members-

2. Family type: □Unit or Nucleus □Joint □Other (specify)

3. Information of household members/human resources

Sl. No.	Relations hip with HH Head	Age	Sex(M/F)	Marital Status	Education level	Earning members (Yes=1, No=0)	Staying with family (Yes=1, No=0)	Profes Main	sion Secondary
1	Self						110-0)		
2									
3									
4									
5									
6									
7									

HH Head = 1, Spouse = 2, Father = 3, Mother = 4, Brother = 5, Sister = 6, Son = 7, Daughter = 8, Daughter-in-law = 9, Grandson = 10, Granddaughter = 11, Others = 12 (specify).....,

Marital Status: Unmarried =1, Married = 2, Widow = 3, Divorced = 4, Other = 5 (specify)

4. Land property

Types of land	Amount (in decimal)
Own cultivable land	
Rented-in	
Rented-out	
Mortgage in	
Mortgage-out	
Homestead land	
Orchard/Garden	
Pond	
Fellow land	

5. Household yearly income (Tk.)

Income from agricultural sources		Income from non-agricultural sources		
Items Total Taka		Items	Income	
	(Tk.)		(Tk.)	

Rice	Service
Wheat	Business
Spices / pulses / oil seed	Pension
crops	
Fruits and vegetables	Remittance
Livestock and poultry	Other occupation
	(selling labour,
	rickshaw pulling, boat,
	handicrafts etc.)
Fisheries	Safety net (VGD/VGF,
	food for work etc.)
	allowance
Others (if any)	Charity (<i>fitra, jakat</i> ,
	help etc.) and begging
	Others (if any)

6. Household yearly expenditure (Tk.)

Expenditure head	Total	Expenditure	Total expenditure
	expenditure	head	(Tk/Month)
Crop farming		Food (1)	
Clothing		Rice	
Children education		Wheat	
Health care		Fish	
House making/repairing		Meat	
Festivals		Egg	
Livestock rearing		Milk	
Poultry keeping		Pulses	
Other costs		Species & Oil	
		Fruits &	
		Vegetables	

7. Household wealth

Items	Quantity	Value	Items	Quantity	Value
TV			Rickshaw/van		
Mobile phone			Tempo		
Radio			Furniture		
Camera			Cassette player		
VCD/DVD			Iron		
Computer			Refrigerator		
Laptop			STW		
Motor cycle			Generator/motor		
Cycle			Sewing machine		
Torch light			others		

8. Livestock resources

Type of livestock	Number	Present value
1. Bull/Ox		
2. Cow		

3. Calf	
4. Pig (adult)	
5. Pig (calf)	
6. Goat (adult)	
7. Chicken(adult)	
8. Other (please specify)	

9. Overall livelihood status of the households

Questions on livelihood status	YES	NO	If NO, why ?
Does your family use sanitary toilet?			
Does your family use tube well water?			
Does your family use electricity?			
Does your family buy new clothes during			
festivals?			
Does your family offer gifts to relatives			
during different social events?			
Do you use any contraceptive method?			
What type of doctors do you normally visit	(a) MBB	S, (b) Vi	llage doctor
while you are sick?	(c) Hom	eopathic	, (d) Quack
Does your family send children to school?			
Are you a member of any cooperative			
society?			
Do you have any saving accounts?			
Are your family members a member of			
cooperative society?			
Do you explore and utilise information			
technology for professional, health and family			
planning activities?			
Have you received any training in your			
profession?			
Do you get cooperation from other village			
people in case of your need?			
Do you adopt zero tillage cultivation?			
Do you adopt new cropping practice?			
Do you adopt improved management of			
weeds?			
Do you adopt improved management of			
manure?			
Do you adopt IPM			
Do you cultivate multiple crops?			
Do you allow women to participate in			
decision making processes?			
Current health condition of the household	(a) Good	=1 (b) l	Poor = 0
head (observing physical health)?			
How many days you were absent from the			
work due to sickness?			
How many years have you had working in		Years	
your main profession?			

Months	Level of food availability			
	Adequate	Inadequate	Scarce	
Baisakh (mid-April to mid-				
May)				
Jaisthya				
Ashaar				
Shraban				
Bhadra				
Ashwin				
Kartik				
Agrahayon				
Puosh				
Magh				
Falgun				
Chaitra (mid-March to mid-				
April				

11. Household food insecurity situation (Perception)

Food Insecurity Status		If yes, how many times did these happen?			
Food Insecurity Status	Yes No	Most of the time	Sometimes	Sudden	
a) Were members of your household					
anxious about the lack of sufficient food during the last three months?					
b) Were you or any member of your					
household bound to eat fewer than three					
sufficient food during the last three					
months?					
c) Did you or any other member of your					
household go to bed hungry due to lack					
of sufficient food during the last three months?					

12. Availability of food from own production

Type of food	No.	of	months	Type of food	No. of months cover
	cover				
Rice				Pulse	
Wheat				Species	
Fish				Oil	
Meat				Fruits	
Egg				Vegetables	
Milk					

Food Items	Unit	Yeste	erday	Day before vesterday		Two days before	
1. Cereals		Quantity	No. of people (Adult+ Child)	Quantity	No. of people (Adult + Child)	Quantity	No. of people (Adult + Child)
Rice	kg				,		,
Chira	gm						
Muri & Khai	gm						
Wheat/Flour	gm						
Samai/Sugi	gm						
Bread/Biscuit	gm						
Cake	gm						
Others	gm						
2. Pulses							
Masoor	gm						
Khesari	gm						
Moog	gm						
Boot	gm						
Kalai	gm						
Others	gm						
3. Spices & Oil							
Onion	gm						
Garlic	gm						
Oil	gm						
Chili & ginger	gm						
Salt	gm						
Other spices	gm						
4. Fruits							
Banana	gm						
Coconut	gm						
Papaya	gm						
Apple	gm						
Others	gm						
5. Vegetables							
Potato	gm						
Brinjal	gm						
Radish	gm						
Bean	gm						
Cabbage/	gm						

13. Food Consumption in Last 3 Days

Cauliflower				
Others				
6. Meat& Egg				
Cow& Buffalo	gm			
Goat	gm			
Hen/Poultry	gm			
Duck	gm			
Egg	gm			
Others	gm			
7. Fish				
Small Fish	gm			
Medium Fish	gm			
Big Fish	gm			
Shutki	gm			
8. Milk & others	litre			
Milk	litre			
Other milk products	litre			
Sugar	litre			
Molasses	litre			
Others	litre			

14. Please rank the following questions

In your household, who get food first, then second and so on during a severe food crisis?

Types	Rank	Types	Rank
Husband		Siblings	
Wife		Daughter	
Son		Old-age person	
Pregnant wife		All have equal chance	

15. Local orientation

Were you affected by rivert	ank erosion?	□Yes	\Box No			
\Box If you are not affected by riverbank erosion, move on to Question 18						
How were you affected by riverbank□ Loss of farm land□ Homestead land□ Pond(tick all those that apply)						
How many times were you	affected by riverba	nk erosion	during last 10 years?			

In which year did the most recent riverbank erosion event occur?							
Was your household food secure before the first riverbank erosion? \Box Yes \Box No							
□ Not sure/ Don't know							
Distance of homestead from the river bank (m)							
Amount of land within 1/2 km of river bank (decimal)							
Housing condition- \Box Tin \Box House with only roof with tin \Box Hut							

16. Impact of river bank erosion on different household indicators

Items	After riverbank erosion	Reasons
Housing condition (No change=0, deteriorate = 1, improve=2)		
Occupation opportunities of household members (No change=0, reduce = 1, improve=2)		
Working hours of the household members (Average hours) (No change=0, reduce = 1, improve=2)		
Income of the household (No change=0, reduce = 1, improve=2)		
Education facilities (No change=0, reduce = 1, improve=2)		
Health facility (No change=0, reduce = 1, improve=2)		
Electricity (No change=0, reduce = 1, improve=2)		
Sources of drinking water (No change=0, reduce = 1, improve=2)		
Sanitary (toilet) conditions (No change=0, reduce = 1, improve=2)		
Transport facilities (No change=0, reduce = 1, improve=2)		
Communication facilities (No change=0, reduce = 1, improve=2)		
NGO services (No change=0, reduce = 1, improve=2)		
Institutional credit facilities (Bank) (No change=0, reduce = 1, improve=2)		
Mechanised agricultural instruments (Number) (No change=0, reduce = 1, improve=2)		
Household assets (No change=0, reduce = 1, improve=2)		
Food security condition (No change=0, reduce = 1, improve=2)		
Availability of labour (No change=0, reduce = 1, improve=2)		

Marketing facilities (No change=0, reduce = 1, improve=2)		
	At present	Before riverbank erosion
Land size (in decimal – square metres)		
Land use patterns (major)		
Occupation of the household head		
Do you earn more income than your previous occupation? (If your occupation has changed)		JNo
Main source of household income (Remittance=1, Agriculture=2, Job=3, Business=4, Labour Sale=5, Non-agriculture=6, others=7)		
Wage of day labour (amount/day)		
Income of the household		
Changes in livestock numbers (no.)		
Changes in poultry and ducks (no.)		

17. Perceptions about riverbank erosion hazards/ and climate change

- (a) Have you noticed/ perceived any change to the climate in your locality over the last 20 years?
 - \Box Yes \Box No \Box Not sure/Don't know
- (b) If yes, identify which of the climate variables you think have changed and describe how they have changed.

Climate components	Time	Increased	Decreased	No	Don't
	period			change/	know
				same	
Temperatures	Annual				
	Winter				
	Summer				
Rainfall	Annual				
	Winter				
	Summer				
Extreme events such as	Annual				
drought					
Availability of groundwater	Annual				
Availability of surface water	Annual				
Severity of cold wave	Annual				
Severity of heatwave/hot	Annual				
days					
Severity of riverbank	Annual				
erosion					
Frequency of flood	Annual				
Frequency of cyclones	Annual				
Winter period	Annual				

Summer period	Annual		
Others, if any (please	Annual		
specify)			

18. Coping strategies towards food security during stress situation

SL.	Food security strategies	Degree of responses			
No.		Regularly	Occasionally	Rarely	Not
					at all
1.	Reduce amount of food per meal				
2.	Reduce number of meals per day				
3.	Go bed without food				
4.	Rely upon less expensive or less				
	preferred food items				
5.	Reduce buying children food(i.e.				
	milk) from market				
6.	Purchase food on credit				
7.	Borrow money from NGOs/GB				
8.	Borrow from money lenders				
9.	Migrate to city or other area				
10.	Rely on casual labour for food				
11.	Sell cattle/livestock				
12.	Spend money from deposit				
13.	Borrow money or food from				
	friends/relatives				
14.	Sell land and other asset (specify)				
15.	Sell labour in advance				

Regularly = 3; Occasionally = 2; Rarely = 1; Not at all = 0

19. Government/NGO programs for food security in the area

- a) Are you or family members are under any safety net programs available in your locality from government sector? □Yes □No
- b) If yes, for how many years?---
- c) Please describe the nature of that support (amount and months): -----.....TK.
- d) What is your opinion about the impact of these programs on your household food security? □ Adequate □Inadequate
- e) Is there any NGOs program towards food security? □Yes □No
- f) If yes, please describe-
- g) What is your opinion about their support to ensure food security? □
 Adequate □Inadequate

20. Adaptation to riverbank erosion hazards/climate change

Have you made any changes to your farm operations due to riverbank erosion or changes in climate attributes in order to reduce the adverse impacts?

 \Box Yes \Box No

21. What adaptive measures (adjustments) do you practice in your farming system?

Adaptive measures	Please put 1 for main measure and
	tick (\Box) for others that you practice
Change planting time	
Cultivation of pulses	
Cultivation of wheat and other crops	
Tree plantation	
Cultivation of spices and oil seed	
Cultivation of local Aman rice	
Cultivation of vegetables	
Cultivation of HYV rice varieties	
(e.g., BRRI-28, 29)	
Livestock rearing	
Poultry rearing	
Duck rearing	
Homestead gardening	
Migration	
Off-farm work	
(van, rickshaw, tempo driving)	
Petty business	
No adaptation	
Others (please specify)	

22. What are your barriers in taking adaptive measures?

Barriers to adaptation	Please put 1 for main measure and
	tick (\Box) for others that you practice
Lack of information about riverbank	
erosion and related climatic issues	
Lack of appropriate variety of crops	
Lack of knowledge concerning	
appropriate adaptation strategies	
Lack of credit/money/saving	
Lack of suitable land for cultivation	
Lack of own land	
Lack of storage facilities	
Lack of marketing facilities	
Lack of transportation facilities	
Others (please specify)	

--Thank you for your cooperation---